ORIGINAL

SQUIRE, SANDERS & DEMPSEY L.L.P. Including STEEL HECTOR & DAVIS LLP

215 South Monroe Street, Suite 601 Tallahassee, Florida 32301-1804

Office: +1.850.222.2300 Fax: +1.850.222.8410

> Charles A. Guyton cguyton@ssd.com

> > RECEIVED-FPSC

SEP -8 PM 4: 3

September 8, 2006

VIA HAND DELIVERY

Blanca S. Bayó, Director Division of the Commission Clerk & Administrative Services Florida Public Service Commission 2540 Shumard Oak Boulevard Tallahassee, Florida 32399-0850

LEGAL

COUNSEL

WORLDWIDE

Re: Docket No.: 060512-EU

Dear Ms. Bayó:

Enclosed for filing on behalf of Florida Electric Cooperatives Association, Inc. are the original and seven (7) copies of the following:

(i) Comments of the Florida Electric 25-6.0343; Characteric (ii) Direct Testimony of William B. (iii) Direct Testimony of John Martz (iii) Direct Testimony of John Martz TR	Cooperatives Association, Inc. to Proposed Rule Willingham; and のなるうる- 0ゆ のるるうろ- 0し rding this transmittal, please contact me at 222-2300.
CR	Very truly yours,
CL	
RECEIVED & FILED	SQUIRE, SANDERS & DEMPSEY L.L.P.
EC CAG:gcm	
TH Enclosure Kuller Copy to: Interested Parties	
	DOCUMENT - HIGHER - THOMAS - THE STORE - T
CINCINNATI • CLEVELAND • COLUMBUS • HOUSTON • LOS ANGELES • MIAMI • NEW	YORK • PALO ALTO • PHOENIX • SAN FRANCISCO • I ALLAHASSEE • LAMPA D'AGNS CORNI

Washington DC · West Palm Beach | Caracas · Rio de Janeiro · Santo Domingo | Bratislava · Brussels · Budapesto London · Madrid · Milan · Moscow Prague · Warsaw | Beijing · Hong Kong · Shanghai · Tokyo | Associated Offices: Bucharest · Buenos Aires Dulli · ISPP Sagrago

www.ssd.com

FPSC-COMMISSION OF --

BEFORE THE FLORIDA PUBLIC SERVICE COMMISSION

)

)

In Re: Adoption of new rule 25-6.0343, F.A.C., standards of construction -municipal electric utilities and rural electric cooperatives

Docket No. 060512-EU

<u>COMMENTS OF THE FLORIDA ELECTRIC COOPERATIVES</u> <u>ASSOCIATION, INC. TO PROPOSED RULE 25-6.0343</u>

The Florida Electric Cooperatives Association, Inc. ("FECA"), on behalf of its member cooperatives,¹ by and through its counsel, files the following comments to proposed Rule 25-6.0343, Municipal Electric Utilities and Rural Electric Cooperatives, that was issued on June 28, 2006 in Order No. PSC-06-0556-NOR-EU. While proposed Rule 25-6.0343 was proposed in Docket Nos. 060172-EU and 060173-EU, the Commission has created a separate docket for consideration of the proposed rule, Docket No. 060512-EU. See, Order PSC-06-0632-PCO-EU. FECA also adopts and incorporates herein its written comments filed on May 3 and 26, 2006, and oral comments given on April 17, May 19 and June 20, 2006. Contemporaneous with these comments, FECA is also filing with the Florida Public Service Commission's ("Commission") the testimony of Mr. John Martz and Mr. William B. Willingham.

¹Alabama Electric Cooperative, Inc., Central Florida Electric Cooperative, Inc., CHELCO, Clay Electric Cooperative, Inc., Escambia River Electric Cooperative, Inc., Florida Keys Electric Cooperative Association, Inc., Glades Electric Cooperative, Inc., Gulf Coast Electric Cooperative, Inc., Okefenoke Rural Electric Membership Corporation, Peace River Electric Cooperative, Inc., Seminole Electric Cooperative, Inc., Sumter Electric Cooperative, Inc., Suwannee Valley Electric Cooperative, Inc., Talquin Electric Cooperative, Inc., Tri-County Electric Cooperative, Inc., West Florida Electric Cooperative, Inc., Withlacoochee River Electric Cooperative, Inc. Lee County Electric Cooperative is not a member of FECA.

I INTRODUCTION

In response to the impacts of hurricanes over the last two years and in anticipation of future storms which could interrupt service and require significant time for restoration of electric infrastructure, the Commission has proposed rules requiring all electric utilities subject to its jurisdiction to undertake conduct which the Commission believes will enhance the reliability of transmission and distribution facilities and reduce storm restoration time. FECA shares with the Commission a concern about the reliability of electric transmission and distribution facilities in severe weather events and the need to minimize storm restoration time through acts that are reasonable, practical, feasible and cost-effective. Indeed, in response to the storms of the last two years, FECA's members have undertaken a number of actions designed to enhance the reliability of their systems during severe storm weather events. See pages 13 - 14.

While FECA shares the Commission's concerns about enhanced storm reliability, FECA is concerned about the Commission's approach thus far. The Commission has proposed that the same requirements should apply to not-for-profit, self-governing rural electric cooperatives ("cooperatives") which have elected boards comprised of members served by the cooperatives, as apply to investor-owned electric public utilities ("IOUs"). Given the dramatically different relationship between cooperatives and their members and IOUs and their ratepayers as well as the sharply different relationship between the Commission and comprehensively regulated IOUs and the Commission and cooperatives, FECA respectfully submits that no rule for cooperatives is warranted. If the Commission believes a rule for cooperatives is warranted, a separate rule tailored to the circumstances of cooperatives would be appropriate. Any rule adopted regarding cooperatives must necessarily recognize the much more limited jurisdiction the Commission has

over cooperatives than it does over IOUs. The proposed rule fails to recognize any jurisdictional difference.

The Commission, in response to the request of the cooperatives and municipalities, has agreed to propose a separate rule for those entities. More recently, the Commission has agreed to a separate docket for that rule and those entities. Those are positive and encouraging developments. However, the rule proposed for cooperatives is, in its current form, the same rule as has been proposed for IOUs.

FECA respectfully submits there are multiple reasons why no rule for cooperatives is warranted or that if a rule for cooperatives is to be adopted, the rule applicable to cooperatives should be different from the rule proposed for IOUs. FECA appreciates the opportunity the Commission has provided the cooperatives with a separate docket to develop those differences, explore whether a rule for cooperatives is needed and to propose a reasonable alternative. FECA is optimistic that when the record is fully developed, the Commission will acknowledge that (a) the significantly different relationship between self-governed, not-for-profit cooperatives and their members (customers) relative to the relationship of IOUs and their ratepayers, (b) the significantly different relationship of the Commission to cooperatives and their members relative to the relationship between for-profit, IOUs and their ratepayers, (c) the role of the Rural Utilities Service ("RUS") with most Florida cooperatives, and (d) the comprehensive jurisdictional grant of authority to the Commission over IOUs and the limited jurisdictional grant of authority to the Commission relative to cooperatives, all warrant either no rule for cooperatives or at most, a less prescriptive rule for cooperatives than the rule proposed for IOUs.

While FECA still advances the option of the Commission proposing no rule for

cooperatives, FECA has proposed an alternative rule to the Staff of the Commission which should meet the Commission's goals in this proceeding. The proposed alternative rule would reinforce the cooperative relationship that has evolved between the Commission and rural electric cooperatives over the last thirty years. FECA's proposed alternative, which is attached hereto as Attachment A, is a least cost regulatory alternative that addresses all of the stated goals of proposed rules 25-6.034, 25-6.341 and 25-6.0342. FECA requests that if the Commission determines that any rule is necessary for cooperatives, that the Commission adopt the attached rule in lieu of proposed Rule 25-6.0343.

FECA's Comments are divided into five sections in addition to this Introduction. Section II addresses the historic relationship of cooperatives, their members and the Commission and provides a rationale for no rule for cooperatives or a rule for cooperatives separate and distinct from IOUs. Section III addresses RUS requirements applicable to and followed by RUS cooperatives. Section IV addresses the unique customer density and cost profiles of cooperatives, the high costs associated with implementing extreme wind load standards for cooperatives, and the efforts cooperatives have undertaken to address system storm reliability. Section V addresses FECA's proposed alternative rule. Section VI addresses Rule 25-6.0343 as proposed by the Commission.

Once again, FECA thanks the Commission for its recognition thus far that cooperatives warrant their own rule and docket. FECA is confident that the same understanding that led to a separate rule and separate proceeding will lead the Commission to the conclusion either that no rule for cooperatives should be adopted or that rule requirements different than IOU rule requirements are warranted.

4

II <u>THE RELATIONSHIP OF COOPERATIVES,</u> <u>THEIR MEMBERS AND THE COMMISSION</u>

Before addressing whether to adopt a rule for cooperatives or whether to adopt either FECA's alternative rule or the Commission's proposed rule for cooperatives and municipalities, it is important to recognize and discuss the unique relationship of not-for-profit, self-governed cooperatives with the members they serve and the Commission's role in that relationship relative to the relationship between for-profit IOUs and their ratepayers and the Commission's role in that relationship. These are very different relationships and roles, and they provide a fundamental rationale for not adopting a rule for cooperatives or for adopting a different rule for cooperatives than for IOUs.

Rural electric cooperatives were organized to meet a growing need for reliable electricity service in rural areas of America. In 1935 when President Roosevelt created the Rural Electrification Administration ("REA") by executive order, nine out of ten rural homes were without electricity. This lack of an essential service was frustrating economic development of rural areas, forcing them to retain an agrarian economy. A year later Congress passed the Rural Electrification Act, creating a low cost lending program administered by REA that allowed rural electric systems to organize and fund necessary facilities.

Florida's electric cooperatives have a proud history of providing reliable, at-cost electric service to the rural and suburban areas of Florida. Florida's electric cooperatives were formed in the 1930s to serve areas that were not being served by other utilities. Cooperatives were created by the people and businesses that needed electricity, and today they are still owned by those they serve.

In 1940 the Florida legislature acted to facilitate the creation of rural electric cooperatives in Florida by enacting the Rural Electric Cooperative Law, which was codified as Chapter 425, Florida Statutes. Section 425.01, Florida Statutes. Under Chapter 425, each cooperative is a "cooperative, nonprofit, membership corporation … organized … for the purpose of supplying electric energy and promoting and extending use thereof in rural areas." Section 425.02, Florida Statutes.

Each cooperative organized under Chapter 425 is governed by a board of trustees, which consists of members (customers) served by the cooperative. Section 425.10, Florida Statutes. The trustees are elected by the members of the cooperatives. Id. In addition, cooperatives conduct annual, open meetings of its members as well as special meetings called by the board of trustees or at least ten percent of the members. Section 425.09, Florida Statutes.

Simply stated, cooperatives are democratic organizations controlled by their members who actively participate in setting policies and making decisions. The boards are comprised of members who have no interest to serve other than those of their fellow members. There are no shareholders with profit expectations. Since the members own the cooperative and control its policies through democratic processes, there is no motive for the cooperative to act in any fashion that is not in the interests of its members.

Recognizing the not-for-profit, self-governing aspects of cooperatives, from 1940 until 1974, the Florida Legislature withheld from the Commission any regulatory oversight of rural electric cooperatives. Since 1974, when the Legislature gave the Commission limited jurisdiction over rural electric cooperatives, the Legislature has continued to recognize there is not the need for the Commission to act to protect cooperative members in the same fashion as it needs to act to protect ratepayers of investor-owned public utilities. Just as the Legislature has recognized there is not the need to regulate cooperatives as there is the need to regulate IOUs, the Commission should recognize that the same rule is not necessary for cooperatives and IOUs.

Just as there is no need for the Commission to set rates to protect cooperative customers, there is not the same level of need for the Commission to act to assure reliability of distribution facilities owned by the members of cooperatives. These facilities are owned by the members they serve. The facilities exist solely to provide reliable service to the members. They are not owned by shareholders who expect a market based return on their investment. The boards of trustees when making decision regarding construction standards and vegetation policies and other matters that affect reliability do not have to balance competing interests of shareholders and ratepayers. The boards of trustees are simply acting, as democratically elected representatives, to preserve and enhance the reliability and quality of service to their fellow members. Thus, the fundamental relationship between cooperatives and their members suggests there is far less need for the Commission to act to protect the interests of members of cooperatives. This should be considered by the Commission in its rulemaking. It is a rational, indeed compelling, basis for making distinctions between the rule proposed for IOUs and the rule proposed for cooperatives or for deciding not to adopt at all for cooperatives.

III RURAL UTILITIES SERVICE REQUIREMENTS

Most of Florida's cooperatives have low interest loans from the RUS. RUS borrowers are required by their loan covenants to comply with the RUS' rules and regulations. Most of those Florida cooperatives which are not RUS borrowers nonetheless follow RUS guidelines to preserve their future ability to borrow from RUS.

The RUS has expertise in the area of designing rural electric facilities and has created construction specifications that its borrowers must use. RUS' specifications have been developed over decades based upon RUS' extensive history with nearly 1000 electric cooperatives in the United States, and by adopting national standards of groups such as the American National Standards Institute, American Wood Preservers Association, various national engineering societies and the National Electrical Safety Code ("NESC"). The RUS also requires borrowers to maintain and test their Emergency Response Plans.

RUS' requirements regarding distribution system planning, construction, operation and maintenance are extensive and are contained not only in regulations in the Code of Federal regulations ("CFR"), but also in Bulletins and Information Publications. The Commission is familiar with RUS Bulletins and their guidance, as RUS pole inspection requirements were relied upon, in part, by the Commission in entering Order No. PSC-06-0144-PAA-EI, its pole inspection order in Docket No. 060078-EI. Some of the RUS Bulletins are incorporated by reference into the CFR regulations.

It is not practical for FECA to forward to the Commission as part of its comments all applicable RUS regulations and bulletins. However, it is helpful to provide to the Commission indices of the RUS regulations and bulletins and the text of the RUS regulations applicable to distribution systems and storm restoration. It is important for the Commission to understand that RUS has already acted extensively in the areas covered by the Commission's proposed rule and that in significant measure the Commission's rule is redundant, unnecessary and could possibly even conflict with RUS requirements. The Rural Utilities Service Electric Program Regulations are posted on the United States Department of Agriculture's ("USDA") web site. The index of those regulations is found on the following website: <u>http://www.usda.gov/rus/electric/regs/index.htm</u>. A copy of that index is attached as part of Attachment C. At a minimum, the Commission should be aware of the following regulations mentioned in that index:

 7 CFR Part 1724, Electric engineering, architectural services and design policies and procedures.
7 CFR Part 1726, Electric system construction policies and procedures
7 CFR Part 1728, Electric standards and specifications for materials and construction
7 CFR Part 1730, Electric system operations and maintenance

For the Commission's ease of reference, all those regulations are also found in Attachment C.

The RUS requires compliance with the National Electrical Safety Code ("NESC"). 7 CFR Part 1724.50. It then goes beyond the requirements of the NESC and requires for distribution facilities conformance "to the applicable RUS construction standards" and utilization of "RUS accepted materials." 7 CFR Part 1724.51(a). RUS also requires the preparation of work plans and specifications for distribution facilities, 7 CFR part 1724.53, and RUS approval of such plans, 7 CFR Part 1724.54(a)(b).

In 7 CFR Part 1728, RUS provides extensive guidance regarding specifications and standards for materials, equipment and construction units that will be used for RUS financial assistance. RUS uses standards from national groups (American National Standards Institute, American Wood Preservers' Association, national engineering societies and the NESC) "to the greatest extent practical." 7 CFR 1728.20(a). RUS has an extensive procedure for including items for its standards listings or technical acceptance, 7 CFR Part 1728.30 – 1728.60, and requires borrowers to procure listed items, 7 CFR Part 1728.70. RUS incorporates by reference

numerous electric bulletins that it has issued. 7 CFR Part 1728.97 through 1728.202.

RUS' regulations also contain various operations and maintenance requirements that are relevant to this proceeding. Each borrower must maintain its system in compliance with "prudent utility practice ... and all applicable laws, regulations and orders" and "shall maintain its systems in good repair, working order and condition, and shall make all needed repairs, renewals, replacements, alterations, additions, betterments and improvements...." 7 CFR Part 1730.20. Each borrower must also perform Vulnerability and Risk Assessments and maintain an Emergency Restoration Plan. Id. RUS borrowers also must conduct necessary inspections and tests, and the inspections must include determinations of compliance with the NESC. 7 CFR Part 1730.21. Borrowers must periodically analyze and document its security and O&M practices and performs ratings, which are subject to RUS review. 7 CFR Part 1730. 22 through 24.

As previously noted, there are extensive Bulletins issued by the RUS that supplement the requirements of RUS' regulations. An index of those Bulletins is found in Attachment D. The index is found at the following website, where specific Bulletins can be accessed: <u>http://www.usda.gov.rus/electric/bulletins.htm</u>. As one can see from the index, the vast bulk of the Bulletins corresponds to and supplements Parts 1724 through 1730 of the regulations.

FECA respectfully submits that given the existing requirements of RUS in the form of its regulations and bulletins applicable to RUS cooperatives, there is no need for the Commission to require by rule the adoption of construction standards or compliance with the National Electrical Safety Code. Exacting and demanding standards already are in place for RUS cooperatives. Moreover, Florida's cooperatives borrowing or hoping to borrow from the RUS already have to

comply with not only the NESC but also RUS' requirements.

IV COOPERATIVE DEMOGRAPHICS, COSTS AND STORM RESPONSES

The demographics and nature of a cooperative's service territory are unique. Cooperatives serve more than sixty percent of Florida's landmass, but they serve less than twelve percent of Florida's population. Nationally, the majority of most cooperatives' service territories are rural, and cooperatives have only seven (7) member-owners per mile of line. This compares to average customers per mile of line for IOUs and municipalities of 35 and 47, respectively.²

Despite the low density and the corresponding high cost per customer of serving the rural areas, cooperatives' rates are competitive with their neighboring utilities. However, cooperatives are concerned that if the same rule requirements are applied to cooperatives as are applied to IOUs, given the cooperatives' low customer density and high cost service characteristics, cooperatives rates will be forced to increase rates without any assurance of improved reliability or storm restoration time.

For instance, in earlier comments, FECA provided cost estimates associated with complying with extreme wind loading standards. Those costs are significant, and they appear to have been overlooked. They warrant re-emphasis here, given the Commission's proposed rule that requires cooperatives "to be guided by the extreme wind loading standards specified by Figure 250-2(d) of the 2002 edition of the NESC."

Withlacoochee River Electric Cooperative, Inc., which is located in an extreme wind loading area of 130 mph, has estimated the materials cost of complying with the extreme wind

² This is based on 2004 EIA and RUS data.

loading standards of NESC 250 C rather than the applicable wind loading standard of NESC 250 B. Those materials cost estimates (exclusive of labor, vehicles, etc.) are shown on Attachment B. The materials cost of construction of new distribution facilities would escalate alarmingly for Withlacoochee and similarly situated cooperatives. Different pole types would be required; span lengths would be significantly shortened; and the resulting costs per mile for various circuits would increase dramatically. The estimated increase in materials costs associated with compliance with extreme wind loading standards is as follows:

Facility	Materials Cost Increase
Single Phase #2 AAAC	65%
3 Phase 394 AAAC Single Circuit	96 - 101%
3 Phase 740 AAAC Single Circuit	87 - 94%
3 Phase 394 AAAC Double Circuit	68 - 159%
3 Phase 740 AAAC Double Circuit	50 - 142%.

These dramatic projected cost increases associated with following extreme wind load standards are sobering, but given other testimony the Commission has heard, it is difficult to understand why the Commission is proposing a rule for cooperatives to be guided by extreme wind load standards.

Compliance with extreme wind load standards is very expensive, but it would not even address the primary cause of loss of distribution facilities during storm events – trees and flying debris hitting lines. As FECA has previously testified, during the 2004 and 2005 hurricane seasons, most cooperative pole failure (more than 50%) was due not to direct wind within the cooperatives' applicable extreme wind ratings (which is what the extreme wind loading standards address), but to tornadic winds and flying debris (which the extreme wind load standards do nothing to prevent). For most cooperatives, the number of poles that failed due to straight wind within applicable ratings was insignificant, and many of those poles were built to meet extreme wind loading. Adherence to extreme wind loading standards by cooperatives appears to be a costly but ineffective approach.

Moreover, the adoption by cooperatives of extreme wind loading standards likely would increase rather than decrease storm restoration time. Compliance with extreme wind loading standards significantly decreases the span lengths, requiring more poles and more spans exposed to the same amounts of flying debris. If cooperatives complying with extreme wind standards suffered the same amount of line mileage repair due to tornadic winds, trees and flying debris, the number one cause of distribution system loss, restoration time would necessarily increase, because more poles and more spans would have to be replaced.

Thus, FECA respectfully submits that a rule requiring cooperatives to be guided by extreme wind loading standards would actually frustrate rather than improve storm reliability and storm restoration. That is a decision best left to cooperative's representative boards, which are far more familiar with their service territories, their vulnerability to storm related outages and the service requirements of their members.

Cost considerations aside, in deciding whether to proceed with the existing proposed rule, a less prescriptive rule commensurate with the Commission's more limited jurisdiction over cooperatives, or no rule at all for cooperatives, the Commission should also be aware of the actions Florida's cooperatives have undertaken and are undertaking to improve storm reliability. Florida's cooperatives have been proactive in regard to storm recovery, and their actions suggest there is no need for a prescriptive Commission rule.

As noted previously, most of Florida's cooperatives already comply with RUS' extensive requirements, requirements that the Commission is already relying upon in its pole inspection docket. Thus, there is no need for the Commission to require construction standards for cooperatives.

All of FECA's members have increased their vegetation management programs. Of course, this directly addresses the primary cause of hurricane related, cooperative distribution outages in the two recent hurricane seasons – tornadic winds, trees and flying debris.

Most Florida cooperatives have created generator programs for large and critical loads. In many cases it is less expensive for a cooperative to provide a permanent or portable backup generator during restoration, either on the customer's site or at a substation, than it is to harden a system.

Many cooperatives have also lowered the underground differential charge. This promotes the installation and use of underground facilities.

Some cooperatives are building ties between feeders to add redundancy to the system. This enhances reliability, avoids storm related outages and decreases storm restoration time.

In many cases cooperatives are using stronger poles and more expensive materials for targeted facilities. They have taken this action because the cooperatives' boards have determined that the increased cost is justified and the members are willing to pay higher associated rates.

On their own initiative, cooperatives have considered whether to adopt extreme wind loading standards. One cooperative, Florida Keys Electric Cooperative Association, Inc., has decided to adopt extreme wind loading standards, despite the associated cost. Other

cooperatives, such as Withlacoochee River Electric Cooperative, Inc., have considered the higher materials costs associated with adopting extreme wind loading standards and have targeted transmission facilities and feeders for upgrades to extreme wind loading standards but have declined to adopt such standards across the board.

Before proposing a prescriptive rule for cooperatives, the Commission should seriously consider whether such a rule, particularly one with high associated costs, is warranted. The democratically representative boards of Florida's cooperatives are uniquely qualified to evaluate and implement storm reliability and restoration measures. Their members expect the boards to act to diminish vulnerability to extreme weather events, and those boards have acted and will continue to act. Of course, it is those boards and not the Commission that also have rate making authority. So, they are better positioned than the Commission to consider the cost implications of each of the alternatives available. Thus, FECA respectfully submits that the Commission should think long and hard about proposing a prescriptive rule that imposes significant costs. If any rule is to be adopted for cooperatives, a rule much less prescriptive than the Commission proposed rule should be adopted.

V FECA'S PROPOSED ALTERNATIVE RULE

While FECA advocates that the Commission decline to adopt any rule for cooperatives, as an alternative, FECA is proposing a less prescriptive rule. FECA's proposed alternative rule is set forth in Attachment E. It abandons language in the Commission's proposed rule that requires cooperatives to adopt various standards, recognizing that such standards are already in place for RUS cooperatives. Instead, it creates requirements for certain standards to be made

available for Commission review. In addition, the rule contemplates an annual report to be submitted by each cooperative that addresses compliance with the NESC, pole inspections, vegetation management and other matters the cooperatives deems appropriate, including the extent to which facilities may be upgraded to extreme wind loading standards in the NESC. A section by section analysis follows.

Section (1) of FECA's proposed alternative Rule 25-6.0343 makes it clear that the rule is applicable only to those electric utilities as defined in Chapter 366, Florida Statutes, (municipal electric utilities and rural electric cooperatives) that provide distribution services to end use customers. It was FECA's understanding from discussions with the Commission Staff that the Commission's proposed rule was not intended to address generation and transmission cooperatives, only distribution cooperatives, so this was written into FECA's alternative rule as well.

Section (2) of FECA's proposed rule requires each municipal electric utility and rural electric cooperative serving end use customers to maintain at its corporate headquarters the following information: construction standards, pole inspection standards, vegetation management standards and guidelines, and procedures or methodologies for inspecting transmission structures and poles and distribution poles. These materials are to be readily available to the Commission Staff, and if Staff is unwilling to travel to review these materials, arrangements are to be made to provide Staff access to these materials in Tallahassee.

Section (3) of FECA's alternative rule requires the filing of an annual report with the Commission by March 1 of each year. The report would contain: (a) a statement of compliance with the NESC regarding construction standards (b) a statement of compliance with the NESC

regarding pole attachment contract; (b) a pole inspection report; (c) a vegetation management report; and (d) other appropriate information such as whether facilities were upgraded to meet extreme wind loading standards in the NESC.

FECA's proposed rule recognizes and addresses the many differences between IOUs, cooperatives and municipal utilities, including the differences between the organizational structures, the fiduciary duty of directors to consumers, and the jurisdiction of this Commission, the Federal Communications Commission's ("FCC") and the RUS. Cooperatives are not-forprofit, self-governing entities run by elected boards and commissions that serve at the will of the cooperative's member-owners. Every trustee must be a member of the cooperative, and they must be elected by the member-owners of the cooperative at the cooperative's annual meeting. See Section 425.10, F.S. As not-for-profit consumer controlled organizations, cooperatives do not have a conflicting profit incentive and they serve only one master, the consumer. The elected boards of cooperatives have a fiduciary duty to the cooperative and its member-owners to insure that the cooperative provides reliable service at a reasonable cost. In short, cooperatives' trustees assure distribution reliability; there is no need for the Commission to act to address such distribution reliability, whether storm related or in general. FECA's rule limits its scope to matters within the Commission's safety jurisdiction and calls for cooperatives and municipal's voluntary offering to make other matters available to the Commission and its Staff.

FECA's proposed rule stops short of the Commission mandating that cooperatives and municipal electric utilities adopt standards that go beyond safety standards and which address distribution reliability. So, this alternative proposed rule avoids the cooperatives and municipal electric utilities having to litigate the Commission's jurisdiction (or lack of jurisdiction) over cooperatives and municipal's distribution facility reliability.

VI PROBLEMS WITH PROPOSED RULE 25-6.0343

Proposed Rule 25-6.0343 is not based upon sound policy for cooperatives. It is nothing more than a mere restatement of the requirements of the rules proposed for the IOUs. The proposed rule completely disregards the dramatically different relationship between cooperatives and their members and IOUs and their ratepayers as well as the Commission's relationship to IOUs and cooperatives. It disregards cooperatives' unique cost characteristics, the high costs that would be imposed on cooperatives by the proposed rule and the fact that it is cooperatives' boards and not the Commission that has to balance customer service expectations with rate impacts. The proposed rule also fails to take into account the existing requirements of the RUS as well as the existing requirements of Commission rules that cooperatives comply with the NESC. Thus, it requires standards that are already in place and requires consideration of other standards not required by the RUS or necessary to meet the service expectations of cooperative members.

FECA especially takes issue with the Commission's attempt to resolve conflicts between the cooperative and its members, to define what is cost-effective for a cooperative, to require the use of the extreme wind loading standards, to define construction standards for cooperatives without regard to the existing contracts between cooperatives and their lenders, to require the placement of facilities adjacent to roadways, and to regulate pole attachments for cooperatives. While FECA's members share the Commission's goals of establishing and maintaining adequate construction standards and improving restoration times, FECA maintains that the Commission's rule must be restricted to subjects that are within its jurisdictional limits and must advance sound public policy.

In the following discussion, FECA addresses some of the specific flaws in the proposed rule. More detailed FECA comments are also reflected in Attachment F, on a section by section basis.

Subsection (1)(e)

Proposed subsection (1)(e) appears to require use of the extreme wind loading standards of the NESC for new distribution facilities unless there are extenuating circumstances, such as failing a cost-effectiveness test. However, there are no definitions in the rule for the terms "reasonably practical", "feasible" or "cost-effective". Under a purely monetary costeffectiveness test the extreme wind loading standards would never be implemented because they will always be more expensive than the minimum standards of the NESC. Presumably, there are unidentified factors that must be considered for this test, or else this provision would have no purpose other than to prevent the use of the extreme wind loading standards.

While FECA appreciates the fact that the rule appears to give great discretion to the utilities to determine what is cost-effective, feasible and reasonably practicable, cooperatives already have this discretion. Moreover, when the decision only involves distribution facilities that are for the exclusive use of the cooperative and its members, the Commission lacks authority to review the decision of a cooperative's board unless it is related to a territorial issue. FECA also is concerned that a strict application of the rule would be counterproductive to cooperatives that are building to a standard higher than the minimum.

It cannot be disputed that building to the extreme wind loading standards is more

expensive than building to the minimum standard. This has been discussed in detail above and is shown on Attachment B. In some cases the extreme wind loading standard would more than double construction costs for materials, possibly without providing any significant benefits. More importantly, there is no research or evidence in this record that supports a finding that use of the extreme wind loading standards is the best approach for cooperatives. As FECA demonstrated in its presentation to the Commission on June 5, many poles that were constructed to the extreme wind loading standards nevertheless failed due to tornadic wind and tree limbs during hurricanes Charley, Ivan and Wilma.

There are alternatives to improving system performance that may be more effective and cheaper for a cooperative than to double construction costs for infrastructure that may inevitably fail no matter how much is spent to reinforce it. The majority of cooperatives' pole failures in the hurricanes of 2004 and 2005 were the result of tornadic winds and trees falling into the lines or on poles. As explained above, FECA's members have all undertaken specific actions to improve their storm reliability. FECA respectfully submits that a cooperative's board is uniquely qualified to evaluate and implement these alternatives. Moreover, cooperative Boards are the exclusive entity to make rate decisions for their members. It is far better for the body charged with rate making to decide which storm reliability measures should be undertaken by cooperatives.

For some cooperatives moving to the extreme wind loading standards will result in substantial rate increases. While the Commission has rate structure jurisdiction over cooperatives, it does not have ratemaking jurisdiction. *City of Tallahassee v. Mann*, 411 So.2d 162 (Fla. 1981). Ratemaking falls exclusively within the discretion of each cooperative's

governing board, and mandating or imposing significant costs on an electric utility constitutes ratemaking or is inconsistent with the exercise of ratemaking authority. See, *Florida Power Corp. v. Seminole County*, 579 So.2d 105, 107 (Fla. 1991).

FECA's also is concerned that a strict application of the proposed rule could prohibit the use of construction standards that exceed the minimum standards of the NESC. The higher standards are more expensive, and arguably would not pass a cost effectiveness test unless factors other than cost are considered. At least two cooperatives are building all of their distribution facilities to a standard that exceeds the minimum criteria of the NESC. In both cases the cooperative's board determined that the higher construction standard was desired by their members and that the members were willing to pay higher rates for the higher standard. FECA believes that regardless of any tests set forth by the Commission, cooperative boards have the right to build to standards that exceeded the minimum loading criteria of the NESC, and the Commission is without jurisdiction to prevent such construction.

FECA is further concerned that the test set forth in this subsection may conflict with the standards imposed by RUS. Therefore, the Commission's proposed rule may impair a cooperative's contract with RUS.

Section (2)

Proposed subsection (2) appears to require distribution facilities to be placed adjacent to a public road and in front of the customer's premises unless there are extenuating circumstances, such as failing a cost-effectiveness test. There are no definitions in the rule for the terms "reasonably practical", "feasible" or "cost-effective". FECA appreciates the fact that the rule appears to give great discretion to the utilities to determine what is cost-effective, feasible and

reasonably practicable, but cooperatives already have this discretion. A cooperative's management and board are uniquely qualified to establish guidelines for the placement of facilities without rule mandated preferences from the Commission which fail to recognize legitimate alternatives that might be superior in individual circumstances.

A front-lot presumption should not apply in rural areas. In many cases the cooperative will construct lines across open fields because it is a significantly shorter and cheaper path to serve a new member. An alternative route along established roads would be significantly longer and therefore more expensive, and it probably would fail under the cost-effectiveness test. Nevertheless, the presumption in the rule that facilities should be placed adjacent to a public road is troubling and may unintentionally create a legal burden on cooperative boards that dare to place facilities in locations other than along roadways.

FECA also takes exception to the rule as it applies to commercial buildings. FECA agrees that in residential neighborhoods it usually is a good policy to place distribution facilities in the front of the building so that the equipment is more readily accessible (but even that preference is not universal, as there are instances where there is better or equal access to other sides of residential lots). However, commercial buildings are different. In some cases commercial properties have holding ponds and other obstructions in front of the building that would render the utility's facilities inaccessible by vehicles. In some cases it is advantageous to place a pad mounted transformer in the rear of a commercial building to avoid contact with vehicles that travel at high speeds. Arguably, these are extenuating circumstances that should allow the utility to avoid the presumptions in the rule for commercial properties, but the lack of definitions in the rule are cause for concern, and may create undesirable liability for cooperatives

and other utilities that chose to install facilities in a place that is not adjacent to a public road or in front of the premises.

Section (3)

Pole attachment rates for cooperatives and municipals are exempt from the FCC's rate, terms and conditions regulation. If an entity wishes to attach to cooperative facilities, they must pay the full cost of changes to our facilities that are required to maintain the minimum criteria set forth in the NESC. Cooperatives have contracts with entities that attach to their facilities, and RUS cooperatives attachment contracts require attachments to comply with the NESC. Section (3) of the proposed rule could result in the impairment of a cooperative's contracts with attachers and is absolutely unnecessary for cooperatives.

Section (4)

Proposed section (4) usurps the right of a cooperative to resolve disputes with its members. It also usurps the jurisdiction of the courts to resolve contract disputes and other cases between a cooperative and an attacher. These actions are clearly beyond the Commission's limited jurisdiction over cooperatives. In addition, it will be unnecessarily burdensome and costly for the cooperative's member and the cooperative if they are forced to travel to Tallahassee for a hearing on an issue that could have been resolved at home.

CONCLUSION

FECA respectfully submits no rule for cooperatives is warranted. Existing Commission rules and/or RUS requirements already sufficiently address cooperatives. As a second best alternative, FECA has suggested an alternative proposed rule. If the Commission decides to

proceed with the adoption of a rule for cooperatives, the proposed alternative rule attached hereto as Attachment A provides a least cost regulatory alternative to the Commission's proposed rule while also accomplishing all of the stated goals of the Commission's proposal. FECA respectfully requests that the Commission not adopt any rule for cooperatives, but that if the Commission decides to adopt a rule for cooperatives, the Commission adopt its alternative rule in lieu of proposed rule 25-6.0343.

Respectfully submitted,

Charles A. Guyton, Esq. Elizabeth C. Daley, Esq. Squire, Sanders & Dempsey L.L.P. 215 South Monroe St., Suite 601 Tallahassee, Florida 32301

Langs A Luft Charles A. Guyton By:

William B. Willingham, Esq. Michelle Hershel, Esq. Florida Electric Cooperatives Assoc., Inc. 2916 Apalachee Parkway Tallahassee, Florida 32301

CERTIFICATE OF SERVICE

I hereby certify that a true and correct copy of the foregoing Comments Of The Florida Electric Cooperatives Association, Inc. To Proposed Rule 25-6.0343 was furnished by Hand Delivery (*) or U.S. Mail this 8th day of September, 2006, to the following:

Lawrence Harris* Legal Division Florida Public Service Commission 2540 Shumard Oak Blvd. Tallahassee, FL 32399-0850

Trevor G. Underwood 2425 Sunrise Key Blvd Fort Lauderdale, FL 33314-3827

Florida Municipal Electric Association, Inc. Frederick M. Bryant Jody Lamar Finklea Post Office Box 3209 Tallahassee, FL 32315-3209 Florida Cable Telecommunications Association, Inc. Michael A. Gross 246 E. 6th Avenue Suite 100 Tallahassee, FL 32303

Lee County Electric Cooperative, Inc. Donald Schleicher William Hamilton P. O. Box 3455 North Fort Myers, FL 33918-3455

Charles A. Guyton

Attachment

ATTACHMENT A FECA'S ALTERNATIVE RULE

25-6.0343 Access to Standards of Municipal Electric Utilities and Rural Electric Cooperatives and Reporting of Pole Inspections and Vegetation Management

(1) Application and Scope. The purpose of this rule is to define certain reporting requirements by municipal electric utilities and rural electric cooperatives providing distribution service to end-use customers in Florida.

(2) Each municipal electric utility and rural electric cooperative shall maintain at its corporate headquarters a copy of its construction standards, pole attachment standards, vegetation management standards and the guidelines, procedures or methodologies for inspecting transmission structures and poles and distribution poles, including the pole inspection cycle and pole selection process information. Upon request, the utility shall provide access to a copy of these standards, guidelines, procedures and methodologies to the Commission staff at the utility's headquarters. If the Commission staff is unable to travel to the municipal's or cooperative's headquarters, arrangements will be made to provide access to the documents in Tallahassee. (3) Each utility shall submit a report to the Director of the Division of Economic Regulation by March 1 of each year for the preceding calendar year which shall include:

(a) A statement of whether the utility's current construction standardscomply with the applicable edition of the National Electrical Safety Code (ANSI C-2) [NESC].

(b) A statement of whether the utility contractually requires attachments by others to the utility's transmission and distribution facilities to comply with the applicable edition of the NESC.

(c) A pole inspection report which shall include information for the previous 12 months on the following:

- (1) The number and percentage of transmission structure and pole and distribution pole inspections planned and completed.
- (2) The number and percentage of transmission structures and poles and distribution poles failing the inspection and the cause for such failure, if known.
- (3) The number and percentage of transmission structures and poles and distribution poles replaced or for which remediation was taken, including a description of the remediation taken.

2

(d) A vegetation management report which shall describe the utility's vegetation management plan, including the percentage of the cycle completed for transmission, three-phase distribution, distribution secondary and lateral circuits in the previous 12-month period, if available.

(e) Any other information the utility deems appropriate, which may include facilities which were upgraded to the extreme wind loading standards specified by Figure 250-2(d) of the 2002 edition of the NESC.

History: New

Legislative Authority: 366.04(6)

Single Phase #2 AAAC						
NESC Code	250B	250C				
Pole Type	40/5 Wood	40/3 Wood				
Span Length						
(ft)	450	270				
	\$	\$				
Cost per Mile	36,694	60,378				

ATTACHMENT B EXTREME WIND LOADING COST COMPARISONS

3 Phase 394 AAAC Single Circuit					
NESC Code	250B	250C	250C		
			50/H2		
Pole Type	50/3 Wood	50/2 Wood	Steel		
Span Length					
(ft)	375	170	240		
	\$	\$	\$		
Cost per Mile	75,000	150,624	147,327		

3 Phase 740 AAAC Single Circuit					
NESC Code	250B	250C	250C		
1. 101 1. pl 18-18-1-plates in the spectrum is an examination of the spectrum is a second sec			50/H2		
Pole Type	50/3 Wood	50/2 Wood	Steel		
Span Length					
(ft)	300	140	200		
n an far a shekara a na far ann an far an shekara shekara shekara a na 2000 ka sa kara a	\$	\$	\$		
Cost per Mile	95,815	185,494	179,597		

3 Phase 394 AAAC Double Circuit				
NESC Code	250B	250C	250C	
			55/H3	
Pole Type	50/2 Wood	50/2 Wood	Steel	
Span Length				
(ft)	325	110	220	
	\$	\$	\$	
Cost per Mile	149,496	387,690	251,316	

3 Phase 740 AAAC Double Circuit				
NESC Code	250B	250C	250C	
			55/H4	
Pole Type	50/2 Wood	50/2 Wood	Steel	
Span Length				
(ft)	250	90	200	
	\$	\$	\$	
Cost per Mile	198,091	479,739	297,468	

ATTCHMENT C APPLICABLE RUS REGULATIONS



Committed to the future of rural communities.

Electric Programs

Rural Development / Utilities Programs / Electric / Telecommunications | Water & Environmental

Electric Home Page

About the Electric Programs

Frequently Asked Questions (FAQ's)

Interagency Electric Energy Market Competition Task Force

GIS Resources

Success Stories

Staff Directory

Loan Programs

Grant Programs

Interest Rates

Box Score

Cushion of Credit

List of Materials

Federal Register

Regulations

Bulletins

Engineering

Renewable Energy

Photovoltaic Systems

Environmental

Forms

Data Collection System (DCS)

Borrower Directory

Links

Recently Revised Pages

Electric Programs >> Regulations

Rural Utilities Service Electric Program Regulation:

Regulations Bulletins Codified or Incorporated by Reference Federal Register Issuances

The regulations are available in **html** format for viewing in your browser and in Adobe Acrobat **PDF** format for downloading and printing. The regulations below are from the Code of Federal Regulations (**CFR**) as of January 1, 2006. Amendments since that date are noted in the description.

The <u>E-CFR beta site</u>, maintained by the National Archives and Records Administration, provides a copy of the CFR which is updated daily.

Regulations

7 CFR Part	HTML	PDF	Description	
1709	html	pdf	Assistance to high energy cost rural communities	
1710	<u>html</u>	pdf	General and pre-loan policies and procedures common to insure and guaranteed electric loans	
1714	html	pdf	Pre-loan policies and procedures for insured electric loans	
1717	<u>html</u>	pdf	Post-loan policies and procedures common to insured and guaranteed electric loans	
1718	html	pdf	Loan security documents for electric borrowers	
1720	<u>html</u>	pdf	Guarantees for bonds and notes issued for electrification or telephone purposes	
1721	html	pdf	Post-loan policies and procedures for insured electric loans	
1724	<u>html</u>	pdf	Electric engineering, architectural services and design policies and procedures	
1726	<u>html</u>	pdf	Electric system construction policies and procedures	
1728	html	pdf	Electric standards and specifications for materials and construction.	
1730	<u>html</u>	pdf	Electric system operations and maintenance	
1767	html	pdf	Accounting requirements for RUS electric borrowers	
1773	<u>html</u>	<u>pdf</u>	Policy on audits of RUS borrowers	
1785	<u>html</u>	<u>pdf</u>	Loan account computations, procedures and policies for electric and telephone borrowers	

1786	<u>html</u>	pdf	Prepayment of RUS guaranteed and insured loans to electric an telephone borrowers	
1788	<u>html</u>	pdf	RUS fidelity and insurance requirements for electric and telephone borrowers	
1789	<u>html</u>	pdf	Use of consultants funded by borrowers	
1792	html	<u>pdf</u>	Compliance with other Federal statutes, regulations, and Executive orders	
1794	<u>html</u>	pdf	Environmental policies and procedures	

Bulletins Codified or Incorporated by Reference

Bulletin	Size (.doc)	Text	MS Word	PDF	Description
1728F-803	10.6M	N/A	N/A	<u>.pdf</u>	Specifications and Drawings for 24.9/14.4 kV Line Construction (Incorporated by reference - §1728.97). See letter dated March 7, 2001, letter concerning assembly numbering (<u>html</u>) (<u>pdf</u>).
1728F-804	12.8M	N/A	N/A	<u>.pdf</u>	Specifications and Drawings for 12.5/7.2 kV Line Construction (Incorporated by reference - §1728.97) (4/21/2005)
1728F-806	4.2M	N/A	N/A	<u>.pdf</u>	Specifications and Drawings for Underground Electric Distribution (Incorporated by reference - §1728.97)
1728F-700	4.1M	N/A	<u>.doc</u>	<u>.pdf</u>	Specification for Wood Poles, Stubs and Anchor Logs (Incorporated by reference - §1728.97)
1728H-701	30K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Specifications for wood crossarms, transmission timbers, and pole keys (Codified - §1728.201)
1728H-702	24K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Specifications for quality control and inspection of timber products (Codified - §1728.202)

For bulletins and informational publications, see the Electric Program Bulletins Page.

The free Adobe Adobat Reader is required to view PDF files. You may download it from: http://www.adobe.com/prodindex/acrobat/readstep.html

Electric Programs borrowers will be notified of new and revised regulations by memo or hard copy. Borrowers should notify their business associates of the availability of these regulations. All new and revised Electric Program regulations will be available here upon issuance. If you have any questions regarding these documents or documents not included here, call (202) 720-8674 or FAX (202) 205-3654.

For other Rural Development Utilities Programs regulations, visit the main <u>Utilities</u> <u>Programs Publications and Directives Page</u>.

E-mail suggestions and comments to the <u>Electric Programs Webmaster</u>. Please include your name, e-mail address, telephone number, and company affiliation in the body of your message so that we may be able to contact you for additional information, if necessary.

Perform a USDA wide Search For questions, contact the Electric Programs Webmaster Policies & Statements: <u>Nondiscrimination | Accessibility | Privacy Policy | Freedom of Information Act | Quality of Information</u>

§1721.107

begin with the next payment. For example: the amount deferred in the October payment will be reamortized over a 84 month period starting with the next payment (November if paying on a monthly basis). When a Borrower defers principal under any of these programs the scheduled payment on the account will increase by an amount sufficient to pay off the deferred amount, with interest, by the date specified in the agreement (usually 84 months (28 quarters)).

[67 FR 485, Jan. 4, 2002, as amended at 68 FR 37954, June 26, 2003]

§1721.107 Agreement.

After approval of the Borrower's request for a deferment of principal and interest, an extension agreement, containing the terms of the extension, together with associated materials, will be prepared and forwarded to the Borrower by RUS. The extension agreement will then be executed and returned to RUS by the Borrower.

§1721.108 Commencement of the deferment.

The deferment of principal and interest will not begin until the extension agreement and other supporting materials, in form and substance satisfactory to RUS, have been executed by the Borrower and returned to RUS. Examples of other supporting materials are items such as approving legal opinions from the Borrower's attorney and approvals from the relevant regulatory body for extending the maturity of existing debt and for the additional debt service payment incurred.

§1721.109 OMB control number.

The information collection requirements in this part are approved by the Office of Management and Budget and assigned OMB control number 0572-0123.

PART 1724—ELECTRIC ENGINEER-ING, ARCHITECTURAL SERVICES AND DESIGN POLICIES AND PROCEDURES

Subpart A—General

Sec. 1724.1 Introduction.

7 CFR Ch. XVII (1-1-06 Edition)

- 1724.2 Waivers.
- 1724.3 Definitions. 1724.4 Qualifications.
- 1724.5 Submission of documents to RUS.
- 1724.6 Insurance requirements.
- 1724.7 Debarment and suspension.
- 1724.8 Restrictions on lobbying.
- 1724.9 Environmental compliance.
- 1724.10 Standard forms of contracts for borrowers.

1724.11-1724.19 [Reserved]

Subpart B—Architectural Services

1724.20 Borrowers' requirements—architectural services.

1724.21 Architectural services contracts.

1724.22-1724.29 [Reserved]

Subpart C-Engineering Services

- 1724.30 Borrowers' requirements-engineering services.
- 1724.31 Engineering services contracts.
- 1724.32 Inspection and certification of work
- order construction. 1724.33-1724.39 [Reserved]

Subpart D—Electric System Planning

- 1724.40 General.
- 1724.41-1724.49 [Reserved]

Subpart E—Electric System Design

- 1724.50 Compliance with National Electrical Safety Code (NESC).
- 1724.51 Design requirements. 1724.52 Permitted deviations from RUS construction standards.
- 1724.53 Preparation of plans and specifications.
- 1724.54 Requirements for RUS approval of plans and specifications.
- 1724.55 Dam safety.
- 1724.56-1724.69 [Reserved]
- APPENDIX A TO SUBPART E—HAZARD POTEN-TIAL CLASSIFICATION FOR CIVIL WORKS PROJECTS

Subpart F-RUS Contract Forms

- 1724.70 Standard forms of contracts for borrowers.
- 1724.71 Borrower contractual obligations.
- 1724.72 Notice and publication of listed contract forms.
- 1724.73 Promulgation of new or revised contract forms.
- 1724.74 List of electric program standard contract forms.
- 1724.75-1724.99 [Reserved]
- AUTHORITY: 7 U.S.C. 901 et seq., 1921 et seq., 6941 et seq.
- SOURCE: 63 FR 35314, June 29, 1998, unless otherwise noted.

Rural Utilities Service, USDA

Subpart A-General

§1724.1 Introduction.

(a) The policies, procedures and requirements in this part implement certain provisions of the standard form of loan documents between the Rural Utilities Service (RUS) and its electric borrowers.

(b) All borrowers, regardless of the source of financing, shall comply with RUS' requirements with respect to design, construction standards, and the use of RUS accepted material on their electric systems.

(c) Borrowers are required to use RUS contract forms only if the facilities are financed by RUS.

§1724.2 Waivers.

The Administrator may waive, for good cause on a case-by-case basis, requirements and procedures of this part.

§1724.3 Definitions.

Terms used in this part have the meanings set forth in §1710.2 of this chapter. References to specific RUS forms and other RUS documents, and to specific sections or lines of such forms and documents, shall include the corresponding forms, documents, sections and lines in any subsequent revisions of these forms and documents. In addition to the terms defined in §1710.2 of this chapter, the following terms have the following meanings for the purposes of this part:

Architect means a registered or licensed person employed by the borrower to provide architectural services for a project and duly authorized assistants and representatives.

Engineer means a registered or licensed person, who may be a staff employee or an outside consultant, to provide engineering services and duly authorized assistants and representatives.

Force account construction means construction performed by the borrower's employees.

GPO means Government Printing Office.

NESC means the National Electrical Safety Code.

RE Act means the Rural Electrification Act of 1936 as amended (7 U.S.C. 901 et seq.).

Repowering means replacement of the steam generator or the prime mover or both at a generating plant. RUS means Rural Utilities Service.

RUS approval means written approval by the Administrator or a representative with delegated authority. RUS approval must be in writing, except in emergency situations where RUS approval may be given orally followed by a confirming letter.

RUS financed means financed or funded wholly or in part by a loan made or guaranteed by RUS, including concurrent supplemental loans required by §1710.110 of this chapter, loans to reimburse funds already expended by the borrower, and loans to replace interim financing.

[63 FR 35314, June 29, 1998, as amended at 63 FR 58284, Oct. 30, 1998]

§1724.4 Qualifications.

The borrower shall ensure that:

(a) All selected architects and engineers meet the applicable registration and licensing requirements of the States in which the facilities will be located:

(b) All selected architects and engineers are familiar with RUS standards and requirements; and

(c) All selected architects and engineers have had satisfactory experience with comparable work.

§1724.5 Submission of documents to RUS.

(a) Where to send documents. Documents required to be submitted to RUS under this part are to be sent to the office of the borrower's respective RUS Regional Director, the Power Supply Division Director, or such other office of RUS as designated by RUS. (See part 1700 of this chapter.)

(b) Contracts requiring RUS approval. The borrower shall submit to RUS three copies of each contract that is subject to RUS approval under subparts B and C of this part. At least one copy of each contract must be an original signed in ink (i.e., no facsimile signature). Each contract submittal must be accompanied by a certified copy of the board resolution awarding the contract.

(c) Contract amendments requiring RUS approval. The borrower shall submit to

§ 1724.6

RUS three copies of each contract amendment (at least one copy of which must be an original signed in ink) which is subject to RUS approval. Each contract amendment submittal to RUS must be accompanied by a certified copy of the board resolution approving the amendment.

§1724.6 Insurance requirements.

(a) Borrowers shall ensure that all architects and engineers working under contract with the borrower have insurance coverage as required by part 1788 of this chapter.

(b) Borrowers shall also ensure that all architects and engineers working under contract with the borrower have insurance coverage for Errors and Omissions (Professional Liability Insurance) in an amount at least as large as the amount of the architectural or engineering services contract but not less than \$500,000.

§1724.7 Debarment and suspension.

Borrowers shall comply with the requirements on debarment and suspension in connection with procurement activities as set forth in part 3017 of this title, particularly with respect to lower tier transactions, e.g., procurement contracts for goods or services.

§1724.8 Restrictions on lobbying.

Borrowers shall comply with the restrictions and requirements in connection with procurement activities as set forth in part 3018 of this title.

§1724.9 Environmental compliance.

Borrowers shall comply with the requirements of part 1794 of this chapter, Environmental Policies and Procedures for Electric and Telephone Borrowers.

§1724.10 Standard forms of contracts for borrowers.

The standard loan agreement between RUS and its borrowers provides that, in accordance with applicable RUS regulations in this chapter, the borrower shall use standard forms of contracts promulgated by RUS for construction, procurement, engineering services, and architectural services financed by a loan made or guaranteed by RUS. This part implements these provisions of the RUS loan agreement. 7 CFR Ch. XVII (1-1-06 Edition)

Subparts A through E of this part prescribe when and how borrowers are required to use RUS standard forms of contracts for engineering and architectural services. Subpart F of this part prescribes the procedures that RUS follows in promulgating standard contract forms and identifies those contract forms that borrowers are required to use for engineering and architectural services.

[63 FR 58284, Oct. 30, 1998]

§§ 1724.11-1724.19 [Reserved]

Subpart B—Architectural Services

§1724.20 Borrowers' requirements—architectural services.

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) Each borrower shall select a qualified architect to perform the architectural services required for the design and construction management of headquarters facilities. The selection of the architect is not subject to RUS approval unless specifically required by RUS on a case by case basis. Architect's qualification information need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

(b) The architect retained by the borrower shall not be an employee of the building supplier or contractor, except in cases where the building is prefabricated and pre-engineered.

(c) The architect's duties are those specified under the Architectural Services Contract and under subpart E of this part, and, as applicable, those duties assigned to the "engineer" for competitive procurement procedures in part 1726 of this chapter.

(d) If the facilities are RUS financed, the borrower shall submit or require the architect to submit one copy of each construction progress report to RUS upon request.

(e) Additional information concerning RUS requirements for electric borrowers' headquarters facilities are set forth in subpart E of this part. See also RUS Bulletin 1724E-400, Guide to Presentation of Building Plans and Specifications, for additional guidance.
This bulletin is available from Program Development and Regulatory Analysis, Rural Utilities Service, U.S. Department of Agriculture, Stop 1522, 1400 Independence Ave., SW., Washington, DC 20250-1522.

§1724.21 Architectural services contracts.

The provisions of this section apply only to RUS financed electric system facilities.

(a) RUS Form 220, Architectural Services Contract, must be used by electric borrowers when obtaining architectural services.

(b) The borrower shall ensure that the architect furnishes or obtains all architectural services related to the design and construction management of the facilities.

(c) Reasonable modifications or additions to the terms and conditions in the RUS contract form may be made to define the exact services needed for a specific undertaking. Such modifications or additions shall not relieve the architect or the borrower of the basic responsibilities required by the RUS contract form, and shall not alter any terms and conditions required by law. All substantive changes must be approved by RUS prior to execution of the contract.

(d) Architectural services contracts are not subject to RUS approval and need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

(e) Closeout. Upon completion of all services and obligations required under each architectural services contract, including, but not limited to, submission of final documents, the borrower must closeout that contract. The borrower shall obtain from the architect a final statement of cost, which must be supported by detailed information as appropriate. For example, out-of-pocket expense and per diem types of compensation should be listed separately with labor, transportation, etc., itemized for each service involving these types of compensation. RUS Form 284, Final Statement of Cost for Architectural Service, may be used. All computations of the compensation must be made in accordance with the terms of the architectural services contract. Closeout documents need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

§§1724.22-1724.29 [Reserved]

Subpart C—Engineering Services

§1724.30 Borrowers' requirements engineering services.

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) Each borrower shall select one or more qualified persons to perform the engineering services involved in the planning, design, and construction management of the system.

(b) Each borrower shall retain or employ one or more qualified engineers to inspect and certify all new construction in accordance with §1724.32. The engineer must not be the borrower's manager.

(c) The selection of the engineer is not subject to RUS approval unless specifically required by RUS on a case by case basis. Engineer's qualification information need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

(d) The engineer's duties are specified under the Engineering Services Contract and under part 1726 of this chapter. The borrower shall ensure that the engineer executes all certificates and other instruments pertaining to the engineering details required by RUS.

(e) Additional requirements related to appropriate seismic safety measures are contained in part 1792, subpart C, of this chapter, Seismic Safety of Federally Assisted New Building Construction.

(f) If the facilities are RUS financed, the borrower shall submit or require the engineer to submit one copy of each construction progress report to RUS upon RUS' request.

§1724.31 Engineering services contracts.

The provisions of this section apply only to RUS financed electric system facilities.

(a) RUS contract forms for engineering services shall be used. Reasonable

7 CFR Ch. XVII (1-1-06 Edition)

modifications or additions to the terms and conditions in the RUS contract form may be made to define the exact services needed for a specific undertaking. Any such modifications or additions shall not relieve the engineer or the borrower of the basic responsibilities required by the RUS contract form, and shall not alter any terms and conditions required by law. All substantive changes to the RUS contract form shall be approved by RUS prior to execution of the contract.

(b) RUS Form 236, Engineering Service Contract—Electric System Design and Construction, shall be used for all distribution, transmission, substation, and communications and control facilities. These contracts are not subject to RUS approval and need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

(c) RUS Form 211, Engineering Service Contract for the Design and Construction of a Generating Plant, shall be used for all new generating units and repowering of existing units. These contracts require RUS approval.

(d) Any amendments to RUS approved engineering services contracts require RUS approval.

(e) Closeout. Upon completion of all services and obligations required under each engineering services contract, in-cluding, but not limited to, submission of final documents, the borrower must closeout the contract. The borrower shall obtain from the engineer a completed final statement of engineering fees, which must be supported by detailed information as appropriate. RUS Form 234, Final Statement of Engineering Fee, may be used. All computations of the compensation shall be made in accordance with the terms of the engineering services contract. Closeout documents need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

§1724.32 Inspection and certification of work order construction.

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) The borrower shall ensure that all field inspection and related services are performed within 6 months of the completion of construction, and are performed by a licensed engineer, except that a subordinate of the licensed engineer may make the inspection, provided the following conditions are met:

(1) The inspection by the subordinate is satisfactory to the borrower;

(2) This practice is acceptable under applicable requirements of the States in which the facilities are located;

(3) The subordinate is experienced in making such inspections;

(4) The name of the person making the inspection is included in the certification; and

(5) The licensed engineer signs such certification which appears on the inventory of work orders.

(b) The inspection shall include a representative and sufficient amount of construction listed on each RUS Form 219, Inventory of Work Orders (or comparable form), being inspected to assure the engineer that the construction is acceptable. Each work order that was field inspected shall be indicated on RUS Form 219 (or comparable form.) The inspection services shall include, but not be limited to, the following:

(1) Determination that construction conforms to RUS specifications and standards and to the requirements of the National Electrical Safety Code (NESC), State codes, and local codes;

(2) Determination that the staking sheets or as-built drawings represent the construction completed and inspected:

(3) Preparation of a list of construction clean-up notes and staking sheet discrepancies to be furnished to the owner to permit correction of construction, staking sheets, other records, and work order inventories;

(4) Reinspection of construction corrected as a result of the engineer's report;

(5) Noting, initialing, and dating the staking or structure sheets or as-built drawings and noting the corresponding work order entry for line construction; and

(6) Noting, initialing, and dating the as-built drawings or sketches for generating plants, substations, and other major facilities.

§ 1724.32

(c) *Certification*. (1) The following certification must appear on all inventories of work orders:

I hereby certify that sufficient inspection has been made of the construction reported by this inventory to give me reasonable assurance that the construction complies with applicable specifications and standards and meets appropriate code requirements as to strength and safety. This certification is in accordance with acceptable engineering practice.

(2) A certification must also include the name of the inspector, name of the firm, signature of the licensed engineer, the engineer's State license number, and the date of signature.

§§1724.33-1724.39 [Reserved]

Subpart D—Electric System Planning

§1724.40 General.

Borrowers shall have ongoing, integrated planning to determine their short-term and long-term needs for plant additions, improvements, replacements, and retirements for their electric systems. The primary components of the planning system consist of long-range engineering plans and construction work plans. Long-range engineering plans identify plant investments required over a long-range period, 10 years or more. Construction work plans specify and document plant requirements for a shorter term, 2 to 4 years. Long-range engineering plans and construction work plans shall be in accordance with part 1710, subpart F, of this chapter. See also RUS Bulletins 1724D-101A, Electric System Long-Range Planning Guide, and 1724D-101B, System Planning Guide, Construction Work Plans, for additional guidance. These bulletins are available from Program Development and Regulatory Analysis, Rural Utilities Service, U.S. Department of Agriculture, Stop 1522, 1400 Independence Ave., SW., Washington, DC 20250-1522.

§ 1724.51

§§ 1724.41-1724.49 [Reserved]

Subpart E—Electric System Design

§1724.50 Compliance with National Electrical Safety Code (NESC).

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) A borrower shall ensure that its electric system, including all electric distribution, transmission, and generating facilities, is designed, constructed, operated, and maintained in accordance with all applicable provisions of the most current and accepted criteria of the National Electrical Safety Code (NESC) and all applicable and current electrical and safety requirements of any State or local governmental entity. Copies of the NESC may be obtained from the Institute of Electrical and Electronic Engineers, Inc., 445 Hoes Lane, Piscataway, NJ 08855. This requirement applies to the borrower's electric system regardless of the source of financing.

(b) Any electrical standard requirements established by RUS are in addition to, and not in substitution for or a modification of, the most current and accepted criteria of the NESC and any applicable electrical or safety requirements of any State or local governmental entity.

(c) Overhead distribution circuits shall be constructed with not less than the Grade C strength requirements as described in Section 26. Strength Requirements, of the NESC when subjected to the loads specified in NESC Section 25. Loadings for Grades B and C. Overhead transmission circuits shall be constructed with not less than the Grade B strength requirements as described in NESC Section 26.

§1724.51 Design requirements.

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) *Distribution*. All distribution facilities must conform to the applicable RUS construction standards and utilize RUS accepted materials.

§1724.51

(b) *Transmission lines*. (1) All transmission line design data must be approved by RUS.

(2) Design data consists of all significant design features, including, but not limited to, transmission line design data summary, general description of terrain, right-of-way calculations, discussion concerning conductor and structure selection, conductor sag and tension information, design clearances, span limitations due to clearances, galloping or conductor separation, design loads, structure strength limitations, insulator selection and design, guying requirements, and vibration considerations. For lines composed of steel or concrete poles, or steel towers, in which load information will be used to purchase the structures, the design data shall also include loading trees, structure configuration and selection, and a discussion concerning foundation selection.

(3) Line design data for uprating transmission lines to higher voltage levels or capacity must be approved by RUS.

(4) Transmission line design data which has received RUS approval in connection with a previous transmission line construction project for a particular borrower is considered approved by RUS for that borrower, provided that:

(i) The conditions on the project fall within the design data previously approved; and

(ii) No significant NESC revisions have occurred.

(c) *Substations.* (1) All substation design data must be approved by RUS.

(2) Design data consists of all significant design features, including, but not limited to, a discussion of site considerations, oil spill prevention measures, design considerations covering voltage, capacity, shielding, clearances, number of low and high voltage phases, major equipment, foundation design parameters, design loads for line support structures and the control house, seismic considerations, corrosion, grounding, protective relaying, and AC and DC auxiliary systems. Reference to applicable safety codes and construction standards are also to be included.

(3) Substation design data which has received RUS approval in connection

7 CFR Ch. XVII (1-1-06 Edition)

with a previous substation construction project for a particular borrower is considered approved by RUS for that borrower, provided that:

(i) The conditions on the project fall within the design data previously approved; and

(ii) No significant NESC revisions have occurred.

(d) Generating facilities. (1) This section covers all portions of a generating plant including plant buildings, the generator step-up transformer, and the transmission switchyard at a generating plant. Warehouses and equipment service buildings not associated with generation plants are covered under paragraph (e) of this section. Generation plant buildings must meet the requirements of paragraph (e)(1) of this section.

(2) For all new generation units and for all repowering projects, the design outline shall be approved by RUS, unless RUS determines that a design outline is not needed for a particular project.

(3) The design outline will include all significant design criteria. During the early stages of the project, RUS will, in consultation with the borrower and its consulting engineer, identify the specific items which are to be included in the design outline.

(e) *Headquarters*—(1) *Applicable laws.* The design and construction of headquarters facilities shall comply with all applicable Federal, State, and local laws and regulations, including, but not limited to:

(i) Section 504 of the Rehabilitation Act of 1973, (29 U.S.C. 794), which states that no qualified individual with a handicap shall, solely by reason of their handicap, be excluded from participation in, be denied the benefits of, or be subject to discrimination under any program or activity receiving Federal financial assistance. The Uniform Federal Accessibility Standards (41 CFR part 101-19, subpart 101-19.6, appendix A) are the applicable standards for all new or altered borrower buildings, regardless of the source of financing.

(ii) The Architectural Barriers Act of 1968 (42 U.S.C. 4151), which requires that buildings financed with Federal funds are designed and constructed to

be accessible to the physically handicapped.

(iii) The Earthquake Hazards Reduction Act of 1977 (42 U.S.C. 7701 *et seq.*), and Executive Order 12699, Seismic Safety of Federal and Federally Assisted or Regulated New Building Construction (3 CFR 1990 Comp., p. 269). Appropriate seismic safety provisions are required for new buildings for which RUS provides financial assistance. (See part 1792, subpart C, of this chapter.)

(2) The borrower shall provide evidence, satisfactory in form and substance to the Administrator, that each building will be designed and built in compliance with all Federal, State, and local requirements.

(f) Communications and control. (1) This section covers microwave and powerline carrier communications systems, load control, and supervisory control and data acquisition (SCADA) systems.

(2) The performance considerations for a new or replacement master system must be approved by RUS. A master system includes the main controller and related equipment at the main control point. Performance considerations include all major system features and their justification, including, but not limited to, the objectives of the system, the types of parameters to be controlled or monitored, the communication media, alternatives considered, and provisions for future needs.

§1724.52 Permitted deviations from RUS construction standards.

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) Structures for raptor protection. (1) RUS standard distribution line structures may not have the extra measure of protection needed in areas frequented by eagles and other large raptors to protect such birds from electric shock due to physical contact with energized wires. Where raptor protection in the design of overhead line structures is required by RUS; a Federal, State or local authority with permit or license authority over the proposed construction; or where the borrower voluntarily elects to comply with the recommendations of the U.S. Fish and Wildlife Service or State wildlife agency, borrowers are permitted to deviate from RUS construction standards, provided:

(i) Structures are designed and constructed in accordance with "Suggested Practices for Raptor Protection on Powerlines: The State of the Art in 1996" (Suggested Practices for Raptor Protection); and,

(ii) Structures are in accordance with the NESC and applicable State and local regulations.

(2) Any deviation from the RUS construction standards for the purpose of raptor protection, which is not in accordance with the Suggested Practices for Raptor Protection, must be approved by RUS prior to construction. Suggested Practices for Raptor Protection on Powerlines: The State of the Art in 1996," published by the Edison Electric Institute/Raptor Research Foundation, is hereby incorporated by reference. This incorporation by reference is approved by the Director of the Office of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies of this publication may be obtained from the Raptor Research Foundation, Inc., c/o Jim Fitzpatrick, Treasurer, Carpenter Nature Center, 12805 St. Croix Trail South, Hastings, Minnesota 55033. It is also available for inspection during normal business hours at RUS. Electric Staff Division, 1400 Independence Avenue, SW., Washington, DC, Room 1246-S, and at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, or http://www.archives.gov/ to: go federal register/

code_of_federal_regulations/

ibr__locations.html.

(b) Transformer neutral connections. Where it is necessary to separate the primary and secondary neutrals to provide the required electric service to a consumer, the RUS standard transformer secondary neutral connections may be modified in accordance with Rule 97D2 of the NESC.

(c) Lowering of neutral conductor on overhead distribution lines. (1) It is permissible to lower the neutral attachment on standard construction pole-

§1724.53

top assemblies an additional distance not exceeding two feet (0.6 m) for the purpose of economically meeting the clearance requirements of the NESC.

(2) It is permissible to lower the transformer and associated neutral attachment up to two feet (0.6 m) to provide adequate clearance between the cutouts and single-phase, conventional distribution transformers.

(3) It is permissible to lower the neutral attachment on standard construction pole-top assemblies an additional distance of up to six feet (2 m) for the purpose of performing construction and future line maintenance on these assemblies from bucket trucks designed for such work.

[63 FR 35314, June 29, 1998, as amended at 69 FR 18803, Apr. 9, 2004]

§1724.53 Preparation of plans and specifications.

The provisions of this section apply to all borrower electric system facilities regardless of the source of financing.

(a) *General.* (1) The borrower (acting through the engineer, if applicable) shall prepare plans and specifications that adequately represent the construction to be performed.

(2) Plans and specifications for distribution, transmission, or generating facilities must be based on a construction work plan (as amended, if applicable), engineering study or construction program which has been approved by RUS if financing for the facilities will at any time be requested from RUS.

(b) Composition of plans and specifications package. (1) Whether built by force account or contract, each set of plans and specifications must include:

(i) *Distribution lines.* Specifications and drawings, staking sheets, key map and appropriate detail maps;

(ii) *Transmission lines*. Specifications and drawings, transmission line design data manual, vicinity maps of the project, a one-line diagram, and plan and profile sheets;

(iii) Substations. Specifications and drawings, including a one-line diagram, plot and foundation plan, grounding plan, and plans and elevations of structure and equipment, as well as all other necessary construction drawings,

7 CFR Ch. XVII (1-1-06 Edition)

in sufficient detail to show phase spacing and ground clearances of live parts; (iv) *Headquarters*. Specifications and

drawings, including: (A) A plot plan showing the location of the proposed building plus paving

and site development; (B) A one line drawing (floor plan and elevation view), to scale, of the pro-

posed building with overall dimensions shown; and (C) An outline specification including

(C) An outline specification including materials to be used (type of frame, exterior finish, foundation, insulation, etc.); and

(v) Other facilities (e.g., generation and communications and control facilities). Specifications and drawings, as necessary and in sufficient detail to accurately define the scope and quality of work required.

(2) For contract work, the appropriate standard RUS construction contract form shall be used as required by part 1726 of this chapter.

§1724.54 Requirements for RUS approval of plans and specifications.

The provisions of this section apply only to RUS financed electric system facilities.

(a) For any contract subject to RUS approval in accordance with part 1726 of this chapter, the borrower shall obtain RUS approval of the plans and specifications, as part of the proposed bid package, prior to requesting bids. RUS may require approval of other plans and specifications on a case by case basis.

(b) Distribution lines. RUS approval of the plans and specifications for distribution line construction is not required if standard RUS drawings, specifications, RUS accepted material, and standard RUS contract forms (as required by part 1726 of this chapter) are used. Drawings, plans and specifications for nonstandard distribution construction must be submitted to RUS and receive approval prior to requesting bids on contracts or commencement of force account construction.

(c) Transmission lines. (1) Plans and specifications for transmission construction projects which are not based on RUS approved line design data or do not use RUS standard structures must

receive RUS approval prior to requesting bids on contracts or commencement of force account construction.

(2) Unless RUS approval is required by paragraph (a) of this section, plans and specifications for transmission construction which use previously approved design data and standard structures do not require RUS approval. Plans and specifications for related work, such as right-of-way clearing, equipment, and materials, do not require RUS approval unless required by paragraph (a) of this section.

(d) Substations. (1)(i) Plans and specifications for all new substations must receive RUS approval prior to requesting bids on contracts or commencement of force account construction, unless:

(A) The substation design has been previously approved by RUS; and

(B) No significant NESC revisions have occurred.

(ii) The borrower shall notify RUS in writing that a previously approved design will be used, including identification of the previously approved design.

(2) Unless RUS approval is required by paragraph (a) of this section, plans and specifications for substation modifications and for substations using previously approved designs do not require RUS approval.

(e) Generation facilities. (1) This paragraph (e) covers all portions of a generating plant including plant buildings, the generator step-up transformer, and the transmission switchyard at a generating plant. Warehouses and equipment service buildings not associated with generation plants are covered under paragraph (f) of this section.

(2) The borrower shall obtain RUS approval, prior to issuing invitations to bid, of the terms and conditions for all generating plant equipment or construction contracts which will cost \$1,500,000 or more. Unless RUS approval is required by paragraph (a) of this section, plans and specifications for generating plant equipment and construction do not require RUS approval.

(f) Headquarters buildings. (1) This paragraph (f) covers office buildings, warehouses, and equipment service buildings. Generating plant buildings are covered under paragraph (e) of this section.

(2) Unless RUS approval is required by paragraph (a) of this section, plans and specifications for headquarters buildings do not require RUS approval. The borrower shall submit two copies of RUS Form 740g, Application for Headquarters Facilities. This form is available from Program Development and Regulatory Analysis, Rural Utilities Service, United States Department of Agriculture, Stop 1522, 1400 Inde-pendence Ave., SW., Washington, DC 20250-1522. The application must show floor area and estimated cost breakdown between office building space and space for equipment warehousing and service facilities, and include a one line drawing (floor plan and elevation view), to scale, of the proposed building with overall dimensions shown. The information concerning the planned building may be included in the borrower's construction work plan in lieu of submitting it with the application. (See 7 CFR part 1710, subpart F.) Prior to issuing the plans and specifications for bid, the borrower shall also submit to RUS a statement, signed by the architect or engineer, that the building design meets the Uniform Federal Accessibility Standards (See §1724.51(e)(1)(i)).

(g) Communications and control facilities. (1) This paragraph (g) covers microwave and powerline carrier communications systems, load control, and supervisory control and data acquisition (SCADA) systems.

(2) The borrower shall obtain RUS approval, prior to issuing invitations to bid, of the terms and conditions for communications and control facilities contracts which will cost \$500,000 or more. Unless RUS approval is required by paragraph (a) of this section, plans and specifications for communications and control facilities do not require RUS approval.

(h) Terms and conditions include the RUS standard form of contract, general and special conditions, and any other non-technical provisions of the contract. Terms and conditions which have received RUS approval in connection with a previous contract for a particular borrower are considered approved by RUS for that borrower.

[63 FR 35314, June 29, 1998, as amended at 65 FR 63196, Oct. 23, 2000]

§ 1724.54

§1724.55

§1724.55 Dam safety.

(a) The provisions of this section apply only to RUS financed electric system facilities.

(1)(i) Any borrower that owns or operates a RUS financed dam must utilize the 'Federal Guidelines for Dam Safety,"(Guidelines), as applicable. A dam, as more fully defined in the Guidelines, is generally any artificial barrier which either:

(A) Is 25 feet (8 m) or more in height; or

(B) Has an impounding capacity at maximum water storage elevation of 55 acre-feet (68,000 m3) or more.

(ii) The"Federal Guidelines for Dam Safety,"FEMA 93, June, 1979, published by the Federal Emergency Management Agency (FEMA), is hereby incorporated by reference. This incorporation by reference is approved by the Director of the Office of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies of the"Federal Guidelines for Dam Safety"may be obtained from the Federal Emergency Management Agency, Mitigation Directorate, PO Box 2012, Jessup, MD 20794. It is also available for inspection during normal business hours at RUS, Electric Staff Division, 1400 Independence Avenue, SW., Washington, DC, Room 1246-S, and at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, or go to: http:// www.archives.gov/federal_register/ code_of_federal_regulations/

ibr locations.html.

 $(\overline{2})$ The borrower shall evaluate the hazard potential of its dams in accordance with Appendix E of the U.S. Army Corps of Engineers Engineering and Design Dam Safety Assurance Program, ER 1110-2-1155, July 31, 1995. A summary of the hazard potential criteria is included for information as Appendix A to this subpart. The U.S. Army Corps of Engineers Engineering and Design Dam Safety Assurance Program, ER 1110-2-1155, July 31, 1995, published by the United States Army Corps of Engineers, is hereby incorporated by reference. This incorporation by reference is approved by the Director of the Office of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR 7 CFR Ch. XVII (1-1-06 Edition)

part 51. Copies of the U.S. Army Corps of Engineers Engineering and Design Dam Safety Assurance Program may be obtained from the U. S. Army Corps of Engineers, Publications Depot, 2803 52nd Äve., Hyattsville, MD 20781. It is also available for inspection during normal business hours at RUS, Electric Staff Division, 1400 Independence Avenue, SW., Washington, DC, Room 1246-S, and at the National Archives and Records Administration (NARA), For information on the availability of this material at NARA, call 202-741-6030, or to: http://www.archives.gov/ go federal_register/ code_of_federal_regulations/

ibr locations.html.

(3) For high hazard potential dams, the borrower must obtain an independent review of the design and critical features of construction. The reviewer must have demonstrated experience in the design and construction of dams of a similar size and nature. The reviewer must be a qualified engineer not involved in the original design of the dam or a Federal or State agency responsible for dam safety. The re-viewer must be approved by RUS.

(4) The independent review of design must include, but not necessarily be limited to, plans, specifications, design calculations, subsurface investigation reports, hydrology reports, and redesigns which result from encountering unanticipated or unusual conditions during construction.

(5) The independent review of construction shall include:

(i) Foundation preparation and treatment. When the foundation has been excavated and exposed, and before critical structures such as earth embankments or concrete structures are placed thereon, the borrower shall require the reviewer to conduct an independent examination of the foundation to ensure that suitable foundation material has been reached and that the measures proposed for treatment of the foundation are adequate. This examination must extend to the preparation and treatment of the foundation for the abutments.

(ii) Fill placement. During initial placement of compacted fill materials, the borrower shall require the reviewer to conduct an independent examination

to ensure that the materials being used in the various zones are suitable and that the placement and compaction procedures being used by the contractor will result in a properly constructed embankment.

(6) If the reviewer disagrees with any aspect of the design or construction which could affect the safety of the dam, then the borrower must meet with the design engineer and the reviewer to resolve the disagreements.

(7) Emergency action plan. For high hazard potential dams, the borrower must develop an emergency action plan incorporating preplanned emergency measures to be taken prior to and following a potential dam failure. The plan should be coordinated with local government and other authorities involved with the public safety and be approved by the borrower's board of directors.

(b)(1) For more information and guidance, the following publications regarding dam safety are available from FEMA:

(i) "Emergency Action Planning Guidelines for Dams," FEMA 64.

Pt. 1724, Subpt. E, App. A

(ii)''Federal Guidelines for Earthquake Analysis and Design of Dams,''FEMA 65.

(iii)"Federal Guidelines for Selecting and Accommodating Inflow Design Floods for Dams,"FEMA 94.

(iv)"Dam Safety: An Owner's Guidance Manual,"FEMA 145, August, 1987.

(2) These publications may be obtained from the Federal Emergency Management Agency, Mitigation Directorate, PO Box 2012, Jessup, MD 20794.

[63 FR 35314, June 29, 1998, as amended at 69 FR 18803, Apr. 9, 2004]

§§1724.56-1724.69 [Reserved]

APPENDIX A TO SUBPART E OF PART 1724—HAZARD POTENTIAL CLASSI-FICATION FOR CIVIL WORKS PROJECTS

The source for this appendix is U.S. Army Corps of Engineers Engineering and Design Dam Safety Assurance Program, ER 1110-2-1155, Appendix E. Appendix E is available from the address listed in $\S1724.55(a)(2)$.

Category 1	Low	Significant	High
Direct Loss of Life ²	None expected (due to rural location with no permanent structures for human habitation)	Uncertain (rural location with few residences and only transient or industrial development)	Certain (one or more extensive residentia), commercial or industrial development).
ifeline Losses ³	No disruption of servicesrepairs are cosmetic or rapidly repairable damage	Disruption of essential facilities and access	Disruption of critical facilities and access.
Property Losses ⁴	Private agricultural lands, equipment and isolated buildings	Major public and private facilities	Extensive public and private facilities.
nvironmental Losses ⁵	Minimal Incremental damage	Major mitigation required	Extensive mitigation cost or impossible to mitigate.

NOTES: ¹Categories are based upon project performance and do not apply to individual structures within a project. ²Loss of life potential based upon inundation mapping of area downstream of the project. Analysis of loss of life potential should take into account the extent of development and associ-tics of the potential based upon inundation mapping of area downstream of the project. Analysis of loss of life potential should take into account the extent of development and associ-tics of the potential based upon inundation mapping the time. ated population at risk, time of flood wave travel and warning time.

³Indirect threats to life caused by the interruption of lifeline services due to project failure, or operation, i.e., direct loss of (or access to) critical medical facilities or loss of water or power

*Direct economic impact of value of property damages to project facilities and down stream property and indirect economic impact do us to project services, i.e., impact on navigation industry of the loss of a dam and navigation pool, or impact upon a community of the loss of water or power supply.
 *Environmental impact downstream caused by the incremental flood wave produced by the project failure, beyond which would normally be expected for the magnitude flood event under

a without project conditions.

4 CFR Ch. XVII (1-1-06 Edition)

Pť.

1724, Subpt. E, App.

⊳

.

ı

Subpart F—RUS Contract Forms

§1724.70 Standard forms of contracts for borrowers.

(a) General. The standard loan agreement between RUS and its borrowers provides that, in accordance with applicable RUS regulations in this chapter, the borrower shall use standard forms of contract promulgated by RUS for construction, procurement, engineering services, and architectural services financed by a loan made or guaranteed by RUS. (See section 5.16 of appendix A to subpart C of part 1718 of this chapter.) This subpart prescribes RUS procedures in promulgating electric program standard contract forms and identifies those forms that borrowers are required to use.

(b) Contract forms. RUS promulgates standard contract forms, identified in the List of Required Contract Forms, $\S1724.74(c)$, that borrowers are required to use in accordance with the provisions of this part. In addition, RUS promulgates standard contract forms identified in the List of Guidance Contract Forms contained in \$1724.74(c) that the borrowers may but are not required to use in the planning, design, and construction of their electric systems. Borrowers are not required to use these guidance contract forms in the absence of an agreement to do so.

[63 FR 58284, Oct. 30, 1998]

§1724.71 Borrower contractual obligations.

(a) Loan agreement. As a condition of a loan or loan guarantee under the RE Act, borrowers are normally required to enter into RUS loan agreements pursuant to which the borrower agrees to use RUS standard forms of contracts for construction, procurement, engineering services and architectural services financed in whole or in part by the RUS loan. Normally, this obligation is contained in section 5.16 of the loan contract. To comply with the provisions of the loan agreements as implemented by this part, borrowers must use those forms of contract (hereinafter sometimes called 'listed contract forms") identified in the List of Required Standard Contract Forms contained in §1724.74(c).

(b) Compliance. If a borrower is required by this part or by its loan agreement with RUS to use a listed standard form of contract, the borrower shall use the listed contract form in the format available from RUS, either paper or electronic format. Exact electronic reproduction is acceptable. The approved RUS standard forms of contract shall not be retyped, changed, modified, or altered in any manner not specifically authorized in this part or approved by RUS in writing on a case-bycase basis. Any modifications approved by RUS on a case-by-case basis must be clearly shown so as to indicate the modification difference from the standard form of contract.

(c) Amendment. Where a borrower has entered into a contract in the form required by this part, no change may be made in the terms of the contract, by amendment, waiver or otherwise, without the prior written approval of RUS.

(d) Waiver. RUS may waive for good cause, on a case by case basis, the requirements imposed on a borrower pursuant to this part. Borrowers seeking a waiver by RUS must provide RUS with a written request explaining the need for the waiver.

(e) Violations. A failure on the part of the borrower to use listed contracts as prescribed in this part is a violation of the terms of its loan agreement with RUS and RUS may exercise any and all remedies available under the terms of the agreement or otherwise.

[63 FR 58285, Oct. 30, 1998, as amended at 69 FR 7108, Feb. 13, 2004]

§1724.72 Notice and publication of listed contract forms.

(a) Notice. Upon initially entering into a loan agreement with RUS, borrowers will be provided with all listed contract forms. Thereafter, new or revised listed contract forms promulgated by RUS, including RUS approved exceptions and alternatives, will be sent by regular or electronic mail to the address of the borrower as identified in its loan agreement with RUS.

(b) Availability. Listed contract forms are published by RUS. Interested parties may obtain the forms from: Rural Utilities Service, Program Development and Regulatory Analysis, U.S. Department of Agriculture, Stop 1522, 1400 Independence Avenue, SW., Stop 1522, Washington, DC 20250-1522, telephone number (202) 720-8674. The list of contract forms can be found in \$1724.74(c), List of Required Contract Forms.

[63 FR 58285, Oct. 30, 1998]

§1724.73 Promulgation of new or revised contract forms.

RUS may, from time to time, undertake to promulgate new contract forms or revise or eliminate existing contract forms. In so doing, RUS shall publish notice of rulemaking in the FEDERAL REGISTER announcing, as appropriate, a revision in, or a proposal to amend §1724.74, List of Electric Program Standard Contract Forms. The amendment may change the existing identification of a listed contract form; for example, changing the issuance date of a listed contract form or by identifying a new required contract form. The notice of rulemaking will describe the new standard contract form or the substantive change in the listed contract form, as the case may be, and the issues involved. The standard contract form or relevant portions thereof may be appended to the supplementary information section of the notice of rulemaking. As appropriate, the notice of rulemaking shall provide an opportunity for interested persons to provide comments. A copy of each such FED-ERAL REGISTER document shall be sent by regular or electronic mail to all borrowers.

[63 FR 58285, Oct. 30, 1998]

§1724.74 List of electric program standard contract forms.

(a) General. The following is a list of RUS electric program standard contract forms for architectural and engineering services. Paragraph (c) of this section contains the list of required contract forms, *i.e.*, those forms of contracts that borrowers are required to use by the terms of their RUS loan agreements as implemented by the provisions of this part. Paragraph (d) of this section contains the list of guidance contract forms, *i.e.*, those forms of contracts provided as guidance to borrowers in the planning, design, and construction of their systems. All of 7 CFR Ch. XVII (1–1–06 Edition)

these forms are available from RUS. See \$1724.72(b) for availability of these forms.

(b) Issuance date. Where required by this part to use a standard form of contract in connection with RUS financing, the borrower shall use that form identified by issuance date in the List of Required Contract Forms in paragraph (c) of this section, as most recently published as of the date the borrower executes the contract.

(c) List of required contract forms. (1) RUS Form 211, Rev. 4-04, Engineering Service Contract for the Design and Construction of a Generating Plant. This form is used for engineering services for generating plant construction.

(2) RUS Form 220, Rev. 6-98, Architectural Services Contract. This form is used for architectural services for building construction.

(3) RUS Form 236, Rev. 6-98, Engineering Service Contract—Electric System Design and Construction. This form is used for engineering services for distribution, transmission, substation, and communications and control facilities.

(d) List of guidance contract forms. (1) RUS Form 179, Rev. 9-66, Architects and Engineers Qualifications. This form is used to document architects and engineers qualifications.

(2) RUS Form 215, Rev. 5-67, Engineering Service Contract—System Planning. This form is used for engineering services for system planning.

(3) RUS Form 234, Rev. 3-57, Final Statement of Engineering Fee. This form is used for the closeout of engineering services contracts.

(4) RUS Form 241, Rev. 3-56, Amendment of Engineering Service Contract. This form is used for amending engineering service contracts.

(5) RUS Form 244, Rev. 12-55, Engineering Service Contract—Special Services. This form is used for miscellaneous engineering services.

(6) RUS Form 258, Rev. 4-58, Amendment of Engineering Service Contract—Additional Project. This form is used for amending engineering service contracts to add an additional project.

(7) RUS Form 284, Rev. 4-72, Final Statement of Cost for Architectural Service. This form is used for the close-out of architectural services contracts.

(8) RUS Form 297, Rev. 12-55, Engineering Service Contract-Retainer for Consultation Service. This form is used for engineering services for consultation service on a retainer basis.

(9) RUS Form 459, Rev. 9-58, Engineering Service Contract-Power Study. This form is used for engineering services for power studies.

[63 FR 58285, Oct. 30, 1998, as amended at 65 FR 63196, Oct. 23, 2000; 69 FR 52595, Aug. 27, 20041

§§1724.75-1724.99 [Reserved]

ART 1726—ELECTRIC SYSTEM CONSTRUCTION POLICIES AND PART PROCEDURES

Subpart A-General

Sec.

- 1726.1-1726.9 [Reserved]
- 1726.10 Introduction. 1726.11 Purpose.
- 1726.12 Applicability. 1726.13 Waivers.
- 1726.14 Definitions.
- "Buy American". 1726.15
- 1726.16 Debarment and suspension.
- 1726.17 Restrictions on lobbying.
- 1726.18 Preloan contracting.
- 1726.19 Use of competitive procurement. 1726.20 Standards and specifications.
- 1726.21 New materials.
- 1726.22 Methods of construction.
- 1726.23 Qualification of bidders. 1726.24 Standard forms of contracts for bor-
- rowers
- 1726.25 Subcontracts. 1726.26 Interest on overdue accounts.
- 1726.27 Contractor's bonds.
- 1726.28-1726.34 [Reserved]
- 1726.35 Submission of documents to RUS.
- 1726.36 Documents subject to RUS approval. 1726.37 OMB control number.
- 1726.38-1726.49 [Reserved]

Subpart B-Distribution Facilities

1726.50 Distribution line materials and equipment.

1726.51 Distribution line construction.

1726.52-1726.74 [Reserved]

Subpart C-Substation and Transmission Facilities

- 1726.75 General
- 1726.76 Substation and transmission line materials and equipment.
- 1726.77 Substation and transmission line construction.
- 1726.78-1726.124 [Reserved]

Subpart D-Generation Facilities

1726.125 Generating plant facilities. 1726.126-1726.149 [Reserved]

Subpart E-Buildings

1726.150 Headquarters buildings.

1726.151-1726.174 [Reserved]

Subpart F-General Plant

- 1726.175 General plant materials.
- 1726.176 Communications and control facili-
- ties 1726.177-1726.199 [Reserved]

Subpart G-Procurement Procedures

- 1726.200 General requirements
- 1726.201 Formal competitive bidding.
- 1726.202 Informal competitive bidding.
- 1726.203 Multiparty negotiation.
- 1726.204 Multiparty unit price quotations.
- 1726.205 Multiparty lump sum quotations.
- 1726.206-1726.249 [Reserved]

Subpart H-Modifications to RUS Standard Contract Forms

- 1726.250 General.
- 1726.251 Prior approved contract modification related to price escalation on transmission equipment, generation equipment, and generation construction contracts
- 1726.252 Prior approved contract modification related to liability for special and consequential damages.
- 1726.253 Prior approved contract modification related to alternative bid provision for payment to contractor for bulk purchase of materials.
- 1726.254 [Reserved] 1726.255 Prior approved contract modifications related to indemnification.
- 1726.256-1726.299 [Reserved]

Subpart I-RUS Standard Forms

- 1726.300 Standard forms of contracts for borrowers.
- 1726.301 Borrower contractual obligations.
- 1726.302 Notice and publication of listed contract forms.
- 1726.303 Promulgation of new or revised contract forms.
- 1726.304 List of electric program standard contract forms
- 1726.305-1726.399 [Reserved]

Subpart J-Contract Closeout

- 1726.400 Final contract amendment.
- 1726.401 Material contract closeout.
- 1726.402 Equipment contract closeout.
- 1726.403 Project construction contract closeout.

Pt. 1726

(8) RUS Form 297, Rev. 12-55, Engineering Service Contract-Retainer for Consultation Service. This form is used for engineering services for consultation service on a retainer basis.

(9) RUS Form 459, Rev. 9-58, Engineering Service Contract Study. This form is used for engineer-

[63 FR 58285, Oct. 30, 1998, as amended at 65 FR 63196, Oct. 23, 2000; 69 FR 52595, Aug. 27, 20041

§§1724.75-1724.99 [Reserved]

1726-ELECTRIC **SYSTEM** PART CONSTRUCTION POLICIES AND PROCEDURES

Subpart A-General

Sec.

- 1726.1-1726.9 [Reserved]
- 1726.10 Introduction. 1726.11 Purpose.
- 1726.12 Applicability. 1726.13 Waivers.
- 1726.14 Definitions. 1726.15 "Buy American".
- 1726.16 Debarment and suspension.
- 1726.17 Restrictions on lobbying.
- 1726.18 Preloan contracting.
- 1726.19 Use of competitive procurement.
- 1726.20 Standards and specifications.
- 1726.21 New materials.
- 1726.22 Methods of construction.
- 1726.23 Qualification of bidders. 1726.24 Standard forms of contracts for bor-
- rowers.
- 1726.25 Subcontracts.
- 1726.26 Interest on overdue accounts.
- 1726.27 Contractor's bonds.
- 1726.28-1726.34 [Reserved]
- 1726.35 Submission of documents to RUS.
- 1726.36 Documents subject to RUS approval.
- 1726.37 OMB control number. 1726.38-1726.49 [Reserved]

Subpart B—Distribution Facilities

- 1726.50 Distribution line materials and equipment. 1726.51 Distribution line construction.
- 1726.52-1726.74 [Reserved]

Subpart C-Substation and Transmission Facilities

- 1726.75 General
- 1726.76 Substation and transmission line materials and equipment.
- 1726.77 Substation and transmission line construction.
- 1726.78-1726.124 [Reserved]

Subpart D-Generation Facilities

Pt. 1726

1726.125 Generating plant facilities. 1726.126-1726.149 [Reserved]

Subpart E-Buildings

1726.150 Headquarters buildings. 1726.151-1726.174 [Reserved]

Subpart F-General Plant

- 1726.175 General plant materials.
- 1726.176 Communications and control facili-
- ties. 1726.177-1726.199 [Reserved]

Subpart G—Procurement Procedures

- 1726.200 General requirements
- 1726.201 Formal competitive bidding.
- Informal competitive bidding. 1726.202 1726.203 Multiparty negotiation.
- 1726.204
- Multiparty unit price quotations. 1726.205 Multiparty lump sum quotations.
- 1726.206-1726.249 [Reserved]

Subpart H-Modifications to RUS Standard Contract Forms

- 1726.250 General.
- 1726.251 Prior approved contract modification related to price escalation on transmission equipment, generation equipment, and generation construction contracts.
- 1726.252 Prior approved contract modifica-tion related to liability for special and consequential damages. 1726.253 Prior approved contract modifica-
- tion related to alternative bid provision for payment to contractor for bulk purchase of materials.
- 1726.254 [Reserved] 1726.255 Prior approved contract modifications related to indemnification.
- 1726.256-1726.299 [Reserved]

Subpart I-RUS Standard Forms

- 1726.300 Standard forms of contracts for borrowers.
- 1726.301 Borrower contractual obligations. 1726.302 Notice and publication of listed
- contract forms.
- 1726.303 Promulgation of new or revised contract forms.
- 1726.304 List of electric program standard contract forms.
- 1726.305-1726.399 [Reserved]

271

Subpart J-Contract Closeout

- 1726.400 Final contract amendment.
- 1726,401 Material contract closeout.
- 1726.402 Equipment contract closeout
- 1726.403 Project construction contract closeout.

§§ 1726.1-1726.9

1726.404 Non-site specific construction contract closeout.
1726.405 Inventory of work orders (RUS)

Form 219).

AUTHORITY: 7 U.S.C. 901 et seq., 1921 et seq., 6941 et seq.

SOURCE: 60 FR 10155, Feb. 23, 1995, unless otherwise noted.

Subpart A—General

§§ 1726.1-1726.9 [Reserved]

§1726.10 Introduction.

The policies, procedures and requirements included in this part are intended to implement provisions of the standard form of loan documents between the Rural Utilities Service (RUS) and its electric borrowers. Unless prior written approval is received from RUS, borrowers are required to comply with RUS policies and procedures as a condition to RUS providing loans, loan guarantees, or reimbursement of general funds for the construction and improvement of electric facilities. Requirements relating to RUS approval of plans and specifications, duties and responsibilities of the engineer and architect, and engineering and architectural services contracts, are contained in other RUS regulations. The terms "RUS form", "RUS standard form", "RUS specification", "and RUS bulletin'' have the same meanings as the terms "REA form", "REA standard form", "REA specification", "and REA bulletin", respectively, unless otherwise noted.

§1726.11 Purpose.

Each borrower is responsible for the planning, design, construction, operation and maintenance of its electric system. RUS, as a secured lender, has a legitimate interest in accomplishing RUS's programmatic objectives, and in assuring that the costs of construction, materials, and equipment are reasonable and economical and that the property securing the loans is constructed adequately to serve the purposes for which it is intended.

§1726.12 Applicability.

The requirements of this part apply to the procurement of materials and equipment for use by electric bor-

7 CFR Ch. XVII (1-1-06 Edition)

rowers in their electric systems and to the construction of their electric systems if such materials, equipment, and construction are financed, in whole or in part, with loans made or guaranteed RUS, including reimbursable bν projects. In order for general fund expenditures for procurement or construction to be eligible for reimbursement from loan funds, the borrower must comply with the procedures required by this part. In the case of jointly owned projects, RUS will determine on a case by case basis the applicability of the requirements of this part.

§1726.13 Waivers.

The Administrator may waive, for good cause on a case by case basis, certain requirements and procedures of this part. RUS reserves the right, as a condition of providing loans, loan guarantees, or other assistance, to require any borrower to make any specification, contract, or contract amendment subject to the approval of the Administrator.

§1726.14 Definitions.

Terms used in this part have the meanings set forth in 7 CFR 1710.2. References to specific RUS forms and other RUS documents, and to specific sections or lines of such forms and documents, shall include the corresponding forms, documents, sections and lines in any subsequent revisions of these forms and documents. In addition to the terms defined in 7 CFR 1710.2, the following terms have the following meanings for the purposes of this part:

Approval of proposed construction means RUS approval of a construction work plan or other appropriate engineering study and RUS approval, for purposes of system financing, of the completion of all appropriate requirements of part 1794 of this chapter.

Architect means a registered or licensed person employed by the borrower to provide architectural services for a project and duly authorized assistants and representatives.

Bona fide bid means a bid which is submitted by a contractor on the borrower's list of qualified bidders for the specific contract, prior to bid opening.

"Buy American" certificate means a certification that the contractor has complied with the "Buy American" requirement (see § 1726.15).

Competitive procurement means procurement of goods or services based on lowest evaluated bid for similar products or services when three or more bids are received.

Construction unit means a specifically defined portion of a construction project containing materials, labor, or both, for purposes of bidding and payment.

Contracting committee means the committee consisting of three to five members representing the borrower's management and board of directors and the engineer. The contracting committee represents the borrower during contract clarifying discussions or negotiations under informal competitive bidding or multiparty negotiation, respectively.

Encumbrance means the process of approval for advance of loans funds by RUS.

Engineer means a registered or licensed person, who may be a staff employee or an outside consultant, to provide engineering services and duly authorized assistants and representatives.

Equipment means a major component of an electric system, e.g., a substation transformer, heat exchanger or a transmission structure.

Force account construction means construction performed by the borrower's employees.

Formal competitive bidding means the competitive procurement procedure wherein bidders submit sealed proposals for furnishing the goods or services stipulated in the specification. Bids are publicly opened and read at a predetermined time and place. If a contract is awarded, it must be to the lowest evaluated responsive bidder (see §1726.201).

Goods or services means materials, equipment, or construction, or any combination thereof.

Informal competitive bidding means the competitive procurement procedure which provides for private opening of bids and allows clarifying discussions between the contracting committee and the bidders. During the clarifying discussions any exceptions to the bid documents must be eliminated, or the bid rejected, so that the contract is awarded to the lowest evaluated responsive bidder (see § 1726.202).

Material means miscellaneous hardware which is combined with equipment to form an electric system, e.g., poles, insulators, or conductors.

Minor error or irregularity means a defect or variation in a bid that is a matter of form and not of substance. Errors or irregularities are "minor" if they can be corrected or waived without being prejudicial to other bidders and when they do not affect the price, quantity, quality, or timeliness of construction. A minor error or irregularity is not an exception for purposes of determining whether a bid is responsive.

Minor modification or improvement means a project where the cost is less than \$50,000, exclusive of the cost of owner furnished materials.

Multiparty lump sum quotations means the procurement of goods or services on a lump sum basis, based on the lowest evaluated offering, when three or more offers are received. (See §1726.205).

Multiparty negotiation means the procurement procedure where three or more bids are received and provides for negotiations between the contracting committee and each bidder to determine the bid which is in the borrower's best interest (see §1726.203).

Multiparty unit price quotations means the procurement of goods or services on a unit price basis, based on the lowest evaluated offering, when three or more offers are received (See § 1726.204).

Net utility plant (NUP) means Part C, Line 5 of RUS Form 7 for distribution borrowers or Section B, Line 5 of RUS Form 12a for power supply borrowers for the immediately preceding calendar year.

Procurement method means a procedure, including, but not limited to, those in subpart G of this part, that a borrower uses to obtain goods and services.

Owner furnished materials means materials or equipment or both supplied by the borrower for installation by the contractor.

§ 1726.15

Responsive bid means a bid with no exceptions or non-minor errors or irregularities on any technical requirement or in the contract terms and conditions.

RUS approval means written approval by the Administrator or a representative with delegated authority. RUS approval must be in writing, except in emergency situations where RUS approval may be given over the telephone followed by a confirming letter.

Unit prices means individual prices for specific construction units defined in accordance with RUS approved units specified in RUS standard contract forms.

§1726.15 "Buy American".

The borrower must ensure that all materials and equipment financed with loans made or guaranteed by RUS complies with the "Buy American" provisions of the Rural Electrification Act of 1938 (7 U.S.C. 903 note), as amended by the North American Free Trade Agreement Implementation Act (107 Stat 2129). When a "Buy American" certificate is required by this part, this must be on RUS Form 213.

§1726.16 Debarment and suspension.

Borrowers are required to comply with certain requirements on debarment and suspension in connection with procurement activities as set forth in part 3017 of this title, particularly with respect to lower tier transactions, e.g., procurement contracts for goods or services.

§1726.17 Restrictions on lobbying.

Borrowers are required to comply with certain restrictions and requirements in connection with procurement activities as set forth in part 3018 of this title.

§1726.18 Preloan contracting.

Borrowers must consult with RUS prior to entering into any contract for material, equipment, or construction if a construction work plan, general funds, loan or loan guarantee for the proposed work has not been approved. While the RUS staff will work with the borrower in such circumstances, nothing contained in this part is to be construed as authorizing borrowers to

7 CFR Ch. XVII (1-1-06 Edition)

enter into any contract before the availability of funds has been ascertained by the borrower and all the requirements of part 1794 of this chapter, Environmental Policies and Procedures for Electric and Telephone Borrowers, have been fulfilled.

§1726.19 Use of competitive procurement.

RUS borrowers' procurement is not subject to the provisions of the Federal Acquisition Regulation (48 CFR chapter 1); however, since borrowers receive the benefit of Federal financial assistance borrowers must use competitive procurement to the greatest extent practical. The borrower must use competitive procurement for obtaining all goods or services when a RUS loan or loan guarantee is involved except:

(a) As specifically provided for in subparts B through F of this part; or(b) A waiver is granted.

§1726.20 Standards and specifications.

All materials, equipment, and construction must meet the minimum requirements of all applicable RUS standards and specifications. (See part 1728 of this chapter, Electric Standards and Specifications for Materials and Construction, which is applicable regardless of the source of funding.)

[69 FR 7109, Feb. 13, 2004]

§1726.21 New materials.

The borrower shall purchase only new materials and equipment unless otherwise approved by RUS, on a case by case basis, prior to the purchase.

§1726.22 Methods of construction.

The borrower is generally responsible for determining whether construction will be by contract or force account. If construction is by contract, the borrower must determine whether materials will be supplied by the contractor or will be furnished by the borrower. RUS reserves the right to require contract construction in lieu of force account construction on a case by case basis.

§1726.23 Qualification of bidders.

(a) Qualified bidder list (QBL). The borrower shall (acting through its engineer, if applicable) review the qualifications of prospective bidders for contract construction and for material and equipment procurement, and select firms qualified for inclusion on the borrower's list of qualified bidders for each contract. (See also §1726.16 and §1726.17.) A bid may not be solicited from a prospective bidder or opened by the borrower unless that bidder has been determined to be a qualified bidder for the contract. When preparing the QBL, in addition to the actual experience of the borrower, if any, in dealing with a prospective bidder, the borrower may solicit information from that bidder or from other parties with firsthand experience regarding the firm's capabilities and experience. It is also important to consider the firm's performance record, safety record, and similar factors in determining whether to include that firm on the QBL, since the borrower may not evaluate these factors when evaluating a bid from a qualified and invited bidder.

(b) Conflict of interest. If there is a relationship between the borrower or engineer and a prospective bidder which might cause the borrower or engineer to have or appear to have a conflict of interest, that prospective bidder shall not be included on the QBL unless the engineer discloses the nature of the relationship to the borrower. In the case of the borrower, if its employees or directors have a relationship with a prospective bidder, the prospective bidder shall not be included on the qualified bidders list unless the nature of the relationship is disclosed to the board of directors, and the board of directors specifically approves the inclusion of that bidder in light of the potential for a conflict of interest.

§1726.24 Standard forms of contracts for borrowers.

(a) General. The standard loan agreement between RUS and the borrowers provides that, in accordance with applicable RUS regulations in this chapter, the borrower shall use standard forms of contracts promulgated by RUS for construction, procurement, engineering services, and architectural services financed by a loan made or guaranteed by RUS. This part implements these provisions of the RUS loan agreement. Subparts A through H and J of this part prescribe when and how borrowers are required to use RUS standard forms of contracts in procurement and construction. Subpart I of this part prescribes the procedures that RUS follows in promulgating standard contract forms and identifies those contract forms that borrowers are required to use for procurement and construction.

(b) Amendments to contracts—(1) Contract forms. The borrower must use RUS Form 238, Construction or Equipment Contract Amendment, for any change or addition in any contract for construction or equipment.

(2) Special considerations. Each time an amendment to a construction contract is executed, the borrower must ensure that contractor's bond is adequate, that all necessary licenses and permits have been obtained, and that any environmental requirements associated with the proposed construction have been met.

 (3) Amendment approval requirements.
 (i) If a RUS approved form of contract is required by this part, an amendment must not alter the terms and conditions of the RUS approved form of contract without prior RUS approval.

(ii) The borrower must make a contract amendment subject to RUS approval if the underlying contract was made subject to RUS approval and the total amended contract price exceeds 120 percent of the original contract price (excluding any escalation provision contained in the contract).

(iii) Contract amendments, except as provided in paragraph (b)(3)(ii) of this section, are not subject to RUS approval and need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

[60 FR 10155, Feb. 23, 1995, as amended at 63 FR 58286, Oct. 30, 1998; 69 FR 7109, Feb. 13, 2004]

§1726.25 Subcontracts.

Subcontracts are not subject to RUS approval and need not be submitted to

§ 1726.26

RUS unless specifically requested by RUS on a case by case basis.

[69 FR 7109, Feb. 13, 2004]

§1726.26 Interest on overdue accounts.

Certain RUS contract forms contain a provision concerning payment of interest on overdue accounts. Prior to issuing the invitation to bidders, the borrower must insert an interest rate equal to the lowest "Prime Rate" listed in the "Money Rates" section of the Wall Street Journal on the date such invitation to bid is issued. If no prime rate is published on that date, the last such rate published prior to that date must be used. The rate must not, however, exceed the maximum rate allowed by any applicable state law.

[63 FR 58286, Oct. 30, 1998]

§1726.27 Contractor's bonds.

(a) RUS Form 168b, Contractor's Bond, shall be used when a contractor's bond is required by RUS Forms 200, 257, 786, 790, or 830 unless the contractor's surety has accepted a Small Business Administration guarantee and the contract is for \$1 million or less.

(b) RUS Form 168c, Contractor's Bond, shall be used when a contractor's bond is required by RUS Forms 200, 257, 786, 790, or 830 and the contractor's surety has accepted a Small Business Administration guarantee and the contract is for \$1 million or less.

(c) Surety companies providing contractor's bonds shall be listed as acceptable sureties in the U.S. Department of the Treasury Circular No. 570, Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies. Copies of the circular and interim changes may be obtained directly from the Government Printing Office (202) 512-1800. Interim changes are published in the FEDERAL REGISTER as they occur. The list is also available through the Internet at http:// www.fms.treas.gov/c570/index.html and on the Department of the Treasury's computerized public bulletin board at (202) 874-6887.

[63 FR 58286, Oct. 30, 1998, as amended at 69 FR 7109, Feb. 13, 2004]

7 CFR Ch. XVII (1-1-06 Edition)

§§1726.28-1726.34 [Reserved]

§1726.35 Submission of documents to RUS.

(a) Where to send documents. Documents required to be submitted to RUS under this part are to be sent to the office of the borrower's respective RUS Regional Director, the Power Supply Division Director, or such other office of RUS as designated by RUS (see part 1700 of this chapter.)

(b) Borrower certification. When a borrower certification is required by this part, it must be made by the borrower's manager unless the board of directors specifically authorizes another person to make the required certification. In such case, a certified copy of the specific authorizing resolution must accompany the document or be on file with RUS.

(c) Contracts requiring RUS approval. The borrower shall submit to RUS three copies of each contract that is subject to RUS approval under subparts B through F of this part. At least one copy of each contract must be an original signed in ink (i.e., no facsimile signature). Each contract submittal must be accompanied by:

(1) A bid tabulation and evaluation and, if applicable, a written recommendation of the architect or engineer.

(2) For awards made under the informal competitive bidding procedure or the multiparty negotiation procedure, a written recommendation of the contracting committee (See §§ 1726.202 and 1726.203).

(3) Three copies of an executed contractor's bond on RUS approved bond forms as required in the contract form (at least one copy of which must be an original signed in ink) and one copy of the bid bond or facsimile of the certified check.

(4) A certification by the borrower or chairperson of the contracting committee, as applicable, that the appropriate bidding procedures were followed as required by this part.

(5) A certified copy of the board resolution awarding the contract.

(6) Evidence of clear title to the site for substations and headquarters construction contracts, if not previously submitted.

(7) Documentation that all reasonable measures were taken to assure competition if fewer than three bids were received.

(d) Contract amendments requiring RUS approval. The borrower must submit to RUS three copies of each contract amendment (at least one copy of which must be an original signed in ink) which is subject to RUS approval under $\S1726.24$ (b). Each contract amendment submittal to RUS must be accompanied by:

(1) A certified copy of the board resolution approving the amendment; and

(2) A bond extension, where necessary.

(e) Encumbrance of loan or loan guarantee funds. (1) For contracts subject to RUS approval, the submittals required under paragraph (c) of this section will initiate RUS action to encumber loan or loan guarantee funds for such contracts.

(2) For contracts not subject to RUS approval (except for generation projects), loan or loan guarantee funds will normally be encumbered using RUS Form 219, Inventory of Work Orders, after closeout of the contracts. In cases where the borrower can show good cause for a need for immediate cash, the borrower may request encumbrance of loan or loan guarantee funds based on submittal of a copy of the executed contract, provided it meets all applicable RUS requirements.

(3) For generation project contracts not subject to RUS approval, the borrower must submit to RUS the following documentation:

(i) \overline{A} brief description of the scope of the contract, including contract identification (name, number, etc.);

(ii) Contract date:

(iii) Contractor's name;

(iv) Contract amount;

(v) Bidding procedure used:

(vi) Borrower certification that:

(A) The board of directors approved the contract;

(B) The bidding procedures and contract award for each contract were in conformance with the requirements of Part 1726, Electric System Construction Policies and Procedures;

(C) If a RUS approved form of contract is required by this part, the terms and conditions of the RUS approved form of contract have not been altered;

(D) If RUS has approved plans and specifications for the contract, the contract was awarded on the basis of those plans and specifications; and

(E) No restriction has been placed on the borrower's right to assign the contract to RUS or its successors.

(4) Contract amendments. (i) For amendments subject to RUS approval, the submittals required under paragraph (c) of this section will initiate RUS action to encumber loan or loan guarantee funds for contract amendments requiring RUS approval.

(ii) For amendments not subject to RUS approval (except generation projects), loan or loan guarantee funds will normally be encumbered using RUS Form 219, Inventory of Work Orders, after closeout of the contracts. In cases where the borrower can justify a need for immediate cash, the borrower may request encumbrance of loan or loan guarantee funds based on submittal of a copy of the executed amendment, providing it meets all applicable RUS requirements.

(iii) For each generation project contract amendment not subject to RUS approval, the borrower must submit to RUS the following information and documentation:

(A) The contract name and number;

(B) The amendment number;

(C) The amendment date;

(D) The dollar amount of the increase or the decrease of the amendment;

(E) Borrower certification that:

(1) The amendment was approved in accordance with the policy of the board of directors (the borrower must ensure that RUS has a certified copy of the board resolution establishing such policy);

(2) If a RUS approved form of contract is required by this part, the terms and conditions of the RUS approved form of contract has not been altered; and

(3) No restriction has been placed on the borrower's right to assign the contract to RUS or its successors.

§1726.36 Documents subject to RUS approval.

Unless otherwise indicated, the borrower shall make all contracts and

7 CFR Ch. XVII (1-1-06 Edition)

§ 1726.37

amendments that are subject to RUS approval effective only upon RUS approval.

§1726.37 OMB control number.

The collection of information requirements in this part have been approved by the Office of Management and Budget and assigned OMB control number 0572-0107.

§§ 1726.38~1726.49 [Reserved]

Subpart B—Distribution Facilities

§1726.50 Distribution line materials and equipment.

(a) *Contract forms.* (1) The borrower shall use RUS Form 198, Equipment Contract, for purchases of equipment where the total cost of the contract is \$500,000 or more.

(2) The borrower may, in its discretion, use RUS Form 198, Equipment Contract, or a written purchase order for purchases of equipment of less than \$500,000 and for all materials.

(b) Standards and specifications. Distribution line materials and equipment must meet the minimum requirements of RUS standards as determined in accordance with the provisions of part 1728 of this chapter, Electric Standards and Specifications for Materials and Construction. The borrower must obtain RUS approval prior to purchasing any unlisted distribution line material or equipment of the types listed in accordance with the provisions of part 1728 of this chapter.

(c) Procurement procedures. It is the responsibility of each borrower to determine the procurement method that best meets its needs for the purchase of material and equipment to be used in distribution line construction.

(d) Contract approval. Contracts for purchases of distribution line materials and equipment are not subject to RUS approval and need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

[60 FR 10155, Feb. 23, 1995, as amended at 69 FR 7109, Feb. 13, 2004]

§1726.51 Distribution line construction.

(a) *Contract forms.* The borrower must use RUS Form 790, or 830, as outlined

in this paragraph (a), for distribution line construction, except for minor modifications or improvements.

(1) The borrower may use RUS Form 790, Electric System Construction Contract—Non-Site Specific Construction, under the following circumstances:

(i) For contracts for which the borrower supplies all materials and equipment; or

(ii) For non-site specific construction contracts accounted for under the work order procedure; or

(iii) If neither paragraph (a)(1)(i) or (a)(1)(i) of this section are applicable, the borrower may use RUS Form 790 for contracts, up to a cumulative total of \$250,000 or one percent of net utility plant (NUP), whichever is greater, per calendar year of distribution line construction, exclusive of the cost of owner furnished materials and equipment.

(2) The borrower must use RUS Form 830, Electric System Construction Contract—Project Construction, for all other distribution line construction.

(b) Procurement procedures. (1) It is the responsibility of each borrower to determine the procurement method that best meets its needs to award contracts in amounts of up to a cumulative total of \$250,000 or one percent of NUP, whichever is greater, per calendar year of distribution line construction (including minor modifications or improvements), exclusive of the cost of owner furnished materials and equipment.

(2) In addition to the cumulative total stipulated in paragraph (b)(1) of this section, a borrower may use Multiparty Unit Price Quotations to award contracts in amounts of up to a cumulative total of \$350,000 or 1.5 percent of NUP, whichever is greater, per calendar year of distribution line construction (including minor modifications or improvements), exclusive of the cost of owner furnished materials and equipment.

(3) The borrower shall use formal competitive bidding for all other distribution line contract construction. The amount of contracts bid using the formal competitive bidding procedure do not apply to the cumulative total stipulated in paragraph (b)(1) of this section.

(4) An amendment which increases the scope of the contract by adding a project is not considered competitively bid, therefore, the amount of that amendment does apply to the cumulative total stipulated in paragraph (b)(1) of this section.

(c) Contract approval. Contracts for distribution line construction are not subject to RUS approval and need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

[60 FR 10155, Feb. 23, 1995, as amended at 69 FR 7109, Feb. 13, 2004]

§§1726.52-1726.74 [Reserved]

Subpart C—Substation and Transmission Facilities

§1726.75 General.

As used in this part, "substations" includes substations, switching stations, metering points, and similar facilities.

§ 1726.76 Substation and transmission line materials and equipment.

(a) Contract forms. (1) The borrower must use RUS Form 198, Equipment Contract, for purchases of equipment where the total cost of the contract is \$500,000 or more.

(2) The borrower may, in its discretion, use RUS Form 198, Equipment Contract, or a written purchase order for purchases of equipment of less than \$500,000 and for all materials.

(b) Standards and specifications. Substation and transmission line materials and equipment must meet the minimum requirements of RUS standards as determined in accordance with the provisions of part 1728 of this chapter, Electric Standards and Specifications for Materials and Construction. The borrower must obtain RUS approval prior to purchasing of any unlisted substation or transmission line material or equipment of the types listed in accordance with the provisions of part 1728 of this chapter.

(c) Procurement procedures. It is the responsibility of each borrower to determine the procurement method that best meets its needs for purchase of material and equipment to be used in substation and transmission line construction.

(d) Contract approval. Contracts for purchases of substation and transmission line materials and equipment are not subject to RUS approval and need not be submitted to RUS unless specifically requested by RUS on a case by case basis.

[60 FR 10155, Feb. 23, 1995, as amended at 69 FR 7109, Feb. 13, 2004]

§1726.77 Substation and transmission line construction.

(a) *Contract forms.* The borrower must use RUS Form 830, Electric System Construction Contract—Project Construction, for construction of substations, except for minor modifications or improvements.

(b) Procurement procedures. (1) It is the responsibility of each borrower to determine the procurement method that best meets its needs to award contracts not requiring RUS approval in amounts of up to a cumulative total of \$250,000 or one percent of NUP (not to exceed \$2,000,000), whichever is greater, per calendar year of substation and transmission line construction (including minor modifications or improvements), exclusive of the cost of owner furnished materials and equipment.

(2) The borrower shall use formal competitive bidding for all other contract construction, including all contracts requiring RUS approval. The amount of contracts bid using the formal competitive bidding procedure do not apply to the cumulative total stipulated in paragraph (b)(1) of this section.

(3) An amendment which increases the scope of the contract by adding a project is not considered competitively bid, therefore, the amount of that amendment does apply to the cumulative total stipulated in paragraph (b)(1) of this section.

(c) Contract approval. Individual contracts in amounts of \$250,000 or more or one percent of NUP (not to exceed \$500,000 for distribution borrowers or \$1,500,000 for power supply borrowers), whichever is greater, exclusive of the cost of owner furnished materials and

RUS approval and the closeout documents need not be sent to RUS unless specifically requested by RUS.

[60 FR 10155, Feb. 23, 1995, as amended at 69 FR 7111, Feb. 13, 2004]

§1726.405 Inventory of work orders (RUS Form 219).

Upon completion of the contract closeout, the borrower shall complete RUS Form 219, Inventory of Work Orders, in accordance with part 1717, Post-Loan Policies and Procedures Common to Insured and Guaranteed Electric Loans, of this chapter.

PART 1728—ELECTRIC STANDARDS AND SPECIFICATIONS FOR MA-TERIALS AND CONSTRUCTION

Sec.

- 1728.10 General purpose and scope.
- 1728.20 Establishment of standards and
- specifications. 1728.30 Inclusion of an item for listing or
- technical acceptance. 1728.40 Procedure for submission of a proposal.
- 1728.50 Removal of an item from listing or technical acceptance.
- 1728.60 List of materials and equipment.
- 1728.70 Procurement of materials.
- 1728.97 Incorporation by reference of electric standards and specifications.
- 1728.201 RUS Bulletin 1728H-701, RUS Specification for Wood Crossarms (Solid and Laminated), Transmission Timbers and Pole Keys.
- 1728.202 RUS Bulletin 1728H-702, RUS Specification for Quality Control and Inspection of Timber Products.

AUTHORITY: 7 U.S.C. 901 et seq., 1921 et seq., 6941 et seq.

§1728.10 General purpose and scope.

(a) The requirements of this part are based on contractual provisions between RUS and the organizations which receive financial assistance from RUS.

(b) RUS will establish certain specifications and standards for materials, equipment, and construction units that will be acceptable for RUS financial assistance for the electric program. Materials and equipment purchased by the electric borrowers or accepted as contractor-furnished material must conform to RUS standards and specifications where they have been established and, if included in RUS Bulletin 43-5, "List of Materials Acceptable for Use on Systems of RUS Electrification Borrowers" (List of Materials), must be selected from that list or must have received technical acceptance from RUS. RUS, through its Technical Standards Committees, will evaluate certain materials, equipment and construction units, and will determine acceptance.

[50 FR 47710, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990]

§1728.20 Establishment of standards and specifications.

(a) National and other standards. RUS will utilize standards of national standardizing groups, such as the American National Standards Institute (ANSI), American Wood Preservers' Association (AWPA), the various national engineering societies and the National Electrical Safety Code (NESC), to the greatest extent practical. When there are no national standards or when RUS determines that the existing national standards are not adequate for rural electric systems, RUS will prepare standards for material and equipment to be used on systems of electric borrowers. RUS standards and specifications will be codified or listed in §1728.97, Incorporation by Reference of Electric Standards and Specifications. RUS will also prepare specifications for materials and equipment when it determines that such specifications will result in reduced costs, improved materials and equipment, or in the more effective use of engineering services.

(b) Deviations from Standards. No member of the RUS staff will be permitted to authorize deviations from the standard specifications, or to establish or change the technical standards, or to authorize the use of items that have not received acceptance by the Technical Standards Committees, except as provided for under § 1728.70, or by authorization and/or delegation of authority by the Administrator of RUS.

(c) Category of Items. Items appearing in the List of Materials are listed by categories of generic items which are used in RUS construction standards in corporated by reference in §1728.97. RUS will establish and define these

§1728.20

7 CFR Ch. XVII (1-1-06 Edition)

§1728.30

categories and will establish all criteria for acceptability within these categories.

[50 FR 47710, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990, and amended at 55 FR 53487, Dec. 31, 1990]

§1728.30 Inclusion of an item for listing or technical acceptance.

(a) *Scope*. RUS, through its Technical Standards Committees "A" and "B" will determine the acceptability of certain standards, standard specifications, standard drawings, and items of materials and equipment to be used in transmission, distribution and general plant (excluding office equipment, tools, and work equipment, and consumer-owned electric wiring facilities).

(b) Addresses of Committees. The address of Technical Standards Committee "A" is: Chairman, Technical Standards Committee "A" (Electric), Rural Utilities Service, U.S. Department of Agriculture, Washington, DC 20250-1500. The address of Technical Standards Committee "B" is: Chairman, Technical Standards Committee "B" (Electric), Rural Utilities Service, U.S. Department of Agriculture, Washington, DC 20250-1500.

(c) Review by Technical Standards Committee ''A''. All proposals for listing a product in the List of Materials must be addressed to Technical Standards Committee "A." This committee will consider all proposals made by sponsors of specifications, drawings, materials, or equipment in categories for which RUS has established criteria for acceptability. A sponsor may be a manufacturer, supplier, contractor or any other person or organization which has made an application for listing or has requested an action by the committee. Committee "A" will consider all relevant information presented in determining whether an item should be accepted by Technical Standards Com-mittee "A." Formal rules of evidence and procedure shall not apply to proceedings before this committee.

(d) Action by Technical Standards Committee ''A''. (1) Committee ''A'' may take one of the following actions:

(i) Accept an item for listing without conditions (domestic items only),

(ii) Reject an item (domestic or non-domestic),¹

(iii) Accept an item for listing with conditions (domestic items only),

(iv) Table an item for a time period sufficient to allow the sponsor to be notified and furnish additional information (domestic or nondomestic),

(v) Grant technical acceptance with or without conditions for a period of one year from the date of notification by RUS (nondomestic items only).

(2) All committee decisions regarding the actions listed above must be unanimous. If the vote is not unanimous, the item shall be referred to Technical Standards Committee "B." Written notice of Technical Standards Committee "A's" decision, stating the basis for the decision, will be provided to the sponsor.

(3) Items accepted without conditions by the Technical Standards Committees will be considered to be accepted on a general basis. No restrictions as to quantity or application will be placed on items which have received general acceptance. Items accepted subject to certain conditions, such as limited use to gain service experience, or limited use appropriate to certain areas and conditions, will be considered to be accepted on a conditional basis. The conditions will be cited as a part of the listing provided for in §1728.60, or as part of the technical acceptance for nondomestic items.

(e) Appeal to Technical Standards Committee "B". A sponsor may request a review of an adverse decision by Technical Standards Committee "A" within ten (10) days of notification of such decision by submitting a letter requesting such review to Technical Standards Committee "B" (Electric).

(f) Action by Technical Standards Committee "B". Committee "B" may take any of the actions listed for Committee "A" in §1728.30(d). However, for a Committee "B" action to be effective it must be by majority vote. Failure to obtain a majority on one of the proposed actions shall mean that the product will not be listed or accepted. Committee "B's" determination shall be

¹Nondomestic items are items which do not qualify as domestic products pursuant to RUS "Buy American" requirement.

based on the record developed before Committee "A" and such additional information as Committee "B" may request. Formal rules of procedure and evidence shall not apply to proceedings before Committee "B." Written notice of Committee "B's" decision, stating the basis of the decision, will be provided to the sponsor.

(g) Appeal to the Administrator. In the event of an adverse decision by Committee "B," the sponsor may, within ten (10) days of notification of such decision, request a review of this decision by submitting a letter to the Administrator requesting such a review.

(h) Change in Design. RUS acceptance of an item will be conditioned on the understanding that no design changes (material or dimensions) affecting the quality, strength, or electrical characteristics of the item shall be made without prior concurrence of Technical Standards Committee "A."

[50 FR 47711, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990]

§1728.40 Procedure for submission of a proposal.

(a) Written Request. Consideration of an item of material or equipment will be obtained by the sponsor through the submission of a written request in an original and five copies addressed to the Chairman, Technical Standards Committee "A" (Electric). The letter must include the catalog number or other identifying number or code as well as a description of the item. In the event that an item being submitted is also intended for consideration by Technical Standards Committee "A" (Telephone), a separate request must be made to the telephone committee. (See part 1755 of this chapter).

(b) Technical and Performance Data. Six copies of the specification of manufacture, drawings and test data must be submitted to the committee. Six copies of the performance history shall also be submitted unless RUS determines that such performance history is not reasonably available.

(c) Sample. One sample of the item must be submitted to the Chairman, Technical Standards Committee "A." unless RUS waives the requirements of the sample. In case of large, bulky or extremely heavy samples, the sponsor should contact the Chairman, Technical Standards Committee "A" (Electric), at the above address, before any sample is shipped.

(d) Action on Proposal. RUS will inform a sponsor of the action taken on the sponsor's proposal.

[50 FR 47711, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990]

§1728.50 Removal of an item from listing or technical acceptance.

(a) *Removal Actions.* An item of material or equipment may be removed from the listing or technical acceptance in accordance with the following procedures upon determination that the item is unsatisfactory or has been misrepresented to the owner or RUS.

(b) Notification by the Committee. The sponsor of an item of material or equipment will be notified in writing of a proposal to remove such item from the listing or technical acceptance.

(c) Supplemental Information. Within ten (10) days of receipt of such notification, the sponsor may submit to Committee "A" a letter expressing the sponsor's intent to submit written supplemental technical information relevant to Committee "A's" determination. The sponsor must submit such information within twenty (20) days from the submission of its letter to Committee "A." Committee "A" will have the discretion of making a decision following the expiration of the time periods provided in this paragraph.

(d) Review by the Technical Standards Committee "A". Committee "A" will consider all relevant information presented in determining whether an item should be removed from the listing or technical acceptance. Formal rules of evidence and procedure shall not apply to proceedings before Technical Standards Committee "A."

(e) Action by the Technical Standards Committee "A". Committee "A" may take one of the following actions:

(1) Order the immediate removal of the item from the listing, or technical acceptance,

(2) Condition the item's continued listing, or technical acceptance,

(3) Recommend a basis of settlement which will adequately protect the interest of the Government, or

§1728.50

§ 1728.60

(4) Delay the effectiveness of its decision for a time period sufficient to allow the sponsor to appeal to Technical Standards Committee "B."

All committee "A" decisions regarding the actions listed above must be by unanimous vote. If the vote is not unanimous, the item will be referred to Technical Standards Committee "B."

Written notice of Technical Standards Committee "A's" decision, stating the basis for the decision, will be provided to the sponsor.

(f) Additional Opportunity to Present Information. At the request of the sponsor, RUS may afford additional opportunity for consideration of relevant information. Such additional opportunity may include, without limitation, a meeting between RUS and the sponsor in such a forum that RUS may determine. In making this decision, RUS will consider, among other things, the best interests of RUS, its borrowers, and the sponsor, and the best manner to develop sufficient information relating to the proposed action.

(g) Appeal to the Technical Standards Committee "B". Within ten (10) days of notification of Committee "A's" decision, a sponsor may appeal in writing to Technical Standards Committee "B" to review Committee "A's" decision, specifying the reasons for such a request. Committee "B's" determination, in response to such request, shall be based on the record developed before Committee "A" and such additional information as Committee "B" may request. Formal rules of procedure and evidence shall not apply to proceedings before Committee "B."

(h) Action by Technical Standards Committee "B". Committee "B," by majority vote, may take one of the following actions:

(1) Order the immediate removal of the item from listing, or technical acceptance,

(2) Condition the item's continued listing, or technical acceptance,

(3) Recommend a basis of settlement which adequately protects the interests of the Government, or

(4) Delay the effectiveness of its decision for a time period sufficient to allow the sponsor to appeal to the Administrator of RUS.

7 CFR Ch. XVII (1-1-06 Edition)

Failure to obtain a majority vote on any of the above actions shall mean that the product will continue to be listed or accepted.

Written notice of Committee "B's" decision stating the basis of the decision will be provided to the sponsor.

(i) Appeal to the Administrator. Within ten (10) days of the receipt of Com-mittee "B's" decision, a sponsor may appeal to the Administrator to review Committee "B's" decision. If an appeal is made, the sponsor shall submit a written request to the Administrator, Rural Utilities Service, Room 4053, South Building, U.S. Department of Agriculture, Washington, DC 20250-1500 specifying the reasons to request reconsideration. The Administrator will have the option to decline the request, in which case the decision of Committee "B" shall stand. If a review is granted, the determination by the Administrator or the Administrator's designee shall be based on the record developed before Committee "A" and Committee "B" and such additional information as the Administrator may request. Formal rules of procedure and evidence shall not apply to the actions of the Administrator.

(j) Action by the Administrator. The Administrator may take one of the following actions:

(1) Order the immediate removal of the item from the listing, or technical acceptance,

(2) Condition its continued listing, or technical acceptance, or

(3) Recommend a basis of settlement which adequately protects the interests of the Government.

Written notice of the Administrator's determination, stating the basis for the decision, will be provided to the sponsor.

The Administrator's actions are final.

[50 FR 47711, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990]

§1728.60 List of materials and equipment.

(a) General. Those items of material or equipment accepted by Technical Standards Committee "A" or "B," with the exception of technically accepted nondomestic items, will be listed in the List of Materials. Items

which do not qualify as domestic products may be accepted on a technical basis only (technical acceptance) for a period of one year as provided in $\S1728.30(c)(1)$ and will not be included in the List of Materials.

(b) Publishing and Revisions. RUS will reissue the List of Materials every year, dated July, and issue supplements, if needed, dated October, January, and April of every year. An RUS office copy, which is the official current copy, of the List of Materials, will be updated every time changes are made by the Technical Standards Committees.

(c) Dual Listings. RUS, through its Technical Standards Committees, will accept for listing only one item of a particular type of material or equipment for each manufacturer. If a manufacturer submits an item to perform the identical function of a listed item, RUS, through its Technical Standards Committees, may accept that item and remove the one previously listed. RUS will list only new items of material and equipment in the List of Materials. Used items will not be considered for listing.

[50 FR 47712, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990]

§1728.70 Procurement of materials.

(a) By Owner. When purchasing the type of materials included in the List of Materials, RUS borrowers shall purchase only materials listed in the List of Materials, or materials which have a current technical acceptance by RUS and meet the "Buy American" requirement.

(b) By Contractor. When performing work for an RUS borrower, contractors shall supply only items from the general acceptance pages of the List of Materials, or obtain the borrower's concurrence prior to purchase and use of a technically nondomestic item or any item listed on a conditional basis.

(c) Procurement of Unlisted Items. (1) The borrower shall request prior approval from RUS for use of an item that does not fall in categories established by RUS in the List of Materials for which acceptability has been established by the Technical Standards Committees. (2) RUS will also determine, on a case-by-case basis, whether to allow use of an unlisted item in emergency situations and for experimental use or to meet a specific need. For purposes of this part 1728, an emergency shall mean a situation wherein the supply of listed material and equipment from the industry is not readily available, or the standard designs are not applicable to the borrower's specific problem under consideration.

(3) RUS will make arrangements for test or experimental use of newly developed items requiring limited trial use. RUS, working with the borrower and the manufacturer, will establish test locations for the items to facilitate installation and observation.

[50 FR 47712, Nov. 20, 1985. Redesignated at 55 FR 39395, Sept. 27, 1990]

§1728.97 Incorporation by reference of electric standards and specifications.

(a) The following electric bulletins have been approved for incorporation by reference by the Director of the Office of the Federal Register. The bulletins containing construction standards (50-4 and 1728F-803 to 1728F-811), may be purchased from the Superintendent of Documents, U.S. Government Printing Office, Washington, DC 20402. The bulletins containing specifications for materials and equipment (50-15 to 50-99 and 1728F-700) may be obtained from the Rural Utilities Service, Program Development and Regulatory Analysis, Stop 1522, Room 4028-S, Washington, DC 20250-1522. The terms "RUS form", "RUS standard form", "RUS specification", and "RUS bulletin" have the same meanings as the terms "REA form", "REA standard form", "REA specification", and "REA bulletin", respectively unless otherwise indicated. The bulletins are available for inspection at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, or go to: http:// www.archives.gov/federal_register/

code_of_federal_regulations/ ibr_locations.html. These materials are incorporated as they exist on the date of the approval and a notice of any

§1728.97

§ 1728.201

change in these materials will be published in the FEDERAL REGISTER. (b) List of Bulletins.

- Bulletin 50-4 (D-801), Specification and Drawings for 34.5/19.9 kV Distribution Line Construction (11-86)
- Bulletin 50-15 (DT-3), RUS Specifications for Pole Top Pins with 1%' Diameter Lead Thread (1-51)
- Bulletin 50-16 (DT-4), RUS Specifications for Angle Suspension Brackets (3-52)
- Bulletin 50-19 (DT-7), RUS Specifications for Clevis Bolts (8-53)
- Bulletin 50-23 (DT-18), RUS Specifications for 60" Wood Crossarm Braces (2-71)
- Bulletin 50-31 (D-3), RUS Specifications for Pole Top Pins with 1" Diameter Lead Threads (2-79)
- Bulletin 50-32 (D-4), RUS Specifications for Steel Crossarm Mounted Pins with 1" Diameter Lead Threads (10-50)
- Bulletin 50-33 (D-5), RUS Specifications for Single and Double Upset Spool Bolts (2-51)
- Bulletin 50-34 (D-6), RUS Specifications for Secondary Swinging Clevises (12-70)
- Bulletin 50-35 (D-7), RUS Specifications for
- Service Swinging Clevises (9-52) Bulletin 50-36 (D-8), RUS Specifications for Service Deadend Clevises (9-52) Bulletin 50-40 (D-14), RUS Specifications for
- Pole Top Brackets for Channel Type Pins (9-51)
- Bulletin 50-41 (D-15), RUS Specifications for
- Service Wireholders (11-51) Bulletin 50-55 (T-2), RUS Specifications for Overhead Ground Wire Support Brackets (5-53)
- Bulletin 50-56 (T-3), RUS Specifications for Steel Plate Anchors for Transmission Lines (12-53)
- Bulletin 50-60 (T-9), RUS Specification-Single Pole Steel Structures, Complete with Arms (12-71)
- Bulletin 50-70 (U-1), RUS Specification for 15 kV and 25 kV Primary Underground Power Cable (12-22-87)
- Bulletin 50-72 (U-4), RUS Specification for Electrical Equipment Enclosures (5-35 kV) (10-79)
- Bulletin 50-73 (U-5), RUS Specifications for Pad-Mounted Transformers (Single and Three-Phase) (1-77)
- Bulletin 50-74 (U-6), RUS Specification for Secondary Pedestals (600 Volts and Below) (10-79)
- Bulletin 50-91 (S-3), RUS Specifications for Step-Down Distribution Substation Transformers (34.4-138 kV) (1-78)
- Bulletin 1728F-700, RUS Specification for Wood Poles, Stubs and Anchor Logs (8-93). Bulletin 1728F-803, Specifications and Draw-ings for 24.9/14.4 kV Line Construction (10-
- 98). Bulletin 1728F-804 (D-804), Specification and Drawings for 12.47/7.2 kV Line Construction October 2005.

7 CFR Ch. XVII (1-1-06 Edition)

- Bulletin 1728F-806 (D-806), Specifications and Drawings for Underground Electric Distribution, June 2000.
- Bulletin 1728F-810, Electric Transmission Specifications and Drawings, 34.5 kV to 69 kV (3-98).
- Bulletin 1728F-811, Electric Transmission Specifications and Drawings, 115 kV to 230 kV (3-98).

[48 FR 31853, July 12, 1983, as amended at 52 FR 22289, June 11, 1987; 52 FR 48799, Dec. 28, 1987; 53 FR 39229, Oct. 6, 1988; 53 FR 44176, Nov. 2, 1988; 55 FR 8909, Mar. 9, 1990. Redesignated at 55 FR 39395, Sept. 27, 1990, as amended at 56 FR 1563, Jan. 16, 1991; 58 FR 41398. Aug. 3, 1993; 59 FR 66440, Dec. 27, 1994; 63 FR 11591, Mar. 10, 1998; 63 FR 72104, Dec. 31, 1998; 65 FR 34047, May 26, 2000; 69 FR 18803, Apr. 9, 2004; 70 FR 20703, Apr. 21, 2005]

§1728.201 RUS Bulletin 1728H-701. RUS Specification for Wood Cross-arms (Solid and Laminated), Transmission Timbers and Pole Keys.

(a) General provisions. (1) This section implements contractual provisions between RUS and borrowers receiving financial assistance from RUS. The contractual agreement between RUS and its borrowers requires the borrower's system to be constructed in accordance with RUS accepted plans and specifications. Each RUS electric borrower must purchase only wood crossarms produced in accordance with the specification in this section.

(2) Each RUS electric borrower shall require each contractor to agree in writing to furnish only materials produced in accordance with the specification in this section.

(3) This specification describes the minimum acceptable quality of wood distribution crossarms and transmission crossarms (hereinafter called crossarms) that are purchased by or for RUS borrowers. Where there is conflict between this specification and any other specification referred to in this section, this specification shall govern.

(4) Various requirements relating to quality control and inspection are contained in §1728.202 of this part, RUS Specification for Quality Control and Inspection of Timber Products. Section 1728.201 of this part and the American National Standards Institute (ANSI) 05.2, 1983, American National Standard for Wood Products-Structural Glued Laminated Timber for Utility Structures, shall be followed exactly and

shall not be interpreted or subjected to judgment by the quality control person or an independent inspector.

(5) The borrower shall purchase from producers only material that meets the requirements of this specification. Each purchaser shall use a written purchase order to purchase material for use in RUS financed systems in order to insure compliance with the standards and specifications of this part. The written purchase order shall contain a provision that specifically requires the producer to comply with the provisions of this part. The purchase order shall contain a provision that specifically requires the producer to make the treating plant, and storage areas available, during normal business hours, in order for representatives of either the purchaser or RUS to inspect such to determine compliance with the standards and specifications of this part.

(6) The borrower shall insure that the producer provides the inspectors with full information (drawings, etc.) relating to the requirements contained in purchase order which is supplementary to this specification.

(7) The borrower shall insure that the producer maintains, or has access to, adequate laboratory facilities at or very near the treating plant. All chemical tests, assays or analyses associated with the treatment shall be independently performed in this laboratory by both the quality control designee and the borrower's inspector. If acceptable to RUS on a case-by-case basis, the producer may use a central laboratory.

 $(\bar{8})$ Inspection and treatment of all timber products produced under this specification should be performed after receipt of the order from the purchaser, except as provided for reserve treated stock.

(9) The borrower shall insure that each inspection agency maintains its own central laboratory with qualified staff capable of completely analyzing the preservative and treatments. If acceptable to RUS, this central laboratory may be used for the independent inspector's routine assays, with results made available the next working day.

(10) The testing and inspection of the lamination process shall be in accord-

ance with American Institute of Timber Construction (AITC) 200-83, Inspection Manual.

(11) With the exception of reserve treated stock, all invoices for treated timber products shall be accompanied, in duplicate, by a copy of the producer's Certificate of Compliance and a copy of either the Independent Inspection Report or a Quality Assurance Plan Certificate. The certificate shall be presented to the purchaser with the invoice. For reserve treated stock, inspection reports shall be available from the inspection agency. When shipped from reserve stock, the invoice shall bear an endorsement and a further certification by the producer that the material meets the requirements of this specification and any supplementary requirements cited in the purchase order under which it is purchased.

(12) Crossarms shall be warranted to conform to this specification. If any crossarm is determined to be defective or does not conform to this specification within 1 year after shipment to the borrower, it shall be replaced as promptly as possible by the producer. In the event of failure to do so, the purchaser may make such replacement and the cost of the crossarm, at destination, recoverable from the producer.

(b) Definitions.

Arm refers to structural wood member used to support electrical conductors.

Certificate of compliance is a certification by an authorized employee of the producer that the material shipped meets the requirements of this specification and any supplementary requirements specified in a purchase order from a borrower or the borrower's contractor.

Crossarm is a term used interchangeably with arm.

Independent inspection relates to examination of material by an independent inspector employed by a commercial inspection agency.

Inspection means an examination of material in sufficient detail to insure conformity to all phases of the specification under which it was purchased.

Lot is a quantity of crossarms of like size, conditioning, and fabrication, usually making up one treating charge.

§ 1728.201

§1728.201

Producer is used to describe the party who manufactures and treats cross-arms.

Purchaser refers to either the RUS borrower or contractors acting as the borrower's agent, except where a part of the specification specifically refers to only the RUS borrower or the contractor.

Quality control designee refers to an individual designated by the producer to be responsible for quality control.

Reserve treated stock consists of timber products treated in accordance with this specification, prior to and in anticipation of the receipt of specific orders, and held in storage ready for immediate shipment.

Supplier is a term used interchangeably with producer, or in some cases, may be the distributor selling crossarms to the borrower.

Treating plant is the organization that applies the preservative treatment to the crossarms.

(c) Related specifications and standards incorporated by reference. The following specifications and standards are incorporated by reference. This incorporation by reference was approved by the Director of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies of each reference are available for inspection during normal business hours at RUS, room 1250-S, U.S. Department of Agriculture, Washington, DC 20250, or at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, http:// or go to: www.archives.gov/federal__register/

code_of_federal_regulations/

ibr_locations.html. Copies of these standards and specifications may be purchased from the addresses shown below.

(1) West Coast Lumber Inspection Bureau, Standard No. 17, Grading Rules for West Coast Lumber, September 1, 1991, available from West Coast Lumber Inspection Bureau, P.O. Box 23145, Portland, Oregon 97223, telephone (503) 639-0651, Fax (503) 684-8928.

(2) Southern Pine Inspection Bureau, Standard Grading Rules for Southern Pine Lumber, October 15, 1991, available from Southern Pine Inspection Bureau, 4709 Scenic Highway, Pensa7 CFR Ch. XVII (1-1-06 Edition)

cola, Florida 32504, telephone (904) 434-2611.

(i) Southern Pine Inspection Bureau, Special Product Rules for Structural, Industrial, and Railroad-Freight Car Lumber. October 15, 1991, available from Southern Pine Inspection Bureau, 4709 Scenic Highway, Pensacola, Florida 32504, telephone (904) 434-2611.

(ii) [Reserved]

(3) American Wood Preservers' Association (AWPA), Book of Standards, 1991 edition, available from AWPA, P.O. Box 286, Woodstock, Maryland 21163-0286.

(i) A1-91, Standard Methods for Analysis of Creosote and Oil-Type Preservatives.

(ii) A2-91, Standard Methods for Analysis of Waterborne Preservatives and Fire-Retardant Formulations.

(iii) A3-91, Standard Methods for Determining Penetration of Preservatives and Fire Retardants.

(iv) A5-91, Standard Methods for Analysis of Oil-Borne Preservatives.

(v) A6-89, Method for the Determination of Oil-Type Preservatives and Water in Wood.

(vi) A7-75, Standard Wet Ashing Procedure for Preparing Wood for Chemical Analysis.

(vii) A9-90, Standard Method for Analysis of Treated Wood and Treating Solutions by X-Ray Spectroscopy.

(viii) A11-83, Standard Method for Analysis of Treated Wood and Treating Solutions by Atomic Absorption Spectroscopy.

(ix) C1-91, All Timber Products—Preservative Treatment by Pressure Processes.

(x) C4-91, Poles—Preservative Treatment by Pressure Processes.

(xi) C8-91, Western Red Cedar and Alaska Yellow Cedar Poles—Preservative Treatment by the Full-Length Thermal Process.

(xii) C10-91, Lodgepole Pine Poles— Preservative Treatment by the Full-Length Thermal Process.

(xiii) C12-90, Western Larch Poles— Full-Length Preservative Treatment by Thermal Process.

(xiv) M1-90, Standard for the Purchase of Treated Wood Products.

(xv) M2-91, Standard for Inspection of Treated Timber Products.

(xvi) M3-81, Standard Quality Control Procedures for Wood Preserving Plants.

(xvii) M4-91, Standard for the Care of Preservative-Treated Wood Products.

(xviii) P1/P13-91, Standard for Coal Tar Creosote for Land and, Fresh Water and Marine (Coastal Water Use). (xix) P5-91, Standards for Waterborne

Preservatives. (xx) P8-91, Standards for Oil-Borne

Preservatives.

(xxi) P9-91, Standards for Solvents and Formulations for Organic Preservative Systems.

(4) American Institute of Timber Construction (AITC) 200-83, Inspection Manual, 1987 edition, available from AITC, 333 West Hampden Avenue, Englewood, Colorado 80110, telephone (303) 761-3212.

(5) American National Standards Institute (ANSI) 05.2-1983, American National Standard for Wood Products-Structural Glued Laminated Timber for Utility Structures, available from ANSI, 1430 Broadway, New York, New York 10018.

(6) American Society for Testing and Materials (ASTM) D9-87 (1992), Standard Terminology Relating to Wood, available from ASTM, 1916 Race Street, Philadelphia, PA 19103-1187, telephone number (215) 299-5585.

(d) Independent inspection plan. This plan or a Quality Assurance Plan, as described in paragraph (e) of this section, is acceptable for supplying crossarms. All crossarms produced under the independent inspection plan for use on an RUS financed system shall be inspected by a qualified independent inspector in accordance with §1728.202 of this part.

(1) The borrower has the prerogative to contract directly with the inspection agency for service. The borrower should, where practical, select the inspection agency so that continual employment is dependent only on performance acceptable to the borrower and in accordance with this specification. The selected inspection agency shall not subcontract the service to any other inspection agency without the prior written consent by the borrower.

(2) The producer shall not be a party to the selection of the inspection agen-

cy by the borrower and shall not interfere with the work of the inspector, except to provide notification of the readiness of material for inspection. To obtain the inspection services for reserve stock, the producer may deal directly with the inspection agency. Under the Independent Inspection Plan, the producer shall not treat material before it has been properly inspected in the white, as evidenced by the inspector's hammer mark.

(3) The methods of inspection described in this section and in §1728.202 of this part shall be used no matter which plan crossarms are produced under, i.e., Independent Inspection Plan, or Quality Assurance Plans, as described in this section. The number of crossarms actually inspected by monitors of quality control under a Quality Assurance Plan may vary from the number of crossarms inspected under the Independent Inspection Plan.

(e) Quality assurance plans. The producer shall furnish crossarms conforming to this specification as monitored by a Quality Assurance Plan acceptable to RUS. RUS borrower groups or agents for borrower groups endeavoring to operate Quality Assurance Plans shall submit their plan for assuring quality control to the Director, Electric Staff Division, Rural Utilities Service, Washington, DC 20250-1500, for specific approval prior to contracting with RUS borrowers under such plans.

(f) Material requirements—(1) Material and grade. All crossarms furnished under this specification shall be free of brashy wood, decay, and insect holes larger than 3/32 of an inch (0.24 cm), and shall meet additional requirements as shown on specific drawings. They shall be made of one of the following:

(i) Douglas-fir which conforms to the applicable crossarm provisions of paragraphs 170 and 170a, or the applicable transmission arm provisions of paragraphs 169 and 169a of the 1991 Standard Grading Rules for West Coast Lumber No. 17. All references to Douglas-fir shall be of coastal origin;

(ii) Southern Yellow Pine which conforms to the provisions of Dense Industrial Crossarm 65, as described in paragraph 31.2 in Southern Pine Inspection Bureau 1991 Special Product Rules for Southern Pine; or

§1728.201

§1728.201

(iii) Laminated wood crossarms shall conform to ANSI 05.2-1983, and have at least the same load carrying capacity as the solid sawn arm it replaces. The load carrying capacity of the laminated arms shall be determined by one of the procedures outlined in ANSI 05.2.

(2) Borrowers may use alternative wood crossarms that are listed in RUS Bulletin 1728C-100, List of Materials Acceptable for Use on Systems of RUS Electrification Borrowers.

(3) Knots. Sound, firm, and tight knots, if well spaced, are allowed.

(i) Slightly decayed knots are permitted, except on the top face, provided the decay extends no more than 3/4 of an inch (1.91 cm) into the knot and provided the cavities will drain water when the arm is installed. For knots to be considered well spaced, the sum of the sizes of all knots in any 6 inches (15.24 cm) of length of a piece shall not exceed twice the size of the largest knot permitted. More than one knot of maximum permissible size shall not be in the same 6 inches (15.24 cm) of length. Slightly decayed, firm, or sound "Pin knots" (3/8 of an inch (0.95 cm) or less) are not considered in size, spacing, or zone considerations.

(ii) Knots are subject to the following limits on size and location:

KNOT LIMITS FOR DISTRIBUTION ARMS DRAWING M-19 (SEE FIGURE 1, EXHIBIT A) ALL DIMENSIONS IN INCHES

Class of Knot and Location	Maximum Knot Diame- ter	
	Close Grain	Dense Grain
Round Knots		
Single Knot: Maximum Diameter.		
Center Section*.		
Upper Half	3/4	1
Lower Haif	1	1-1/4
Elsewhere	1-1/4	1-1/2
Sum of Diameters in a 6-Inch Length: Max- imum		
Center Section,		
Upper Half	1-1/2	2
Lower Half	2	2-1/2
Elsewhere	2-1/2	3

7 CFR Ch. XVII (1-1-06 Edition)

Inch	Cm	
3/4	1.91	
1	2.54	
1-1/4	3.18	
1-3/8	3.49	
1-1/2	3.81	
1-3/4	4.45	
1-7/8	4.76	
2	5.08	
2-1/4	5.72	
2-1/2	6.35	
3-1/4	8.26	
3-1/2	8.89	
3-5/8	9.21	
4-5/8	11.75	
5-5/8	14,29	
7-3/8	18.73	
9-3/8	23.81	

KNOT LIMITS FOR TRANSMISSION ARMS (SEE FIGURE 2, EXHIBIT A) ALL DIMENSIONS IN INCHES

Pole Mounting Hole Zone*	Maximum Diameter Fo Single Knot 3/4 1 for close grain 11/4 dense grain		
Upper Half (inner zone) Upper Half (outer zone)			in
	Nar-	Wide Face (Two Sides)	
Size**	row Face	Edge	Along Cen- terline
4-5/8×5-5/8 or less 5-5/8×7-3/8 3-5/8×9-3/8	1 11/4 3/4	1–1/4 1–3/8 1–3/4	11/4 1-7/8 21/4

"No knot will be closer than its diameter to the pole mounting hole. **For cross sections not shown, refer to grading rules.

(iii) Knot clusters shall be prohibited unless the entire cluster, measured on the worst face, is equal to or less than the round knot allowed at the specific location.

(iv) Spike knots shall be prohibited in deadend arms. Any spike knot across the top face shall be limited to the equivalent displacement of a knot 3/8 of an inch (0.95 cm) deep on one face and the maximum round knot for its particular location on the worst face, with a maximum width of 1 inch (2.54 cm) measured at the midpoint of the spiked section. Elsewhere across the bottom or side faces, spike knots shall not exceed 1/2 the equivalent displacement of a round knot permitted at that location, provided that the depth of the knot on the worst face shall not exceed the maximum round knot allowed at that location.

(v) Loose knots and knot holes shall drain water when the arm is normally

installed. In the center section, upper half, they shall not be greater than 1/2 the dimensions of round knots. Elsewhere, they shall not be greater than the round knot dimension. They shall be prohibited in deadend arms.

(vi) All knots except those "spike" knots intersecting a corner shall be measured on the least diameter of the knot.

(vii) A knot shall be considered to occupy a specific zone or section if the center of the knot (i.e. pith of knot) is within the zone or on the zone's boundary.

(viii) If a round or oval knot appears on two faces and is in two zones, each face shall be judged independently. When this does not occur, average the least dimension showing on both faces. Knots which occur on only one face of a free of heart center (FOHC) arm shall be permitted to be 25 percent larger than the stated size.

(ix) Knot spacing. Two or more knots opposite each other on any face shall be limited by a sum not to exceed the size of a maximum single knot permitted for the location. On all four faces, all knots shall be well spaced.

(x) Knots which have a maximum of 5/8 inch (1.59 cm) diameter may intersect pin holes in the center section. One inch (2.54 cm) diameter knots may intersect pin holes elsewhere.

(4) Miscellaneous characteristics, features and requirements. (i) The top face of distribution crossarms shall not have more than four medium pitch and bark pockets in 8 foot (2.4 m) arms, and not more than five pitch and bark pockets in 10 foot (3.0 m) arms. Elsewhere a maximum of six medium pockets in 8 foot (2.4 m) arms and eight in 10 foot (3.0 m) arms shall be permitted. Equivalent smaller pockets shall be permissible. An occasional large pocket is permissible.

(ii) Shakes shall be prohibited.

(iii) Checks. Prior to treatment on properly seasoned arms, single face checks shall not exceed an average penetration of 1/4 the depth from any face and shall be limited to 10 inches (25.40 cm) long on the top face, and 1/3 the arm length on the other faces. Checks shall not be repeated in the same line of grain in adjacent pin holes. The sum of the average depths of checks occurring in the same plane on opposite faces shall be limited to 1/4 the face depth.

(iv) Compression wood shall be prohibited on any face. It is permitted if wholly enclosed in the arm, more than six annual rings from the surface, and not over 3/8 of an inch (0.95 cm) in width.

(v) Insect holes larger than 3/32 of an inch (0.24 cm) shall be prohibited. Pin holes (i.e. holes not over 1/16 of an inch (0.16 cm) diameter) shall be allowed if scattered and not exceeding 10 percent of the arm girth.

(vi) Wane shall be allowed on one edge, limited to approximately 1 inch (2.54 cm), measured across the corner. Outside of the top center section, an aggregate length not to exceed 2 feet may have wane up to 1-1/2 inches (3.81 cm) on an occasional piece on one or both edges. Bark shall be removed.

(vii) \vec{P} rior to preservative treatment, crook, bow, or twist shall not exceed 1/ 2 of an inch (1.27 cm) in 8 foot arms (2.4 m) and 5/8 of an inch (1.59 cm) in 10 foot (3.0 m) arms.

(g) Manufacture. (1) All dimensions and tolerances shall conform to those shown on the drawings in this section or drawings supplied with the purchase order. Drawings supplied shall meet or exceed minimum dimensions and tolerances shown on the drawings in this section. Cross-sectional dimensions shall be measured and judged at about 1/4 the arm length, except when the defects of "skip dressing" or "machine bite or offset" are involved.

(2) Lamination techniques shall comply with ANSI 05.2-1983.

(3) Pin and bolt holes shall be smoothly bored without undue splintering where drill bits break through the surface. The center of any hole shall be within 1/8 of an inch (0.32 cm) of the center-line locations on the face in which it appears. The holes shall be perpendicular to the starting and finishing faces.

(4) Shape. The shape of the arms at any cross section, except for permissible wane, shall be as shown on the respective drawings in this section or supplied with the order. The two top edges may be either chamfered or rounded 3/8 of an inch (0.95 cm) radius. The two bottom edges may be slightly §1728.201

eased 1/8 of an inch (0.32 cm) radius for the entire length.

(5) Incising. The lengthwise surfaces of Douglas-fir crossarms shall be incised approximately 1/4 of an inch (0.64 cm) deep. The incision shall be reasonably clean cut with a spacing pattern that insures uniform penetration of preservative.

(6) Workmanship. All crossarms shall be first quality workmanship. Crossarms shall be dressed on four sides, although "hit and miss skips" may occur on two adjacent faces on occasional pieces. Five (5) percent of a lot or shipment may be 1/8 of an inch (0.32 cm) scant in thickness or width at the ends for a length not exceeding 6 inches (15.24 cm), or may have 1/8 of an inch (0.32 cm) machine bite on offset.

(h) Conditioning prior to treatment. (1) All solid sawn crossarms shall be made of lumber which has been kiln-dried. Douglas-fir arms shall have an average moisture content of 19 percent or less, with a maximum not to exceed 22 percent. Southern Yellow Pine arms shall have an average moisture content of 22 percent or less, with a maximum not to exceed 30 percent.

(2) Moisture content levels shall be measured at about 1/4 the length and at a depth of about 1/5 the crossarm's thickness. Additionally, the moisture content gradient between the shell (i.e. 1/4 of an inch (0.64 cm) deep) and the core (i.e. about 1 inch (2.54 cm) deep) shall not exceed 5 percentage points.

(3) A minimum of at least 20 solid sawn crossarms per treating charge shall be measured to verify moisture content and shall be duly recorded by the quality control designee or independent inspector.

(4) The moisture content of lumber used in laminating shall, at the time of gluing, be within the range of 8 to 12 percent, inclusive.

(i) *Preservatives.* (1) The preservatives shall be:

(i) Creosote which conforms to the requirements of AWPA Standard P1 when analyzed in accordance with the methods in AWPA Standard A1, sections 2, 3, 4, either 5 or 9, and 6;

(ii) Pentachlorophenol which contains not less than 95 percent chlorinated phenols and conforms to AWPA Standard P8 when analyzed in accordance with AWPA Standard A5 or A9. The hydrocarbon solvents for introducing the preservative into the wood shall meet the requirements of AWPA Standard P9 Type A; or

(iii) Waterborne preservatives, which may only be one of the following:

(A) Ammoniacal Copper Arsenates (ACA) and Ammoniacal Copper Zinc Arsenate (ACZA) which shall meet the requirements of AWPA Standard P5, when analyzed in accordance with methods in AWPA Standards A2, A9, or A11; and

(B) Chromated Copper Arsenates (CCA) which shall meet the requirements of one of the formulations given in AWPA Standard P5, sections 4, 5 or 6, and 10. Tests to establish conformity shall be made in accordance with AWPA Standards A2, A9, or A11.

(1) The pH of treating solutions of the waterborne preservatives shown in AWPA Standard P5, section 10, shall be determined in accordance with AWPA Standard A2, section 8.

(2) Waterborne preservatives are available either as oxides, which form non-ionizing chemical compounds in the wood, or as salts, which leave ionizing compounds as well as non-ionizing compounds in the wood. Salt formulations of a waterborne preservative are more corrosive to metal than the oxide formulation and may cause surface deposits. Unless otherwise specified in the purchase order, the oxide formulations of waterborne preservatives shall be supplied.

(3) Douglas-fir crossarms shall not be treated with CCA preservatives.

(4) Materials treated with waterborne preservatives shall be free of visible surface deposits.

(iv) Copper Naphthenate (CuN) concentrate used to prepare wood preserving solutions shall contain not less than 6 percent nor more than 8 percent copper in the form of Copper Naphthenate and shall conform to AWPA Standard P8 when analyzed in accordance with AWPA Standard A5. The hydrocarbon solvents for introducing the preservative into the wood shall meet the requirements of AWPA Standard P9 Type A.

(2) [Reserved]

(j) *Preservative treatment.* (1) All timber products treated under this specification shall be treated by either a pressure or a thermal (nonpressure) process.

(2) These materials may be further conditioned by steaming, or by heating in hot oil (Douglas-fir), within the following limits:

	Time Hours (max.)	Temperature Deg. F (max.)	
Steam	3	220 (104.4C)	
Heating in Preservative	3	210 (98.9C)	

(3) A final steam or hot oil bath may be used only to meet cleanliness requirements of paragraph (k) of this section. Total duration of the final steam bath shall not exceed 2 hours and the temperature shall not exceed 240 degrees Fahrenheit (115.6C).

(k) *Results of treatments.* (1) The quality control designee shall test or supervise the testing of each treated charge for penetration and retention.

(2) Method of sampling. When testing penetration and retention, a borer core shall be taken from not less than 20 crossarms in each treating charge. The borings shall be taken from any face except the top face at a point as close to the end as possible, being at least 3 inches (7.62 cm) from the end of the arm and no closer than 3 inches (7.62 cm) from the edge of the holes. The bored holes shall be plugged with preservative-treated plugs driven into the arm. Borings from laminated arms shall not be taken from the same laminate unless there is an end joint separation.

(3) Penetration by the preservative, as determined in accordance with AWPA Standard A3, shall be 100 percent of the sapwood in crossarms. In the heartwood of Douglas-fir crossarms, the penetration shall be not less than 3 inches (7.62 cm) longitudinally from the edge of holes and ends, and at least 3/16 inch (0.45 cm) from the surface of any face.

(4) Retention of preservative in the outer 6/10 of an inch (1.52 cm) for Douglas-fir and one inch (2.54 cm) for Southern Yellow Pine assay zones at the treating plant shall be not less than: §1728.201

Preservative	Retention (pcf)	AWPA Anal- ysis Method
Creosote	8	A6
Pentachlorophenol	0.4*	A5
ACA, ACZA, or CCA	0.4	A2, A7, A9, or A11
Copper Naphthenate	0.04	A5, A9, or A11

*This retention is for the lime ignition method. The copper pyridine method, retention 0.36 pcf, is required when timbers may have been in contact with sait water, and for all species native to the Pacific coast region. It is not required when it specifically states on the rough sawn material invoice that this material has not been in contact with sait water or is shown by analysis to have no additional chlorides present in the wood before treating.

(5) Cleanliness of lengthwise surfaces of all crossarms shall be free from tarry, greasy, or sticky material, and from oil exudation and pentachlorophenol crystallization (blooming).

(6) Re-treatment of materials which do not meet the penetration and retention requirements of this specification may be done only twice. Initial treatment steaming time plus re-treatment steaming time, combined, shall not exceed time allowed in paragraph (i) of this section.

(1) Marks and brands. (1) All crossarms shall be branded (hot brand) or die-stamped legibly and to a depth of approximately 1/16 of an inch (0.16 cm) before treatment.

(2) The letters and figures shall be not less than 1/2 of an inch (1.27 cm) in height. The top of the brand shall be oriented to the top of the arm.

(3) The brand or die-stamp shall include:

(i) The manufacturer's identification symbol;

(ii) Month and year of manufacture;

(iii) Species of timber such as DF for Douglas-fir and SP for Southern Yellow Pine; and

(iv) The preservative notated with a C for creosote, P for penta, S for salts, or N for Copper Naphthenate.

(4) An example is:

M-6-72 Manufacturer-Month-Year

DF-P Douglas-fir—penta treated

(5) The brand or stamp shall be placed on either of the wide surfaces of the arms, oriented with letters right side up towards the top of the arm and preferably about 1 foot (30.48 cm) from the midpoint of the arm.

§1728.201

(6) The mark should be approximately the same location on each type of crossarm of each producer.

(7) Brands, inspection marks, or quality assurance marks shall be removed from arms that do not meet these specifications

(m) Storage. (1) Producers may treat crossarms for reserve stock under any of the RUS approved plans. Prior to treating reserve stock, and annually thereafter, producers shall notify the Director of the Electric Staff Division of their intent to treat reserve stock. The letter of notification shall be addressed to the Director, Electric Staff Division, Rural Utilities Service, Washington, DC 20250-1500.

(2) RUS shall acknowledge, by letter, each notification of intent to treat material for reserve stock under the RUS specification.

(3) RUS's letter acknowledging the plant's advance notice of intent to treat material for reserve treated stock for the calendar year in question shall be evidence of compliance with the notification requirements.

(4) Producers shall notify RUS of:

(i) The locations of all storage or distribution yards where reserve treated stock will be maintained;

(ii) The designation of the RUS-approved plan;

(iii) The name of the selected inspection agency, where applicable; and

(iv) Any changes that occur during the year.

(5) Crossarms treated with oil-borne preservatives which have been held in storage for more than 1 year before shipment to the borrower, shall be reassayed before shipment and shall be re-treated if found nonconforming for retention on orders placed in accordance with this section.

(6) The crossarms shall meet the assay after re-treatment in accordance with paragraph (k) of this section.

(7) Crossarms which are held in storage after final acceptance shall be

7 CFR Ch. XVII (1-1-06 Edition)

stacked in piles or on skids in such a manner as to assure good ventilation. The stacks shall be covered or stored indoors for protection from the sun and weather to reduce checking, bending, and loss of preservative.

(8) Borrowers or their contractors shall not purchase reserve treated stock from plants that fail to comply with the notification requirements.

(n) *Drawings*. (1) The drawings of Exhibit B of this section, Crossarm Drilling Guide, have a type number and show in detail the hole size, shape, and pattern desired for crossarms ordered under this specification.

(2) Purchase orders shall indicate the type required.

(3) Crossarms shall be furnished in accordance with the details of these drawings or in accordance with drawings attached to the purchase order

(4) Technical drawings for transmission crossarms are published in RUS Bulletin 1728F-T805B (formerly 50-1), Electric Transmission Specifications and Drawings, 115kV through 230kV, and RUS Bulletin 1728F-T805A (formerly 50-2), Electric Transmission Specification and Drawings, 34.5kV through 69kV.

(5) Appropriate drawings for transmission arms are to be specified and included with purchase orders.

(o) Destination inspection. (1) When cross-sectional tolerances are measured at destination, average shrinkage allowance shall be considered using the arm's current moisture content and actual size.

(2) Using the average shrinkage allowances for Douglas-fir and Southern Yellow Pine as 1 percent size change for each four point moisture content change below the fiber saturation point, calculations can be made to determine if the arm met the minimum size at time of manufacture, when the arm was to meet the average moisture content.
§1728.201

EXHIBIT A TO §1728.201-DISTRIBUTION AND TRANSMISSION ARMS



§ 1728.201

7 CFR Ch. XVII (1-1-06 Edition)



EXHIBIT B TO § 1728.201-CROSSARM DRILLING GUIDE

[58 FR 41396, Aug. 3. 1993, as amended at 69 FR 18803, Apr. 9, 2004]

§1728.202 RUS Bulletin 1728H-702, RUS Specification for Quality Control and Inspection of Timber Products.

(a) Scope. This specification describes in more detail the responsibilities and procedures pertaining to quality control for crossarms, as specified in \$1728.201 of this part, and poles, covered in RUS Bulletin 1728F-700, incorporated by reference in \$1728.97 of this part and in \$1755.97 of 7 CFR part 1755.

(b) Related specifications and standards incorporated by reference. The following specifications and standards referenced throughout this section are incorporated by reference. This incorporation by reference is approved by the Director of the Federal Register in accordance with 5 U.S.C. 552(a) and 1 CFR part 51. Copies of each are available for inspection during normal business hours at RUS, room 1250-S, U.S. Department of Agriculture, Washington, DC 20250 or at the National Archives and Records Administration (NARA). For information on the availability of this material at NARA, call 202-741-6030, or go to: http://www.archives.gov/ federal__register/

code_of_federal_regulations/

ibr_locations.html. Copies of these standards and specifications may be purchased from the addresses shown below.

(1) American Wood-Preservers' Association (AWPA), Book of Standards, 1991 edition, available from AWPA, P.O. Box 286, Woodstock, Maryland 21163-0286.

(i) A1-91, Standard for Coal Tar Creosote for Land and Fresh Water Use.

(ii) A2-91, Standard Methods for Analysis of Waterborne Preservatives and Fire-Retardant Formulations.

(iii) A3-91, Standard Methods for Determining Penetration of Preservatives and Fire Retardants.

(iv) A5-91, Standard Methods for Analysis of Oil-Borne Preservatives.

(v) A6-89, Method for the Determination of Water and Oil-Type Preservatives in Wood.

(vi) A7-75. Wet ashing Procedure for Preparing Wood for Chemical Analysis.

(vii) A9-90, Standard Method for Analysis of Treated Wood and Treating Solutions by X-Ray Emission Spectroscopy. (viii) A11-83, Analysis of Treated Wood and Treating Solutions by Atomic Absorption Spectroscopy.

(ix) C1-91, Standard for Preservative Treatment by Pressure Processes All Timber Products.

(x) C4-91. Standard for the Preservative Treatment of Poles by Pressure Processes.

(xi) C8-91, Standard for the Full-Length Thermal Process Treatment of Western Red Cedar Poles.

(xii) C10-91, Lodgepole Pine Poles— Preservative Treatment by the Full-Length Thermal Process.

(xiii) C12-90, Western Larch Poles— Full-Length preservative Treatment by Thermal Process.

 $(xi\nu)\,$ M1-90, Standard for the Purchase and Preservation of Forest Products.

(xv) M2-91, Standard Instructions for the Inspection of Preservative Treatment of Wood.

(xvi) M3-81, Standard Quality Control Procedures for Wood Preserving Plants.

(xvii) M4-91, Standard for the Care of Preservative-Treated Wood Products.

(xviii) P1/P13-91, Standard for Coal Tar Creosote for Land and, Fresh Water and Marine (Coastal Water Use). (xix) P5-91, Standards for Water-

Borne Preservatives. (xx) P8-91, Standards for Oil-Borne

Preservatives.

(xxi) P9-91, Standards for Solvents for Organic Preservative Systems.

(2) American Institute of Timber Construction (AITC) 200-83, Inspection Manual, 1987 edition, available from AITC, 333 West Hampden Avenue, Englewood, Colorado 80110.

(3) American National Standards Institute (ANSI) 05.2-1983, American National Standard for Wood Products— Structural Glued Laminated Timber for Utility Structures, available from ANSI, 1430 Broadway, New York, New York 10018.

(4) American National Standards Institute/American Institute of Timber Construction (ANSI/AITC) A190.1-1983, American National Standard for Wood Products—Structural Glued Laminated Timber, available from ANSI, 1430 Broadway, New York, New York 10018.

§1728.202

(5) American Society for Testing and Materials (ASTM) D9-87 (1992), Standard Terminology Relating to Wood, available from ASTM, 1916 Race Street, Philadelphia, Pennsylvania 19103-1187, telephone number (215) 299-5585.

(c) General stipulations. (1) Each RUS electric borrower shall submit to the Director, Electric Staff Division, Rural Utilities Service, room 1250-S, 14th and Independence Avenue, SW., Washington, DC 20250-1500, in January of each year a list of plants from which it obtained poles or crossarms during the preceding calendar year.

(2) Ultimate quality control is the responsibility of the producer's management; however, a member of the producer's staff shall be designated quality control designee and charged with the responsibility for the exercise of proper quality control procedures. The requirements in American Wood Preservers' Association (AWPA) Standard M3, covering records, adequate laboratory, plant gauges, and other plant facilities including proper storage, shall be followed.

(3) The methods of inspection described in this section shall be used no matter which plan timber products are purchased under, i.e., Insured Warranty Plan, Independent Inspection Plan, or Quality Assurance Plans as described in §1728.201 of this part or RUS Bulletin 1728F-700. The number of poles and crossarms actually inspected by monitors for quality control under a Quality Assurance Plan or the Insured Warranty Plan may vary from the number of poles and crossarms inspected under the Independent Inspection Plan. Under the Independent Inspection Plan, each pole and a sample number of crossarms shall be inspected.

(4) Under the Independent Inspection Plan, the RUS borrower should designate in the purchase order which inspection agency it has selected. Unless the RUS borrower contracts for inspection as a separate transaction, the treating company shall obtain the services of the RUS borrower's designated inspection agency. For reserve treated stock for purchase under the Independent Inspection Plan, the treating company shall obtain the services of an inspection agency. Selection of and changes in inspection agencies for

7 CFR Ch. XVII (1-1-06 Edition)

reserve treated stock shall be promptly reported to the Director, Electric Staff Division, Rural Utilities Service, Washington, DC 20250-1500, in accordance with RUS Bulletin 1728F-700, and §1728.201.

(5) Individual inspectors in the employ of Independent Inspection Agencies shall be experienced and competent. The inspector shall perform all phases of the inspection personally and in the proper sequence. The primary responsibility of the inspector is to determine, for the borrower, by careful inspection and verification, that the timber products, preservative, and treatment meet the requirements of RUS Bulletin 1728F-700 and Bulletin 1728H-701 and that the methods, storage facilities, and production equipment conform to applicable RUS specifications. For details of the recommended inspector's qualifications see appendix A of this section.

(6) Laminated materials for use on RUS borrower systems shall follow manufacturing and quality control requirements as specified in ANSI 05.2-1983, American National Standard for Wood Products—Structural Glued Laminated Timber for Utility Structures, and ANSI/AITC A190.1-1983, American National Standard for Wood Products—Structural Glued Laminated Timber. The product shall be marked and certified.

(i) Laminated material shall be inspected by a qualified inspection and testing agency.

(ii) Quality control of material shall be performed to determine conformance with §1728.201 of this part and AITC 200-83, Inspection Manual.

(d) Quality control and inspection procedures for product acceptance. It is the responsibility of the plant quality control designee to perform the following procedures to insure that a particular lot of material conforms to the requirements of the applicable RUS specification prior to treatment. After the plant quality control designee has performed these procedures, a particular lot of material shall be released to the inspector for verification of conformance.

(1) Poles can be purchased under any of the three purchase plans. These

plans are Insured Warranty Plan, Independent Inspection Plan, or a Quality Assurance Plan. Under the Independent Inspection Plan, all poles in a lot shall be inspected. Under the Insured Warranty Plan and a Quality Assurance Plan, the number of poles in a lot actually inspected may be less than every pole, depending on the terms of the plans.

(i) Ample space and assistance shall be provided by the treating plant for handling and turning to insure that the surfaces of all items can be adequately inspected.

(ii) Under the Independent Inspection Plan, all poles shall be inspected for conformance to the requirements of RUS Bulletin 1728F-700. If a pole is rejected and the cause of rejection is corrected, the rejected pole may be offered again for inspection as new material.

(iii) Dimensions, length, and circumference shall be measured by a standard steel pole tape to determine that they are in agreement with the details for class and length in the brand and butt stamp. If it is obvious by visual comparison with a measured pole that the brand information is correct, individual poles need not be measured. Pole circumference dimensions made prior to treatment shall govern acceptance. Reduction in dimension due to treatment and shipping shall be not more than 2 percent below the minimum for the pole class.

(iv) If 15 percent of the poles in a lot offered for inspection are defective, the inspector shall terminate the inspection. Re-examination of an entire lot by plant quality control shall be required when the number of rejected poles equals or exceeds 15 percent of the lot inspected. All defective or nonconforming poles either shall be removed from the lot or marked out.

(v) Poles in a lot inspected for decay shall be of the same seasoning condition. If the independent inspector suspects that decay has occurred, he shall cut a slice from both ends for closer examination. If 5 percent of the inspected poles in a lot shows evidence of decay, the entire lot shall be unconditionally rejected without further sorting.

(vi) Moisture content, when limited by the purchaser, as stated on the borrower's purchase order, shall be measured by calibrated electric moisture meter. Calibration of the meter shall include not only the zero settings for the X and Y readings, but also two resistance standards for 12 and 22 percent moisture content.

(vii) Material failing to conform for moisture content may be retested upon request after a recalibration of the instrument. The results of the second test shall govern disposition of the lot.

(viii) Re-examination for any mechanical damage or deterioration and for original acceptance shall be conducted on timber products not treated within 10 days after original inspection.

(2) Crossarms can be purchased only under either of two purchase plans. These plans are the Independent Inspection Plan or Quality Assurance Plans. Under the Independent Inspection Plan, crossarms are to be inspected prior to manufacture, during manufacture, and after treatment. Under a Quality Assurance Plan, crossarms are monitored according to the terms of the quality assurance program acceptable to RUS.

(i) Inspection prior to treatment shall include:

(A) Surface inspection of all ends of all arms. This is usually done on the stacks of arms prior to manufacture. Particular attention shall be paid to defects commonly found in the ends, such as compression wood, red heart and other forms of decay, shakes, splits, through checks, scantiness, honeycomb, and low density, determined by rings per inch (centimeter) and percent of summerwood. Whenever the number of nonconforming arms is found to exceed 0.5 percent of the lot or one arm, whichever is greater, the entire lot shall be rejected for excess number of defective ends. After the producer has removed or marked out the defective material, the arms may be resubmitted for inspection.

(B) Surface inspection of the lengthwise sides performed on a random representative sample. The sample size shall equal 20 percent of a lot size or 200 arms, whichever is smaller. The inspector shall examine side surfaces as they are slowly rotated. When necessary, the rotation may be stopped for

§1728.202

closer inspection. Whenever the number of nonconforming arms is found to exceed 2 percent of the sample size, the entire lot shall be rejected. After the producer has removed or marked out the defective material, the arms may be resubmitted for inspection.

(C) Check of moisture content of the random sample by a calibrated moisture meter.

(D) Check of crossarm dimensions of the random sample measured after surfacing.

(ii) Inspection during manufacture shall consist of:

(A) Checking bolt and insulator pin holes for squareness and excessive splintering;

(B) Checking brands for completeness, location, and legibility; and

(C) Checking arms for conformance.

(iii) Under the Independent Inspection Plan, there shall be a final inspection during and after treatment for preservative retention and penetration and for damage.

(3) Structural glued laminated timber shall be tested and inspected in accordance with AITC 200-83, Inspection Manual. Grade of lumber shall be inspected by a qualified grader for specified quality, and so marked, in accordance with grading rules of the American Lumber Standards. Adhesives used for all structural arms shall meet requirements of ANSI 05.2-83, paragraph 5.2. Melamine urea adhesives shall not be used. End joint spacings and limitations shall be in accordance with ANSI 05.2-83.

(e) *Preservatives.* (1) Creosote shall conform to the requirements of AWPA Standard P1 when analyzed by AWPA Standard A1, sections 2, 3, 4, either 5 or 9, and 6.

(i) Each occasional charge, all material treated in a cylinder at one time, shall be analyzed.

(ii) The first charge and one of every five charges randomly selected in consecutive charges shall be analyzed.

(2) Solutions of waterborne preservatives shall be analyzed for components in accordance with AWPA Standards A2, A9, or A11, and shall meet the requirements of P5 for composition. AWPA A2 shall be used as a referee method. 7 CFR Ch. XVII (1-1-06 Edition)

(3) Pentachlorophenol shall contain not less than 95 percent chlorinated phenols and conform to AWPA Standard P8 in hydrocarbon solvent AWPA P9 Type A.

(4) Copper Naphthenate in hydrocarbon solvent (AWPA P9 Type A) shall contain not less than 6 percent nor more than 8 percent copper in the form of Copper Naphthenate and conform to AWPA Standard P8 when analyzed in accordance with AWPA Standard A5.

(f) Plant facilities and inspection during treatment. (1) Manufacturing and treating plant facilities shall conform to AWPA Standard M3, paragraph 3. Pressure plants shall be equipped with recording instruments to register time, pressure, temperature and vacuum during each cycle of treatment. They shall also be equipped with indicating thermometers and pressure and vacuum gauges to check the accuracy of the recorders. Work tanks shall be equipped with a thermometer. Thermal treating vats shall be equipped with a time and temperature recorder and with an indicating thermometer. Temperature recording devices are not mandatory for plants treating exclusively with waterborne preservatives.

(2) Under the Independent Inspection Plan, the inspector shall be present during the treatment procedure, except at times when it may be impractical, such as during late night or early morning treatments. At such times, temperature, pressure, and vacuum data shall be taken from the recording charts.

(3) Recording instruments shall be checked with indicating gauges and thermometers. Inaccuracies shall be referred to the treating company for prompt correction. In the event of an inaccuracy, indicating possible damage to the material, the inspector shall reject the charge.

(g) Results of treatment. (1) Poles shall be tested for retention and penetration by means of a calibrated increment borer 0.2 inches $(0.51 \text{ cm}) \pm 0.02$ inches (0.05 cm) in diameter in accordance with procedures in AWPA Standard M2, paragraph 5.22. Under the Independent Inspection Plan, all treating charges

shall be tested for retention and penetration. Plant quality control and independent inspection shall do their analyses separately. Under the Insured Warranty Plan and Quality Assurance Plans, the frequency of testing retention and penetration may vary according to the plan.

(i) Western red and northern white cedars and western larch poles shall be bored at any point of the periphery approximately 6-12 inches (15.24-30.48 cm) above ground line and all other species approximately 1 foot (30.48 cm) above or below the brand.

(ii) Penetration shall be determined in accordance with AWPA Standard A3. Chrome Azurol S and Penta-Check shall be used to determine penetration of copper containing preservatives and penta, respectively.

(iii) Retention sampling. (A) When there are 20 or more poles in the treating charge, the retention sample for creosote shall consist of 20 assay zones from southern pine and Douglas-fir poles. All poles in charges with fewer than 20 poles shall be bored once. Charges with less than 15 poles shall be bored once and bored again on a random basis to obtain a minimum of 15 assay zones.

(B) Retention samples shall be taken from 20 poles in charges of 20 or more poles.

(C) Retention samples for Alaska yellow, western red, and northern white cedars shall be comprised of a minimum of 30 assay zones for creosote and waterborne preservatives. For penta charges of fewer than 30 poles, the sample shall contain the assay zone from each pole in the lot.

(D) Retention samples shall be comprised of borings, representative of pole volumes for each class and length in the charge. Further selection and marking of poles of mixed seasoning, volume, and location on the tram shall be made as illustrated in the following table:

Num- ber of Poles	Class/Length	Vol. in cu. ft.	Total Vol- ume	Num- ber of Bor- ings
27	7/30(09.1 m)	232	15	3
26	4/35(10.7 m)	447	29	6
11	5/35(10.7 m)	163	10	2
55*	6/35(10.7 m)	704	46	9
Total		1.546	_	

*If a portion of these poles were green and some partially seasoned, then the number of borings should reflect the approximate percentage of each.

(iv) When material in a lot consists of fewer pieces than the designated minimum number of samples for assay, additional borings shall be taken so as to make up at least the minimum sample, and in such manner that the sample is representative of the lot of material with respect to any variations in size, seasoning condition, or other features that might affect the results of treatment.

(v) Analyses for preservative retention shall be performed.

(A) Creosote shall be analyzed by AWPA Standard A6.

(B) Penta shall be analyzed by AWPA Standard A5 or A9. Copper pyridine method is required when timber may have been in contact with salt water and for all species native to the Pacific coast region, unless the raw material invoice specifically states that the material either has not been in contact with salt water or has been shown by analysis to have contained no additional chlorides before treating.

(C) Copper Naphthenate shall be analyzed by tests in accordance with AWPA Standards A5 or A9.

(D) Waterborne preservatives shall be analyzed by tests in accordance with AWPA Standards A2, A7, A9, or A11.

(E) Prior to unloading a tram, the inspectors may take their own samples and analyze them concurrently with the quality control designee, but each shall work independently, and quality control data shall be presented before acceptance of the charge.

(vi) Penetration sampling of poles. (A) Group A poles consist of poles with a circumference of 37.5 inches (95.25 cm) or less at 6 feet (1.8 m) from butt.

(1) Bore 20 Group A poles or 20 percent of the poles, whichever is greater. Accept if 100 percent of the sample conform; otherwise, bore all poles.

§1728.202

§ 1728.202

(2) Re-treat the charge if more than 15 percent of the borings are found to be nonconforming.

(3) Re-treat all nonconforming poles if 15 percent or fewer fail the requirement.

(B) Group B poles consist of poles with circumference greater than 37.5 inches (95.25 cm) at 6 feet (1.8 m) from the butt.

(1) For Group B poles 50 feet (15.2 m) and shorter, bore each pole and re-treat only those found to be nonconforming, unless more than 15 percent fail; in that case, re-treat the entire lot.

(2) For Group B poles longer than 50 feet (15.2 m), bore each pole twice at 90 degrees apart around the pole and accept only those poles conforming to the penetration requirement in both borings. All nonconforming poles may be re-treated only twice.

(vii) All holes (nominal 0.2 of an inch (0.05 cm) diam. bit) shall be promptly filled with treated, tight-fitting wood plugs.

(2) Under the Independent Inspection Plan, all treating charges of crossarms shall be tested for retention and penetration. Plant quality control inspectors and independent inspectors shall do their analyses *independently*. Under the Quality Assurance Plans, the frequency of testing retention and penetration may vary according to the plan.

(i) The penetration and retention sample shall consist of 20 (48 for creosote) outer 6/10 of an inch (1.52 cm) for Douglas-fir and 1 inch (2.54 cm) for Southern Yellow Pine zones from borings taken from any face except the top face at a location as close to the end as possible being at least 3 inches (7.62 cm) from the end of the arm and no closer than 3 inches from the edge of any holes. For laminated material, borings shall be taken from laminates on a random basis.

(ii) Penetration shall be tested by taking not less than 20 borings from 20 crossarms in each charge, determined in accordance with AWPA Standard A3. Chrome Azurol S and Penta-Check shall be used to determine penetration of copper containing preservatives and penta, respectively.

(3) Laminated material shall be checked for any evidence of 7 CFR Ch. XVII (1-1-06 Edition)

delamination due to treatment and for the identifying quality stamp of AITC or American Plywood Association (APA).

(4) When x-ray fluorescence (XRF) instruments are used to analyze preservative or retention, Periodic Instrument Checks (PIC) shall be made by the treating plant and any outside inspection agency using the treating plant's instrument or its own. Appendix B of this section outlines a recommended procedure.

(5) At a minimum, treating plants shall perform the PIC weekly and record the results in the instrument's log, which shall be stored with the instrument. Independent inspection agencies shall use their own samples to perform the PIC on treater's instrument once per visit, not to exceed one PIC per week. Inspection agencies shall record their results in the instrument's log and state the date of its latest PIC on all treating reports.

(6) XRF instruments shall be accurate and reliable, and they shall generate reproducible results. Instruments shall have thorough instructions which should include recommendations on drying techniques, equipment, and density calculations. These drying recommendations shall be followed when using these instruments.

(h) Product acceptance. Under the Independent Inspection Plan, the inspector shall signify acceptance by marking each piece of accepted material with a clear, legible hammer stamp in one end prior to treatment and in the other end after treatment. The inspector shall personally mark each piece, and shall not delegate this responsibility to another person.

(i) Charge inspection reports. (1) Inspection Reports shall cover the following:

(i) The total pieces in the lot, number of and causes for rejection;

(ii) The conditioning of the material prior to treatment;

(iii) The analyses of preservative identified by the analyst's signature or certification;

(iv) The details of treatment; and

(v) The results of treatment. These results shall include the following:

(A) The depth of penetration \overline{f} or retention sample and a summary of all

poles rejected for insufficient penetration;

(B) Worksheets for retention analyses, each identified by quality control designee and independent inspector;

(C) The number of pieces offered and rejected, together with the cause(s) for rejection;

(D) The date of latest Periodic Instrument Check.

(2) On each inspection report the independent inspector and the plant quality control designee shall certify, in writing, that the material listed on the report has been inspected before, during, and after treatment, and that the preservative used was analyzed in accordance with the requirements of this section.

(3) Each inspector or inspection agency shall retain for a period of 1 year a copy or transcript of each report of inspection, together with laboratory worksheets covering retention by assay and preservative analyses for the purchaser, and on request shall furnish a copy or transcript of any of these reports to the Director, Electric Staff Division, Rural Utilities Service, Washington, DC 20250-1500.

(j) Charge numbers on re-treat poles. The letter "R" shall be added to the original charge number in the butts of all poles that are re-treated for insufficient penetration or retention of preservative. All poles that fail to meet treatment requirements after two retreatments shall be permanently rejected.

(k) Safety provisions. Poles intended for RUS borrowers shall not be inspected when, in the opinion of the inspector, unsafe conditions are present.

APPENDIX A TO § 1728.202—RECOMMENDED INSPECTORS' QUALIFICATIONS

(a) Inspection agencies should see that inspectors assigned to the inspection of timber products and treatment for RUS borrowers are competent and experienced.

(b) Recommended experience. In general, any of the following examples are recommended as minimum qualifying experience before a new inspector may be permitted to inspect timber products for RUS borrowers:

(1) Three years' experience as an inspector of timber and the preservative treatment of timber.

(2) Three years' experience in timber treating plant quality control work. (3) Under the direct supervision of an experienced, well-qualified inspector, who has performed the following:

(i) Inspected at least 2,500 poles and/or crossarms "in the white."

(ii) Checked preservative penetration results on at least 500 poles and crossarms.

(iii) Made at least 35 wood assays for preservative retention.

(iv) Made at least 25 analyses of each type preservative used on material the person is assigned to inspect.

(v) In both (b)(1) and (b)(2) of this appendix A, the experience should be not less than that required in (b)(3)(i), (b)(3)(ii), (b)(3)(ii), and (b)(3)(iv).

(4) Inspectors experienced in the inspections of one product, such as poles, should not be qualified to inspect another product, such as crossarms, until the above experience is gained.

(5) The inspector should be especially well informed in wood preservation and the operation of a timber treating plant, and be competent in preservative analysis and other laboratory work.

(6) In all cases, an inspector should be thoroughly instructed in the application of RUS specifications and the standards pertaining thereto before being permitted to independently inspect timber products and the treatments applied to them. Knowledge of these specifications and standards, as well as the inspector's proficiency. may be checked routinely by members of the RUS staff.

APPENDIX B TO §1728.202—PERIODIC INSTRUMENT CHECK X-RAY FLUORESCENCE

(a) *General.* The following sample calibration standards and procedures may be used in lieu of comparison with analysis by wet ash or lime ignition methods.

(b) Penta. Until such time as AWPA approves calibration standards for penta, the following method should be used to run a salt water solution to measure Cl (chloride).

(1) Standard Solution. Dry approximately 15 grams of reagent grade NaCl at 105 °C for 1 hour. Weigh 10.00 grams into a tared beaker. Add distilled water until the total weight is 100.00 grams. Stir until completely dissolved. This will give a 10 percent weight to weight solution of NaCl.

(2) Baseline Check. (i) Insure that the instrument is in good agreement with lime ignition.

(ii) Record any user correction factors.

(iii) Stabilize and standardize the instrument.

(iv) Run the salt solution five times using the PENTA-OIL calibration mode.

 $\langle v \rangle$ Record the average and standard deviation of the values for percent penta. The average value will now be considered the nominal value.

Pt. 1730

(3) Periodic Instrument Check. Run the salt solution two times and average the results. If the value is more than ± 5 percent of the nominal value, the instrument needs further calibration, following manufacturer's recommendation.

(c) Waterborne preservatives. Treaters and inspection agencies should purchase AWPA Committee P-5 Standard Reference Materials to analyze on their instruments. Reference materials should be in the retention range of the material being produced at the plants. If the value is more than ±5 percent of the nominal value, the instrument needs further calibration. AWPA Committee P-5 Standard Reference Materials may be purchased from:

American Wood Preservers' Association, P.O. Box 286, Woodstock, Maryland 21163, Phone: (410) 456-3169.

[58 FR 41406, Aug. 3, 1993, as amended at 69 FR 18803, Apr. 9, 2004]

PART 1730—ELECTRIC SYSTEM OPERATIONS AND MAINTENANCE

Subpart A-General

Sec.

- 1730.1 Introduction.
- 1730.2 RUS policy.
- 1730.3 RUS addresses
- 1730.4 Definitions.
- 1730.5-1730.19 [Reserved]

Subpart B—Operations and Maintenance Requirements

- 1730.20 General.
- 1730.21 Inspections and tests.
- 1730.22 Borrower analysis.
- 1730.23 Review rating summary, RUS Form
- 1730.24 RUS review and evaluation.
- 1730.25 Corrective action.
- 1730.26 Certification.

1730.27 Vulnerability and Risk Assessment (VRA).

1730.28 Emergency Restoration Plan (ERP). 1730.29 Grants and Grantees.

- 1730.30-1730.99 [Reserved]
- APPENDIX A TO SUBPART B-REVIEW RATING SUMMARY, RUS FORM 300

AUTHORITY: 7 U.S.C. 901 et seq., 1921 et seq., 6941 et seq.

SOURCE: 63 FR 3450, Jan. 23, 1998, unless otherwise noted.

Subpart A—General

§1730.1 Introduction.

(a) This part contains the policies and procedures of the Rural Utilities

7 CFR Ch. XVII (1-1-06 Edition)

Service (RUS) related to electric borrowers' operation and maintenance practices and RUS' review and evaluation of such practices.

(b) The policies and procedures included in this part apply to all electric borrowers (both distribution borrowers and power supply borrowers) and are intended to clarify and implement certain provisions of the security instrument and loan contract between RUS and electric borrowers regarding operations and maintenance. This part is not intended to waive or supersede any provisions of the security instrument and loan contract between RUS and electric borrowers.

(c) The Administrator may waive, for good cause, on a case by case basis, certain requirements and procedures of this part.

§1730.2 RUS policy.

It is RUS policy to require that all property of a borrower be operated and maintained properly in accordance with the requirements of each borrower's loan documents. It is also RUS policy to provide financial assistance only to borrowers whose operations and maintenance practices and records are satisfactory or to those who are taking corrective actions expected to make their operations and maintenance practices and records satisfactory to RUS.

§1730.3 RUS addresses.

(a) Persons wishing to obtain forms referred to in this part should contact: Program Support and Regulatory Analysis, Rural Utilities Service, U.S. Department of Agriculture, Stop 1522, 1400 Independence Ave., SW., Washington, DC 20250-1522, telephone (202) 720-8674. Borrowers or others may reproduce any of these forms in any number required.

(b) Documents required to be submitted to RUS under this part are to be sent to the office of the borrower's assigned RUS General Field Representative (GFR) or such other office as designated by RUS.

§1730.4 Definitions.

Terms used in this part have the meanings set forth in 7 CFR Part 1710.2. References to specific RUS forms and other RUS documents, and

to specific sections or lines of such forms and documents, shall include the corresponding forms, documents, sections and lines in any subsequent revisions of these forms and documents. In addition to the terms defined in 7 CFR Part 1710.2, the term *Prudent Utility Practice* has the meaning set forth in Article 1, Section 1.01 of Appendix A to Subpart B of 7 CFR Part 1718-Model Form of Mortgage for Electric Distribution Borrowers, for the purposes of this Part.

§§ 1730.5-1730.19 [Reserved]

Subpart B—Operations and Maintenance Requirements

§1730.20 General.

Each electric program distribution, transmission and generation borrower (as defined in §1710.2) shall operate and maintain its system in compliance with prudent utility practice, in compliance with its loan documents, and in compliance with all applicable laws, regulations and orders, shall maintain its systems in good repair, working order and condition, and shall make all needed repairs, renewals, replacements, alterations, additions, betterments and improvements, in accordance with applicable provisions of the borrower's security instrument. Each borrower is responsible for on-going operations and maintenance programs, individually or regionally performing a system security Vulnerability and Risk Assessment (VRA), establishing and maintaining an Emergency Restoration Plan (ERP), maintaining records of the physical, cyber and electrical condition and security of its electric system and for the quality of services provided to its customers. The borrower is also responsible for all necessary inspections and tests of the component parts of its system, and for maintaining records of such inspections and tests. Each borrower shall budget sufficient resources to operate and maintain its system and annually exercise its ERP in accordance with the requirements of this part. An actual manmade or natural event on the borrowers system in which a borrower utilizes a significant portion of its ERP shall count as an annual exercise for that calendar year,

provided that after conclusion of the event, the borrower verifies accuracy of the emergency points-of-contact (POC) and the associated contact numbers as listed in their ERP. For portions of the borrower's system that are not operated by the borrower, if any, the borrower is responsible for ensuring that the operator is operating and maintaining the system properly in accordance with the operating agreement.

[69 FR 60540, Oct. 12, 2004]

§1730.21 Inspections and tests.

(a) Each borrower shall conduct all necessary inspections and tests of the component parts of its electric system, annually exercise its ERP, and maintain records of such inspections and tests. For the purpose of this part, "Exercise" means a borrower's Tabletop execution of, or actual implementation of, the ERP to verify the operability of the ERP. Such Exercise may be performed singly by an individual borrower, or as an active participant in a multi-party (to include utilities, government agencies and other participants or combination thereof) Tabletop execution or actual full implementation of the ERP. For the purpose of this part, "Tabletop" means a hypothetical emergency response scenario in which participants will identify the policy, communication, resources, data, coordination, and organizational elements associated with an emergency response.

(b) The frequency of inspection and testing will be determined by the borrower in conformance with applicable laws, regulations, national standards, and Prudent Utility Practice. The frequency of inspection and testing will be determined giving due consideration to the type of facilities or equipment, manufacturer's recommendations, age, operating environment and hazards to which the facilities are exposed, consequences of failure, and results of previous inspections and tests. The records of such inspections and tests will be retained in accordance with applicable regulatory requirements and Prudent Utility Practice. The retention period should be of a sufficient time period to identify long-term trends. Records must be retained at

§ 1730.22

7 CFR Ch. XVII (1–1–06 Edition)

least until the applicable inspections or tests are repeated.

(c) Inspections of facilities must include a determination of whether the facility complies with the National Electrical Safety Code, National Electrical Code (as applicable), and applicable State or local regulations and whether additional security measures are considered necessary to reduce the vulnerability of those facilities which, if damaged or destroyed, would severely impact the reliability and security of the electric power grid, cause significant risk to the safety and health of the public and/or impact the ability to provide service to consumers over an extended period of time. The electric power grid, also known as the transmission grid, consists of a network of electrical lines and related facilities, including certain substations, used to connect distribution facilities to generation facilities, and includes bulk transmission and subtransmission facilities as defined in §1710.2 of this title. Any serious or life-threatening deficiencies shall be promptly repaired, disconnected, or isolated in accordance with applicable codes or regulations. Any other deficiencies found as a result of such inspections and tests are to be recorded and those records are to be maintained until such deficiencies are corrected or for the retention period required by paragraph (b) of this section, whichever is longer.

 $[63\ FR\ 3450,\ Jan.\ 23,\ 1998,\ as\ amended\ at\ 69\ FR\ 60540,\ Oct.\ 12,\ 2004]$

§1730.22 Borrower analysis.

(a) Each borrower shall periodically analyze and document its security, operations and maintenance policies, practices, and procedures to determine if they are appropriate and if they are being followed. The records of inspections and tests are also to be reviewed and analyzed to identify any trends which could indicate deterioration in the physical or cyber condition or the operational effectiveness of the system or suggest a need for changes in security, operations or maintenance policies, practices and procedures. For portions of the borrower's system that are not operated by the borrower, if any, the borrower's written analysis would also include a review of the operator's

performance under the operating agreement.

(b) When a borrower's security, operations and maintenance policies, practices, and procedures are to be reviewed and evaluated by RUS, the borrower shall:

(1) Conduct the analysis required by paragraph (a) of this section not more than 90 days prior to the scheduled RUS review;

(2) Complete RUS Form 300, Review Rating Summary, and other related forms, prior to RUS' review and evaluation; and

(3) Make available to RUS the borrower's completed RUS Form 300 (including a written explanation of the basis for each rating) and records related to the operations and maintenance of the borrower's system.

(c) For those facilities not included on the RUS Form 300 (e.g., generating plants), the borrower shall prepare and complete an appropriate supplemental form for such facilities.

 $[63\ FR\ 3450,\ Jan.\ 23,\ 1998,\ as\ amended\ at\ 69\ FR\ 60541,\ Oct.\ 12,\ 2004]$

§1730.23 Review rating summary, RUS Form 300.

RUS Form 300 in Appendix A shall be used when required by this part.

§1730.24 RUS review and evaluation.

RUS will initiate and conduct a periodic review and evaluation of the operations and maintenance practices of each borrower for the purpose of assessing loan security and determining borrower compliance with RUS policy as outlined in this part. This review will normally be done at least once every three years. The borrower will make available to RUS the borrower's policies, procedures, and records re-lated to the operations and maintenance of its complete system. Reports made by other inspectors (e.g., other Federal agencies, State inspectors, etc.) will also be made available, as applicable. RUS will not duplicate these other reviews but will use their reports to supplement its own review. RUS may inspect facilities, as well as records, and may also observe construction and maintenance work in the field. Key borrower personnel responsible for the facilities being inspected

are to accompany RUS during such inspections, unless otherwise determined by RUS. RUS personnel may prepare an independent summary of the operations and maintenance practices of the borrower. The borrower's management will discuss this review and evaluation with its Board of Directors.

§1730.25 Corrective action.

(a) For any items on the RUS Form 300 rated unsatisfactory (i.e., 0 or 1) by the borrower or by RUS, the borrower shall prepare a corrective action plan (CAP) outlining the steps (both short term and long term) the borrower will take to improve existing conditions and to maintain an acceptable rating. The CAP must include a time schedule and cost estimate for corrective actions, and must be approved by the borrower's Board of Directors. The CAP must be submitted to RUS for approval within 90 days after the completion of RUS' evaluation noted in §1730.24.

(b) The borrower must periodically report to RUS in writing progress under the CAP. This report must be submitted to RUS every six months until all unsatisfactory items are corrected unless RUS prescribes a different reporting schedule.

§1730.26 Certification.

(a) Engineer's certification. Where provided for in the borrower's loan documents, RUS may require the borrower to provide an "Engineer's Certification" as to the condition of the borrower's system (including, but not limited to, all mortgaged property.) Such certification shall be in form and substance satisfactory to RUS and shall be prepared by a professional engineer satisfactory to RUS. If RUS determines that the Engineer's Certification discloses a need for improvements to the condition of its system or any other operations of the borrower, the borrower shall, upon notification by RUS, promptly undertake to accomplish such improvements.

(b) Emergency Restoration Plan certification. The borrower's Manager or Chief Executive Officer shall provide written certification to RUS stating that a VRA has been satisfactorily completed that meets the criteria of \$1730.27 (a), (b), (c), or (d), as applicable

and §1730.27(e)(1) through (e)(8), and that the borrower has an ERP that meets the criteria of §1730.28 (a), (b), (c), or (d), as applicable, and §1730.28 (e), (f), and (g). The written certification shall be in letter form. Applicants for new RUS electric loans, loan guarantees or grants shall include the written certification in the application package submitted to RUS. If the selfcertification of an ERP and VRA are not received as set forth in this section, approval of the loan, loan guarantees or grants will not be considered until the certifications are received by RUS.

[63 FR 3450, Jan. 23, 1998, as amended at 69 FR 60541, Oct. 12, 2004]

§1730.27 Vulnerability and Risk Assessment (VRA).

(a) Each borrower with an approved RUS electric program loan as of October 12, 2004 shall perform an initial VRA of its electric system no later than July 12, 2005. Additional or periodic VRA's may be necessary if significant changes occur in the borrower's system, and records of such additional assessments shall be maintained by the borrower.

(b) Each applicant that has submitted an application for an RUS electric program loan or grant prior to October 12, 2004, but whose application has not been approved by RUS by such date, shall perform an initial VRA of its electric system in accordance with §1730.27(a).

(c) Each applicant that submits an application for an RUS electric program loan or grant between October 12, 2004 and July 12, 2005 shall perform an initial VRA of its electric system in accordance with \S 1730.27(a).

(d) Each applicant that submits an application for an RUS electric program loan or grant on or after July 12, 2005 shall include with its application package a letter certification that such applicant has performed an initial VRA of its electric system. Additional or periodic VRA's may be necessary if significant changes occur in the borrower's system, and records of such additional assessments shall be maintained by the borrower.

(e) The VRA shall include identifying:

§1730.28

(1) Critical assets or facilities considered necessary for the reliability and security of the electric power grid as described in §1730.21(c);

(2) Facilities that if damaged or destroyed would cause significant risk to the safety and health of the public;

(3) Critical assets or infrastructure owned or served by the borrower's electric system that are determined, identified and communicated as elements of national security by the consumer, State or Federal government;

(4) External system impacts (interdependency) with loss of identified system components;

(5) Threats to facilities and assets identified in paragraphs (e)(1), (e)(2), (e)(3), and (e)(4) of this section;

(6) Criticality and risk level of the borrower's system;

(7) Critical asset components and elements unique to the RUS borrower's system; and

(8) Other threats, if any, identified by an individual borrower.

[69 FR 60541, Oct. 12, 2004]

§1730.28 Emergency Restoration Plan (ERP).

(a) Each borrower with an approved RUS electric program loan as of October 12, 2004 shall have a written ERP no later than January 12, 2006. The ERP should be developed by the borrower individually or in conjunction with other electric utilities (not all having to be RUS borrowers) through the borrower's unique knowledge of its system, prudent utility practices (which includes development of an ERP) and the borrower's completed VRA. If a joint electric utility ERP is developed, each RUS borrower shall prepare an addendum to meet the requirements of paragraphs (e), (f), and (g) of this section as it relates to its system.

(b) Each applicant that has submitted an application for an RUS electric program loan or grant prior to October 12, 2004, but whose application has not been approved by RUS by such date, shall have a written ERP in accordance with § 1730.28(a).

(c) Each applicant that submits an application for an RUS electric program loan or grant between October 12, 2004 and January 12, 2006, shall have a 7 CFR Ch. XVII (1-1-06 Edition)

written ERP in accordance with §1730.28(a).

(d) Each applicant that submits an application for an RUS electric program loan or grant on or after January 12, 2006 shall include with its application package a letter certification that such applicant has a written ERP.

(e) The ERP shall include:

(1) A list of key contact emergency telephone numbers (emergency agencies, borrower management and other key personnel, contractors and equipment suppliers, other utilities, and others that might need to be reached in an emergency):

(2) A list of key utility management and other personnel and identification of a chain of command and delegation of authority and responsibility during an emergency;

(3) Procedures for recovery from loss of power to the headquarters, key offices, and/or operation center facilities;

(4) A Business Continuity Section describing a plan to maintain or re-establish business operations following an event which disrupts business systems (computer, financial, and other business systems); and

(5) Other items, if any, identified by the borrower as essential for inclusion in the ERP.

(f) The ERP must be approved and signed by the borrower's Manager or Chief Executive Officer, and approved by the borrower's Board of Directors.

(g) Copies of the most recent approved ERP must be made readily available to key personnel at all times.

(h) The ERP shall be Exercised at least annually to ensure operability and employee familiarity. Completion of the first exercise of the ERP must occur on or before January 12, 2007.

(i) If modifications are made to an existing ERP:

(1) The modified ERP must be prepared in compliance with the provisions of paragraphs (e), (f), and (g) of this section; and

(2) Additional Exercises may be necessary to maintain employee operability and familiarity.

(j) Each borrower shall maintain records of such Exercises.

[69 FR 60541, Oct. 12, 2004]

§1730.29 Grants and Grantees.

For the purposes of this part, the terms "borrower" shall include recipients of RUS electric program grants, and "applicant" shall include applicants for such grants. References to "security documents" shall, with respect to recipients of RUS electric program grants, include grant agreements and other grant-related documents.

[69 FR 60541, Oct. 12, 2004]

§§ 1730.30-1730.99 [Reserved]

APPENDIX A TO SUBPART B OF PART 1730-REVIEW RATING SUMMARY. RUS FORM 300

Borrower Designation

Date Prepared

Ratings on form are:

0: Unsatisfactory-no records

- 1: Unsatisfactory-corrective action needed
- 2: Acceptable, but should be improved-see attached recommendations
- 3: Satisfactory-no additional action required at this time
- N/A: Not applicable

PART I-TRANSMISSION and DISTRIBUTION FACILITIES

- 1. Substations (Transmission and Distribution)
- a. Safety, Clearance, Code Compliance---Rating:
- b. Physical Condition: Structure, Major Equipment, Appearance-Rating:
- c. Inspection Records Each Substation-
- Rating: d. Oil Spill Prevention-Rating:
- 2. Transmission Lines
- a. Right-of-Way: Clearing, Erosion, Ap-pearance, Intrusions-
- Rating:
- b. Physical Condition: Structure, Conductor, Guying-Rating:
- c. Inspection Program and Records-Rating:
- 3. Distribution Lines-Overhead
- a. Inspection Program and Records-Rating:
- b. Compliance with Safety Codes: Clearances-Rating: Compliance with Safety Codes: Foreign Structures-Rating: Compliance with Safety Codes: Attach-
- ments-Rating:
- c. Observed Physical Condition from Field Checking: Right-of-Way-Rating: Observed Physical Condition from Field
- Checking: Other—Rating: 4. Distribution—Underground Cable

Pt. 1730, Subpt. B, App. A

- a. Grounding and Corrosion Control-Rating:
- b. Surface Grading, Appearance-Rating:
- c. Riser Poles: Hazards, Guying, Condi-
- tion-Rating: Distribution Line Equipment: Conditions 5. and Records
- a. Voltage Regulators-Rating:
- b. Sectionalizing Equipment-
- Rating: c. Distribution Transformers-
- Rating:
- d. Pad Mounted Equipment-Safety: Locking Dead Front, Barriers-Rating:
- Pad Mounted Equipment—Appearance: Settlement, Condition-Rating:
- e. Kilowatt-hour and Demand Meter Reading and Testing-Rating:

PART II-OPERATION AND MAINTENANCE

- 6. Line Maintenance and Work Order Procedures
- a. Work Planning and Scheduling-
- Rating:
- b. Work Backlogs: Right-of-Way Maintenance-Rating: _____ Work Backlogs: Poles-Rating:
- Work Backlogs: Retirement of Idle Services-Rating: ______ Work Backlogs: Other-Rating: _____
- 7. Service Interruptions
- a. Average Annual Hours/Consumer by Cause (Complete for each of the previous 5 years)
- 1. Power Supplier
- 2. Major Storm
- 3. Scheduled
- 4. All Other
- 5. Total
- Rating:
- Emergency Restoration Plan-Rat-Ь. ing:
- 8. Power Quality
 - General Freedom from Complaints-Rating:
- 9. Loading and Load Balance
- a. Distribution Transformer Loading-Rating:
- b. Load Control Apparatus-Rating: c. Substation and Feeder Loading-Rating:
- 10. Maps and Plant Records
- a. Operating Maps: Accurate and Up-to-Date-Rating:
- b. Circuit Diagrams-Rating:
- c. Staking Sheets-Rating:___

PART III-ENGINEERING

- 11. System Load Conditions and Losses a. Annual System Loses, ____%-Rat-
- ing: b. Annual Load Factor, ____ %-Rating:___

Pt. 1735

c. Power Factor at Monthly Peak, %-Rating: d. Ratio of Individual Substation Peak kW ___Rating:____ a. Voltage Surveys-Rating: b. Substation Transformer Output Voltage Spread—Rating:_____ 13. Load Studies and Planning a. Long Range Engineering Plan-Rating:_____ Construction Plan-Rat-Work b. ing:______ c. Sectionalizing Study—Rating:_____ d. Load Data for Engineering Studies— Rating: e. Load Forecasting Data-Rating: PART IV-OPERATION AND MAINTENANCE BUDGETS For Previous 2 Years: Normal Operation—Actual \$ Normal Maintenance—Actual \$ Total—Actual \$ For Present Year: Normal Operation—Budget \$_ Normal Maintenance-Budget \$____ Total—Budget \$____ For Future 3 Years: Normal Operation-Budget \$_ Normal Maintenance-Budget \$____ Additional (Deferred) Maintenance-Budget s Total-Budget \$____ 14. Budgeting: Adequacy of Budgets For Needed Work—Rating: 15. Date Discussed with Board of Directors Remarks: EXPLANATORY NOTES Item No. ____ Comments Rated by _____ Title ___ Date Reviewed by _____ Manager _____ Date Reviewed by _____ RUS GFR _____ Date PART 1735—GENERAL POLICIES, TYPES OF LOANS, LOAN RE-QUIREMENTS—TELECOMMUNI-**CATIONS PROGRAM** Subpart A-General 1735.1 General statement.

- 1735.2 Definitions.
- 1735.3 Availability of forms.
- 1735.4-1735.9 [Reserved]

7 CFR Ch. XVII (1-1-06 Edition)

Subpart B---Loan Purposes and Basic Policies

- 1735.10 General.
- 1735.11 Area coverage.
- Nonduplication. 1735.12
- 1735.13 Location of facilities and service for nonrural subscribers.
- 1735.14 Borrower eligibility.
- 1735.15 Civil rights.
- 1735.16 Minimum loan amount.
- 1735.17 Facilities financed. 1735.18 Additional equity
- Mergers and consolidations. 1735.19
- Acquisitions. 1735.20
- 1735.21 Refinancing loans.
- 1735.22 Loan security.
- 1735.23-1735.29 [Reserved]

Subpart C-Types of Loans

- 1735.30 Hardship loans. 1735.31 RUS cost-of-money and RTB loans.
- 1735.32 Guaranteed loans.
- 1735.33 Variable interest rate loans.
- 1735.34-1735.39 [Reserved]

Subpart D—Terms of Loans

- 1735.40 General.
- 1735.41 Notes.
- 1735.42 [Reserved]
- 1735.43 Payments on loans.
- 1735.44 Prepayment premiums. 1735.45 Extension of payments.
- 1735.46 Loan security documents. 1735.47 Rescissions of loans.
- 1735.48-1735.49 [Reserved]

Subpart E-Basic Requirements for Loan Approval

- 1735.50 Administrative findings.
- 1735.51 Required findings.
- 1735.52 Findings required for particular loan purposes. 1735.53-1735.59 [Reserved]

Subpart F-Mortgage Controls on Acquisitions and Mergers

- 1735.60 Specific provisions
- 1735.61 Approval criteria.
- 1735.62 Approval of acquisitions and merg-
- ers 1735.63-1735.69 [Reserved]

Subpart G-Acquisitions Involving Loan Funds

- 1735.70 Use of loan funds.
- 1735.71 Nonrural areas.
- 1735.72 Acquisition agreements.
- 1735.73 Loan design. 1735.74 Submission of data.
- 1735.75 Interim financing.1735.76 Acquisition of affiliates.

- Sec.

ATTACHMENT D INDEX OF APPLICABLE RUS BULLETINS

.



Rural Development 🚽 Utilities Programs / Electric / Telecommunications / Water & Environmental

Electric Home Page

About the Electric Programs

Frequently Asked Questions (FAQ's)

Interagency Electric Energy Market Competition Task Force

GIS Resources

Success Stories

Staff Directory

Loan Programs

Grant Programs

Interest Rates

Box Score

Cushion of Credit

List of Materials

Federal Register

Regulations

Bulletins

Engineering

Renewable Energy

Photovoltaic Systems

Environmental

Forms

Data Collection System (DCS)

Borrower Directory

Links

Recently Revised Pages

<u>Electric Programs</u> >> Publications

Electric Programs Bulletins and Informational Publications

Electric Programs

The publications are available in a variety of formats, depending on the publication. Available formats include **html** format for viewing in your browser and in Adobe Acrobat **pdf** format for downloading and printing, as well as Microsoft Word.

Webmaster note: Due to technical difficulties with viewing Word documents (.doc) in some browsers, we recommend that you download the PDF versions of these files only. If you require the Word (.doc) version, you may save the file to your PC and then open the file locally.

Informational Publication	Size (.doc)	Text	MS Word	PDF	Description
100-1	1M	N/A	N/A	<u>.pdf</u>	Rural Electrification Act of 1936, as amended, 7 U.S.C. 901-950b (as of 1/23/2004)
200-3	17 K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Index of Electric Program Issuances (updated 5/12/98)
202-1	N/A	N/A	N/A	N/A	List of Materials Acceptable for Use on Systems of RUS Electrification Borrowers. Visit the <u>LIST OF</u> <u>MATERIALS</u> page to download the latest copy in Adobe Acrobat pdf format.
250-B6	104K	N/A	N/A	<u>.pdf</u>	Electric Program Directory (February 2006)
250-B10	219K	N/A	N/A	<u>,pdf</u>	Electric and Telecommunications Programs - General Field Representatives (GFR); Program Accounting and Regulatory Analysis - Field Accountants (FA) (May 2005)

Size (.doc)	Text	MS Word	PDF	Description
-	-	-	<u>.pdf</u>	"Buy American" Requirement with related Federal Register Notices (7/28/1955) (also available in <u>html</u>)
-	-	-	-	Design Guide for Rural Substations - No Longer Available - Replaced by <u>Bulletin 1724E-300</u>
217K	-	<u>.doc</u>	<u>.pdf</u>	Record Retention Recommendations for RUS Electric Borrowers (6/26/03)
	(.doc) - - 217K	(.doc) Text 217K -	Text Word - - - - - - 217K -	Image: Note of the second se

-

1717B-2	-	-	.doc	<u>.pdf</u>	Guide for Preparing Financial and Statistical Reports for Electric Distribution Borrowers (files located on <u>DCS</u> <u>website</u>)
1717B-3	217K	-	-	.pdf	Guide For Preparing Financial and Statistical Reports For Power Supply Borrowers and Electric Distribution Borrowers with Generating Facilities (5/12/2006)
1717M-2	199K	-	-	<u>.pdf</u>	Sale or Transfer of Capital Assets by Electric Borrower (2/9/2005)
1717-Y	85K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Settlement of Debt Owed by Electric Borrowers (9/26/97)
1724D- 101A	72K	<u>.txt</u>	<u>.doc</u>	<u>pdf</u>	Electric system long-range planning guide
1724D- 101B	65K	N/A	N/A	<u>.pdf</u>	System Planning Guide - Construction Work Plans (2000) (also available in <u>html format</u>)
1724D- 103	2K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	System planning guide, system mapping guide [only the 328k .pdf file includes all pages] {revised 7/19/93}
1724D- 104	2К	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Engineering Economics Computer Workbook Procedure - Also available: <u>Economic Analysis Worksheet (.xls</u> format)
1724D- 106	32K	N/A	N/A	<u>.pdf</u>	Considerations For Replacing Storm-Damaged Conductors (6/1/05)
1724D- 112	736K	N/A	N/A	<u>.pdf</u>	The Application of Capacitors on Rural Electric Systems (Replaces Bulletin 169-1) (4/27/01)
1724E- 104	102K	N/A	N/A	<u>.pdf</u>	Reduced Size Neutral Conductors for Overhead Rural Distribution Lines (Supersedes REA Bull. 61-4) {revised September 23, 1999}
1724E- 150	249K	N/A	N/A	<u>.pdf</u>	Unguyed Distribution Poles – Strength Requirements (7/30/03)
1724E- 151	511K	N/A	N/A	<u>.pdf</u>	Mechanical Loading on Distribution Crossarms (11/21/02)
1724E- 152	225K	N/A	N/A	<u>.pdf</u>	The Mechanics of Overhead Distribution Line Conductors (7/30/03)
1724E- 153	345K	N/A	N/A	<u>.pdf</u>	Electric Distribution Line Guys and Anchors (4/25/01)
1724E- 154	317K	N/A	N/A	<u>.pdf</u>	Distribution Conductor Clearances and Span Limitations (7/30/03)
1724E- 200	7.6M	N/A	N/A	<u>.pdf</u>	Design Manual for High Voltage Transmission Lines (9/23/04) (with May 2005 revisions) <u>May 2005 Revisions</u> (pdf format)
1724E- 202	52K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	An overview of transmission system studies
1724E- 203	34K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Guide for upgrading transmission lines
1724E- 204	680K	N/A	N/A	<u>.pdf</u>	Guide Specifications for Steel Single Pole and H-Frame Structures {revised 9/6/1997}
1724E- 205	1456K	N/A	N/A	<u>.pdf</u>	Design guide: Embedment depths for concrete and steel poles {revised 8/22/95}
1724E- 206	160K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Guide Specification for Spun, Prestressed Concrete Poles and Concrete Pole Structures (1997)
1724E- 214	950K	N/A	N/A	.pdf	Guide Specification for Standard Class Steel Transmission Poles (7/2/2001)
i i		J		- 1	

•

1724E- 216	2.2M	N/A	N/A	.pdf	Guide Specification for Standard Class Spun, Prestressed Concrete Transmission Poles (2000) (also available in <u>html format</u>)
1724E- 220	478K	N/A	N/A	<u>.pdf</u>	Procurement and Application Guide for Non-Ceramic Composite Insulators, Voltage Class 34.5 kV and Above (3/17/2005)
1724E- 300	10M	N/A	N/A	<u>.pdf</u>	Design Guide for Rural Substations {issued June 2001} - replaces Bulletin 65-1
1724E- 301	241K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Guide for the Evaluation of Large Power Transformer Losses {revised 12/17/97}
1724E- 302	227K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Design guide - oil spill prevention and control at substations
1724E- 400	25K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Building plans and specifications
1726-601	55K	-	-	<u>.pdf</u>	Electric System Construction Policies and Procedures - Interpretations {revised 7/27/2004}
1726A- 125	4K	<u>.txt</u>	<u>.doc</u>	.pdf	Joint use agreements with CATV companies (The <u>PDF</u> <u>file</u> contains scanned images of the pages in the Appendix.)
1726C- 115	4K	N/A	<u>.doc</u>	.pdf	Checking sag in a conductor using the return wave method [The .doc file is MS Word 7]
17261-602	1.18M	N/A	<u>.doc</u>	<u>.pdf</u>	Attachments to Electric Program Standard Contract Forms (2/19/04)
1728F- 700	4101K	N/A	<u>.doc</u>	<u>.pdf</u>	Specification for Wood Poles, Stubs and Anchor Logs [The .doc file is MS Word 7]
1728F- 800	112K	N/A	N/A	<u>.pdf</u>	Construction Assembly Unit Numbers and Standard Format
1728F-	10.6M	N/A	N/A	<u>.pdf</u>	Specifications and Drawings for 24.9/14.4 kV Line Construction
803	10.011				
803 1728F- 803 Notice	22K	N/A	N/A	.pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>).
803 1728F- 803 Notice 1728F- 804	22K 12.8M	N/A N/A	N/A N/A	.pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005)
803 1728F- 803 Notice 1728F- 804 1728F- 806	22K 12.8M 4.2M	N/A N/A N/A	N/A N/A N/A	.pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701	22K 12.8M 4.2M 30K	N/A N/A N/A	N/A N/A N/A . <u>doc</u>	.pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701 1728H- 701	22K 12.8M 4.2M 30K 24K	N/A N/A .txt .txt	N/A N/A N/A .doc	.pdf .pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys Specifications for quality control and inspection of timber products
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701 1728H- 702 1730B-2	22K 12.8M 4.2M 30K 24K 195K	N/A N/A .txt .txt N/A	N/A N/A .doc .doc	.pdf .pdf .pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys Specifications for quality control and inspection of timber products Guide for Electric System Emergency Restoration Plan (1/7/2005) (includes revised page 20 - 3/1/2005)
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701 1728H- 702 1730B-2 1730B-2 1730B-1 21	22K 12.8M 4.2M 30K 24K 195K 2247K	N/A N/A .txt .txt N/A	N/A N/A N/A .doc .doc N/A	.pdf .pdf .pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys Specifications for quality control and inspection of timber products Guide for Electric System Emergency Restoration Plan (1/7/2005) (includes revised page 20 - 3/1/2005) Pole Inspection and Maintenance (4/15/96)
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701 1728H- 701 1728H- 702 1730B-2 1730B-2 1730B-1 121 1730-1	22K 12.8M 4.2M 30K 24K 195K 2247K 32K	N/A N/A .txt .txt N/A N/A . <u>txt</u>	N/A N/A .doc .doc N/A N/A	.pdf .pdf .pdf .pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys Specifications for quality control and inspection of timber products Guide for Electric System Emergency Restoration Plan (1/7/2005) (includes revised page 20 - 3/1/2005) Pole Inspection and Maintenance (4/15/96) Electric System Operation and Maintenance (O&M) (1/26/98)
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701 1728H- 702 1730B-2 1730B- 121 1730-1 1767B-1	22K 12.8M 4.2M 30K 24K 195K 2247K 32K 1M	N/A N/A .txt .txt N/A N/A .txt	N/A N/A N/A .doc N/A N/A .doc	.pdf .pdf .pdf .pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys Specifications for quality control and inspection of timber products Guide for Electric System Emergency Restoration Plan (1/7/2005) (includes revised page 20 - 3/1/2005) Pole Inspection and Maintenance (4/15/96) Electric System Operation and Maintenance (O&M) (1/26/98) Uniform System of Accounts - Electric Program
803 1728F- 803 Notice 1728F- 804 1728F- 806 1728H- 701 1728H- 701 1728H- 702 1730B-2 1730B-2 1730B- 121 1767B-1 1767B-2	22K 12.8M 4.2M 30K 24K 195K 2247K 32K 1M 116K	N/A N/A .txt .txt N/A .txt .txt .txt	N/A N/A N/A .doc N/A N/A .doc .doc	.pdf .pdf .pdf .pdf .pdf .pdf .pdf .pdf	Compliance with Bulletin 50-5 or 1728F-803. Also available in <u>html format</u> . See also letter dated March 7, 2001, letter concerning assembly numbering (<u>PDF</u>) (<u>HTML</u>). Specifications and Drawings for 12.5/7.2 kV Line Construction (4/21/2005) Specifications and Drawings for Underground Electric Distribution Specifications for wood crossarms, transmission timbers, and pole keys Specifications for quality control and inspection of timber products Guide for Electric System Emergency Restoration Plan (1/7/2005) (includes revised page 20 - 3/1/2005) Pole Inspection and Maintenance (4/15/96) Electric System Operation and Maintenance (O&M) (1/26/98) Uniform System of Accounts - Electric Program

1767B-3	2K	<u>.txt</u>	<u>.doc</u>	<u>.pdf</u>	Inventory
1794A- 600	232K	N/A	<u>.doc</u>	<u>.pdf</u>	Guide for Preparing an Environmental Report for Categorically Excluded Projects {revised 12/15/98}
1794A- 601	210K	N/A	<u>.doc</u>	<u>.pdf</u>	Guide for Preparing an Environmental Report for Electric Projects Requiring an Environmental Assessment {revised 12/9/98}
1794A- 603	260K	-	-	<u>.pdf</u>	Scoping Guide for RUS Funded Projects Requiring Environmental Assessments with Scoping and Environmental Impact Statements

For regulations, see the Electric Programs Regulations page.

For RUS Accounting Bulletins, see the <u>Program Accounting Services Division</u> <u>Regulations and Bulletins</u> page.

The free Adobe Acrobat Reader is required to view PDF files. You may download it from: http://www.adobe.com/prodindex/acrobat/readstep.html

RUS Electric Program borrowers will be notified of new and revised publications by memo or hard copy. RUS borrowers should notify their business associates of the availability of these publications. All new and revised RUS Electric Program publications will be available here upon issuance. If you have any questions regarding these documents or documents not included here, call (202) 720-8674 or fax (202) 720-4120. Certain items are available directly from the Government Printing Office (202) 512-1800 (GPO Business Hours: 7:30 a.m. to 4:00 p.m. ET).

For other RUS publications, visit the main RUS Publications and Directives Page.

E-mail suggestions and comments to the <u>Electric Programs Webmaster</u>. Please include your name, e-mail address, telephone number, and company affiliation in the body of your message so that we may be able to contact you for additional information, if necessary.

Perform a USDA wide Search

For questions, contact the <u>Electric Programs Webmaster</u> Policies & Statements: <u>Nondiscrimination | Accessibility | Privacy Policy | Freedom of Information Act</u> | <u>Quality of Information</u>

ATTACHMENT F FECA COMMENTS ON PROPOSED RULE 25-6.0343

25-6.0343 Municipal Electric Utilities and Rural Electric Cooperatives.

(1) Standards of Construction.

(a) Application and Scope. This rule is intended to define construction standards for all overhead and underground electrical transmission and distribution facilities to ensure the provision of adequate and reliable electric service for operational as well as emergency purposes. This rule applies to all municipal electric utilities and rural electric cooperatives.

FECA Comments:

(1) There is no need for the Commission to define construction standards for cooperatives. The RUS has already defined construction standards for RUS cooperatives which ensure the provision of adequate and reliable electric service. Those standards have worked well.

(2) There is no need for the Commission to act to protect cooperative members (customers) as there is for the Commission to protect IOU ratepayers. Unlike IOUs, cooperatives do not have to balance the interests of customers with shareholders. In cooperatives there are no shareholders with profit expectations. There is no incentive to limit expenditures to maximize return. The only basis to determine the appropriate level of expenditures is the reliability of service. Moreover, there is already a democratically-elected organization of members in place to protect the interests of members – each cooperative's board of trustees The Commission does not need to, indeed should not act to protect members and supplant the role of the cooperatives' boards.

(3) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities, which under the plain language of the Grid Bill are not part of the "coordinated electric grid." This conclusion is also supported by more recent expressions of legislative intent as well as more than thirty years of Commission application of the Grid Bill where it has not once asserted jurisdiction over the distribution facilities for purposes of reliability.

(b) Each utility shall establish, no later than 180 days after the effective date of this rule, construction standards for overhead and underground electrical transmission and distribution facilities that conform to the provisions of this rule. Each utility shall maintain a copy of its construction standards at its main corporate headquarters and at each district office. Subsequent updates, changes, and modifications to the utility's construction standards shall be labeled to indicate the effective date of the new version and all revisions from the prior version shall be identified. Upon request, the utility shall provide access, within 2 working days, to a copy of its construction standards for review by Commission staff in Tallahassee.

FECA Comments:

(1) Because of RUS requirements, RUS cooperatives already have construction standards in place. There is no need for the Commission to require the adoption of construction standards.

Cooperatives have volunteered to make their construction standards available to Commission Staff at corporate headquarters and in Tallahassee if Staff is unable to travel.

(2) There is no need for the Commission to act to protect cooperative members (customers) as there is for the Commission to protect IOU ratepayers.

(3) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities.

(c) The facilities of each utility shall be constructed, installed, maintained and operated in accordance with generally accepted engineering practices to assure, as far as is reasonably possible, continuity of service and uniformity in the quality of service furnished.

FECA Comments:

(1) This subsection of the rule is unnecessary. Existing Rule 25-6.0345, F.A.C. already requires compliance with the NESC. In addition, Section 366.04(6), Florida Statutes states that compliance with the NESC constitutes "good engineering practice by the utilities." Thus, this rule mandate is already covered by existing rules and statutes.

(2) Because of RUS requirements, RUS cooperatives already are required to construct, install, maintain and operate facilities in accordance with generally accepted engineering practice. 7 CFR Part 1728. Indeed, RUS' standards are more demanding than generally accepted engineering practice. (3) There is no need for the Commission to act to protect cooperative members (customers) as there is a need for the Commission to protect IOU ratepayers.

(4) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities.

(d) Each utility shall, at a minimum, comply with the applicable edition of the National Electrical Safety Code (ANSI C-2) [NESC].

1. The Commission adopts and incorporates by reference the 2002 edition of the NESC, published August 1, 2001. A copy of the 2002 NESC, ISBN number 0-7381-2778-7, may be obtained from the Institute of Electric and Electronic Engineers, Inc. (IEEE).

2. Electrical facilities constructed prior to the effective date of the 2002 edition of the NESC shall be governed by the applicable edition of the NESC in effect at the time of the initial construction.

FECA Comments:

(1) Because of RUS requirements, RUS cooperatives already must comply with the NESC. 7 CFR Part 1724.50(a). Indeed, RUS' standards are more demanding than the NESC. 7 CFR Part 1724.50(b).

(2) There is no need for the Commission to act to protect cooperative members (customers) as there is a need for the Commission to protect IOU ratepayers.

(3) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities.

(e) For the construction of distribution facilities, each utility shall, to the extent reasonably practical, feasible, and cost-effective, be guided by the extreme wind loading standards specified by Figure 250-2(d) of the 2002 edition of the NESC. As part of its construction standards, each utility shall establish guidelines and procedures governing the applicability and use of the extreme wind loading standards to enhance reliability and reduce restoration costs and outage times for each of the following types of construction:

new construction;

2. major planned work, including expansion, rebuild, or relocation of existing facilities, assigned on or after the effective date of this rule; and

3. targeted critical infrastructure facilities and major thoroughfares taking into account political and geographical boundaries and other applicable operational considerations.

FECA Comments:

(1) Because of RUS requirements, RUS cooperatives already are required to construct, install, maintain and operate facilities in accordance with the NESC and RUS requirements. 7 CFR Parts 1724.50(a)(b), 1728. In addition, RUS cooperatives are required to perform Vulnerability and Risk Assessments that address risks to critical assets or facilities and other facilities that if damaged would cause significant risk to the safety and health of the public. 7 CFR Part 1730.27.

(2) The boards of trustees of cooperatives, who are democratically elected members of the cooperatives, are already assessing the standards necessary to assure reliable service to fellow members. It is presumptuous for the Commission to imply that they are not. Some boards have adopted extreme wind load standards for their systems and other have chosen not to adopt such standards. Setting aside legitimate jurisdictional questions, there is no need for the Commission to promulgate a rule that requires cooperatives' boards to perform their roles in a certain fashion. These boards are already acting in a fashion they deem reasonable, practical and cost-effective, and they should not be told to adopt construction standards with guidelines and procedures governing the applicability and use of the extreme wind loading standards. This presumes an absence of responsible conduct which has not been established by the evidence in this proceeding as well as jurisdiction that the Commission does not have. The extreme wind loading standard does not apply to structures less than 60 feet in height; thus, they are not applicable to most, if not all, distribution facilities. This proposed rule requirement simply goes too far for no apparent purpose.

(3) There is no need for the Commission to act to protect cooperative members (customers) as there is a need for the Commission to protect IOU ratepayers.

(4) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities.

6

(f) For the construction of underground distribution facilities and their supporting overhead facilities, each utility shall, to the extent reasonably practical, feasible, and costeffective, establish guidelines and procedures to deter damage resulting from flooding and storm surges.

FECA Comments:

(1) There is no need for the Commission to define construction standards or guidelines and procedures to deter flood and storm surge damage for cooperatives. The RUS has already defined construction standards for RUS cooperatives which ensure the provision of adequate and reliable electric service. Those standards have worked well. Because of RUS requirements, RUS cooperatives already are required to construct, install, maintain and operate facilities in accordance with generally accepted engineering practice. 7 CFR Part 1728. Indeed, RUS' standards are more demanding than generally accepted engineering practice. Id. Because of RUS requirements, RUS requirements, RUS cooperatives already must comply with the NESC. 7 CFR Part 1724.50(a). Indeed, RUS' standards are more demanding than demanding than the NESC. 7 CFR Part 1724.50(b). RUS standards apply to both overhead and underground facilities.

(2) The boards of trustees of cooperatives, who are democratically elected members of the cooperatives, are already assessing the standards necessary to assure reliable service to fellow members. It is presumptuous for the Commission to imply that they are not. Setting aside legitimate jurisdictional questions, there is no need for the Commission to promulgate a rule that requires cooperatives' boards to perform their roles in a certain fashion. These boards are already acting in a fashion they deem reasonable, practical and cost-effective, and they should

not be told to adopt guidelines and procedures to deter storm surge and flood damage. This presumes an absence of responsible conduct which has not been established by the evidence in this proceeding as well as jurisdiction that the Commission does not have.

(3) There is no need for the Commission to act to protect cooperative members (customers) as there is for the Commission to protect IOU ratepayers. Unlike IOUs, cooperatives do not have to balance the interests of customers with shareholders. In cooperatives there are no shareholders with profit expectations. There is no incentive to limit expenditures to maximize return. The only basis to determine the appropriate level of expenditures is the reliability of service. Moreover, there is already a democratically-elected organization of members in place to protect the interests of members – each cooperative's board of trustees The Commission does not need to, indeed should not act to protect members and supplant the role of the cooperatives' boards

(4) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities, which under the plain language of the Grid Bill are not part of the "coordinated electric grid." This conclusion is also supported by more recent expressions of legislative intent as well as more than thirty years of Commission application of the Grid Bill where it has not once asserted jurisdiction over the distribution facilities for purposes of reliability

(2) Location of the Utility's Electric Distribution Facilities. In order to facilitate safe and efficient access for installation and maintenance, to the extent practical, feasible, and costeffective, electric distribution facilities shall be placed adjacent to a public road, normally in front of the customer's premises.

(a) For initial installation, expansion, rebuild, or relocation of overhead facilities, utilities shall use easements, public streets, roads and highways along which the utility has the legal right to occupy, and public lands and private property across which rights-of-way and easements have been provided by the applicant for service.

(b) For initial installation, expansion, rebuild, or relocation of underground facilities, the utility shall require the applicant for service to provide easements along the front edge of the property, unless the utility determines there is an operational, economic, or reliability benefit to use another location.

(c) For conversions of existing overhead facilities to underground facilities, the utility shall, if the applicant for service is a local government that provides all necessary permits and meets the utility's legal, financial, and operational requirements, place facilities in road rights-of-way in lieu of requiring easements.

FECA Comments:

(1) This stated preference for the location of facilities is unnecessary. RUS Bulletin 1724D-101A already addresses the appropriate consideration of factors regarding the construction and replacement of distribution lines. These factors note that a right-of-way adjacent to a highway might provide more economical maintenance, but the Bulletin stops short of stating a preference for construction front of customer premises. This is appropriate, for in some instance construction in the rear of premises would be appropriate – for instance where there is an alleyway or road and an existing easement or right to use an existing right of way. (2) The remaining prescriptions once again presume that cooperative boards are not properly performing their responsibilities in terms of design of facilities and presume a Commission jurisdiction which it does not have. More importantly, these standards are unnecessary, as they are already being followed to the extent they are not overridden by other appropriate considerations.

(3) There is no need for the Commission to act to protect cooperative members (customers) as there is a need for the Commission to protect IOU ratepayers.

(4) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities.

(3) Third-Party Attachment Standards and Procedures.

(a) As part of its construction standards adopted pursuant to subsection (1), each utility shall establish and maintain written safety, reliability, pole loading capacity, and engineering standards and procedures for attachments by others to the utility's electric transmission and distribution poles (Attachment Standards and Procedures). The Attachment Standards and Procedures shall meet or exceed the applicable edition of the National Electrical Safety Code (ANSI C-2) pursuant to subsection (1)(d) of this rule and other applicable standards imposed by state and federal law so as to assure, as far as is reasonably possible, that third-party facilities attached to electric transmission and distribution poles do not impair electric safety, adequacy, or reliability; do not exceed pole loading capacity; and are constructed, installed, maintained, and operated in accordance with generally accepted engineering practices for the utility's service territory.

(b) No attachment to a utility's electric transmission or distribution poles shall be made except in compliance with such utility's Attachment Standards and Procedures.

FECA Comments:

(1) Pole attachment rates for cooperatives and municipals are exempt from the FCC's rates, terms and conditions regulation. If an entity wishes to attach to cooperative facilities, they must pay the full cost of changes to our facilities that are required to maintain the minimum criteria set forth in the NESC.

(2) Cooperatives have contracts with entities that attach to their facilities, and the contracts require attachments to comply with the NESC. Section (3) of the proposed rule could result in the impairment of a cooperative's contract with an attacher, and is absolutely unnecessary for cooperatives.

(3) RUS already has Bulletins in place addressing joint use agreements with CATV companies (1726A-125).

(4) There is no need for the Commission to act to protect cooperative members (customers) as there is a need for the Commission to protect IOU ratepayers.

(5) The Commission's jurisdiction over cooperatives and municipal electric utilities to preserve reliability is limited to generation and transmission facilities comprising the coordinated grid. It does not extend to distribution facilities.

(4) In establishing the construction standards and the attachment standards and procedures, the utility shall seek input from other entities with existing agreements to share the use of its electric facilities. Any dispute or challenge to a utility's construction standards by a customer, applicant for service, or attaching entity shall be resolved by the Commission. Where the expansion, rebuild, or relocation of electric distribution facilities affects existing third-party attachments, the electric utility shall seek input from and, to the extent practical, coordinate the construction of its facilities with the third-party attacher.

FECA Comments:

(1) Proposed section (4) usurps the right of a cooperative to resolve disputes with its members.

(2) It also usurps the jurisdiction of the courts to resolve contract disputes and other cases between a cooperative and an attacher. This action is clearly beyond the Commission's limited jurisdiction over cooperatives.

(3) This section potentially runs afoul of constitutional provisions prohibiting impairment of contract, as pole attachments for cooperatives are already matters subject to contract.

(4) In addition, it will be unnecessarily burdensome and costly for the cooperative's member and the cooperative if they are forced to travel to Tallahassee for a hearing on an issue that could and should have been resolved at home.

(5) If the Commission finds that a municipal electric utility or rural electric cooperative utility has demonstrated that its standards of construction will not result in service to the utility's general body of ratepayers that is less reliable, the Commission shall exempt the utility from compliance with the rule.

FECA Comments:

- (1) The standard for exemption is unclear. Less reliable than what?
- (2) There is already a statutory standard for rule waiver, and this does not appear to comply.
- (3) There is no need for the Commission to require the promulgation of municipal or

cooperative standards of construction, as set forth above in detail. Thus, there is no

corresponding need for exemption.

Specific Authority: 350.127, 366.05(1) F.S.

Law Implemented: 366.04(2) (c) (f), (5), (6), and 366.05(8) F.S.